A Flow Space for a Relatively Hyperbolic Group

Adam Mole

2013

Mathematik

A Flow Space for a Relatively Hyperbolic Group

Inaugural-Dissertation zur Erlangung des Doktorgrades der Naturwissenschaften im Fachbereich Mathematik und Informatik der Mathematisch-Naturwissenschaftlichen Fakultät der Westfälischen Wilhelms-Universität Münster

> vorgelegt von Adam Mole aus Sandwell 2013

Dekan:Prof. Dr. Martin SteinErste Gutachter:Prof. Dr. Arthur BartelsZweiter Gutachter:Prof. Dr. Johannes EbertTag der mündlichen Prüfung:18. Juni 2013Tag der Promotion:18. Juni 2013

Contents

1.	Introduction	1		
	1.1 The Farrell-Jones Conjecture	1		
	1.2 Relatively hyperbolic groups	2		
	1.3 The flow space	3		
	1.4 Outline of the Construction	5		
2.	Prerequisites	9		
	2.1 Graphs	9		
	2.2 Hyperbolicity	12		
	2.3 Fine graphs	14		
	2.4 Relatively hyperbolic groups	16		
	2.5 The boundary of a fine, Gromov hyperbolic graph	18		
	2.6 Metrising \overline{X}	25		
	2.7 Angles in a graph	28		
3.	Construction of the Flow Space	33		
	3.1 The MY-space	33		
	3.2 Motivation	37		
	3.3 The flow space	38		
	3.4 Constructing a metric	40		
4.	4. Projecting Along Geodesics			
	4.1 The first projection	43		
	4.2 Iterated projections	47		
	4.3 The total projection	51		
	4.4 Varying the initial sphere	53		
5.	A New Metric on the Vertex Set	59		
	5.1 A first attempt at a metric	59		
	5.2 Convergence of the double difference	72		
	5.3 A metric on vertices	89		
6.	The Metric on the Flow Space	94		
	6.1 Extending a metric	94		

	6.2	Extending our metric	102
	6.3	A metric on the flow space	106
7.	Prop	perties of the Flow Space	112
	7.1	The induced topology	112
	7.2	Change of base-point	114
	7.3	The group action on $FS(X)$	115
	7.4	Convergence of formal geodesics	116
8.	A Fl	ow Space for a Relatively Hyperbolic Group	118

1 Introduction

1.1 The Farrell-Jones Conjecture

The Farrell-Jones Conjecture, if true for a group *G* and a ring *R*, provides a strategy for computing the *K*-theory of the group ring R[G] in terms of the *K*-theory of group rings of the form R[V] where *V* is a virtually cyclic subgroup of *G*. (There is also a version of the Farrell-Jones Conjecture for *L*-theory but we are not interested in that version here.)

In the particular case $R = \mathbb{Z}$, the Farrell-Jones Conjecture would allow us to more easily compute $K_*(\mathbb{Z}[G])$. This case is interesting in topology because the Whitehead group, which for example appears in the *s*-cobordism thoerem, is a quotient of $K_1(\mathbb{Z}[G])$.

The Farrell-Jones Conjecture also has direct applications to other conjectures, such as the Kaplansky Conjecture about idempotents in the group ring R[G]. See [BLR08b] or [LR05] for more information about the Farrell-Jones Conjecture and its applications.

In [BLR08c] Bartels-Lück-Reich proved that the Farrell-Jones Conjecture holds for a hyperbolic group *G*. Their proof made use of special covers of $G \times \overline{X}$ where *X* is the Rips complex associated to *G* and \overline{X} is the compactification using the Gromov boundary. These covers were constructed in [BLR08a], using a flow space, first defined by Mineyev in [Min05]. Special covers were constructed over this flow space and then pulled-back to $G \times \overline{X}$ along a suitable embedding.

In this thesis we construct a flow space for a relatively hyperbolic group, i.e. a group *G* that is hyperbolic relative to a set of peripheral subgroups. The hope is to use this flow space to prove that if the Farrell-Jones Conjecture holds for all the peripheral subgroups then the Farrell-Jones Conjecture also holds for the group *G*.

1.2 Relatively hyperbolic groups

The concept of a hyperbolic group was first introduced by Gromov in [Gro87] and since then there has been considerable research on hyperbolic groups.

Relative hyperbolicity generalises this concept by allowing the group to be non-hyperbolic so long as it only behaves in a non-hyperbolic manner within certain subgroups. So, informally, a group *G* is hyperbolic relative to a set \mathcal{P} of subgroups if *G* is hyperbolic outside the subgroups $P \in \mathcal{P}$ and their conjugates (see section 2.4 for an exact definition). Hence, if the subgroups $P \in \mathcal{P}$ are themselves hyperbolic then *G* is hyperbolic everywhere, and thus is a hyperbolic group (see [Osi06, Corollary 2.41] for a proof of this fact).

Since we have refrained from giving an exact definition here we instead give some examples of groups and subgroups to which they are relatively hyperbolic.

- A group is hyperbolic if and only if it is hyperbolic relative to the trivial subgroup, hence relative hyperbolicity is indeed a generalisation of hyperbolicity.
- If G = A * B is a free product then *G* is hyperbolic relative to $\{A, B\}$. More generally, if $G = A *_H B$ is a free product amalgamated over a finite subgroup *H* then *G* is hyperbolic relative to $\{A, B\}$.
- If *X* is a systolic complex with isolated flats and *G* acts cocompactly and properly discontinuously on *X* then *G* is hyperbolic relative to its maximal virtually abelian subgroups of rank 2 (see [Els, Corollary 5.14]).
- A limit group is hyperbolic relative to its maximal non-cyclic abelian subgroups (proven independently by F. Dahmani as [Dah03a, Theorem 4.5] and E. Alibegović as [Ali05, Theorem 3.4]).
- More geometrically, the fundamental group of a hyperbolic manifold with finite volume is hyperbolic relative to the cusp subgroups.

Many facts about hyperbolic groups have been proven to hold for relatively hyperbolic groups, assuming that the fact holds for all of the peripheral subgroups. For example, Osin proved that a relatively hyperbolic group *G* has a solvable word problem if all the peripheral subgroups have a solvable word problem (see [Osi06, Theorem 5.1]) and that if none of the peripheral subgroups contain a copy of a Baumslag-Solitar group then neither does *G* (see [Osi06, Corollary 4.22]). Therefore it is reasonable to ask if a group *G* is hyperbolic relative to \mathcal{P} and the Farrell-Jones Conjecture holds for all the peripheral subgroups then does it also hold for *G*?

The Farrell-Jones Conjecture can be generalised to allow an arbitray family of subgroups instead of using virtually cyclic subgroups. With this more general version there is a transitivity principle, (see [LR05, Theorem 2.9]) that says;

Suppose \mathcal{E}, \mathcal{F} are two families of subgroups of G with $\mathcal{E} \subseteq \mathcal{F}$ and for $F \in \mathcal{F}$ let $\mathcal{E} \cap F = \{E \cap F \mid E \in \mathcal{E}\}$. If the Farrell-Jones Conjecture with the family \mathcal{F} holds for G and the Farrell-Jones Conjecture with the family $\mathcal{E} \cap F$ holds for all $F \in \mathcal{F}$ then the Farrell-Jones Conjecture with the family \mathcal{E} holds for G.

Then given a group *G* that is hyperbolic relative to \mathcal{P} we can define the family $\mathcal{F}[\mathcal{P}]$ to consist of all virtually cyclic subgroups and all peripheral subgroups, and then ask whether the Farrell-Jones Conjecture with the family $\mathcal{F}[\mathcal{P}]$ holds for *G*. The transitivity principle would then imply that if the Farrell-Jones Conjecture (with the family of virtually cyclic subgroups) is true for all the peripheral subgroups then it is also true for *G*.

1.3 The flow space

If a group *G* is hyperbolic relative to a set \mathcal{P} of subgroups then Mineyev and Yaman in [MY06] constructed a simplicial complex *X* associated to *G*, which can be thought of as an analogue of the Rips complex for a hyperbolic group. In particular, the space *X* is contractible, has finite dimension, and its 1-skeleton is a Gromov hyperbolic metric space. Furthermore, we can define a Gromov boundary for *X* and hence get a topological space $\overline{X} = X \cup \partial X$ (see section 2.5).

In [Min05] Mineyev also defined a flow space associated to what he calls a 'hyperbolic complex', which is a simplicial complex whose 1-skeleton is a uniformly locally finite, Gromov-hyperbolic graph.

If (Y, d_Y) is a uniquely geodesic metric space then we can take the flow space FS(Y) to be the space of all geodesics in Y. These geodesics may be finite or infinite in length, so we say a map $c \colon \mathbb{R} \to Y$ is a *generalised geodesic* if there are real numbers $t_0 \leq t_1 \in [-\infty, \infty]$ such that $c|_{(t_0, t_1)}$ is a geodesic and c is constant on $(-\infty, t_0]$ and $[t_1, \infty)$. Then

 $FS(Y) := \{c \colon \mathbb{R} \to Y \mid c \text{ is a generalised geodesic}\}$

with the topology defined by the metric

$$d_{FS}(c,c') := \int_{\mathbb{R}} e^{-\frac{1}{2}|t|} d_Y(c(t),c'(t)) dt$$

The flow on FS(Y) is given by translation on \mathbb{R} , i.e. $\Phi_{\tau}(c)(t) := c(t + \tau)$. Then the flow space constructed by Mineyev in [Min05] is a space that formally replicates what happens in the case of a uniquely geodesic space.

Unfortunately, the space *X* constructed in [MY06] does not have a uniformly locally finite 1-skeleton (in fact the 1-skeleton is not locally finite at all) and so the construction in [Min05] is not applicable.

Therefore, the construction of the flow space has to be adapted to use the properties we do have about the 1-skeleton of *X*, in particular we need to use the *uniform fineness* of the 1-skeleton.

Fineness is a property about graphs that is weaker than local finiteness, and can be thought of as a kind of local finiteness for edges. Fineness was introduced by Bowditch in [Bow97] and in this thesis will be a fundamental ingredient in the construction of the flow space. In section 2.3 we will give a definition of fineness and state the facts we will need later on.

Chapter 7 is about the properties of this flow space. In section 7.1 we examine the topology of the flow space induced by the metric, and in particular prove theorem 7.3 that says there is a homeomorphism

$$\left(\overline{X} \times \overline{X} \setminus \Delta(\overline{X})\right) \times \mathbb{R} \longrightarrow \left(FS(X) \setminus FS(X)^{\mathbb{R}}, \tilde{d}\right)$$

where the domain of the map has the standard product topology.

The important properties are found in section 7.4, where we look at what happens when two formal geodesics have a common end-point. We prove that under the action of the flow two such formal geodesics become arbitrarily close, as stated in the following theorem.

Theorem (7.5). Let X be a simplicial complex whose 1-skeleton is a uniformly fine, Gromov hyperbolic graph. For any $x, x' \in X, y \in \overline{X}$ and all $t, t' \in \mathbb{R}$ there exists a constant $t_0 \in \mathbb{R}$ such that

$$d_{FS}(\Phi_{\tau}(x,y,t),\Phi_{\tau}(x',y,t'+t_0)) \longrightarrow 0 \quad as \ \tau \to \infty.$$

Since we are specifically interested in when we start with a relatively hyperbolic group we can say even more;

Theorem (7.6). Let *G* be a group that is hyperbolic relative to a finite collection \mathcal{P} of subgroups and let *X* be the associated simplicial complex from section 3.1. Fix a base-point $x_0 \in X$ and let *G* act on $G \times \overline{X}$ via the diagonal action.

There is a continuous G-equivariant map $j: G \times \overline{X} \to FS(X)$ that satisfies the following property;

For any $\alpha > 0$ there is a function $f_{\alpha} \colon \mathbb{R} \to [0, \infty)$ with $f_{\alpha}(\tau) \to 0$ as $\tau \to \infty$, and there is a $\beta = \beta(\alpha) > 0$ such that for all $g, h \in G$ and $y \in \overline{X}$, if $d_G(g, h) \le \alpha$ then there exists some $t_0 \in [-\beta, \beta]$ such that for all $\tau \in \mathbb{R}$

$$d_{FS}\left(\Phi_{\tau}j(g,y),\Phi_{\tau+t_0}j(h,y)\right) \leq f_{\alpha}(\tau).$$

An informal explanation as to what this theorem is actually saying can be found after its appearance in section 7.4.

1.4 Outline of the Construction

The goal of this thesis is the construction of a flow space associated to a simplicial complex *X* whose 1-skeleton G is a uniformly fine, Gromov hyperbolic graph (such as the simplicial complex *X* associated to a relatively hyperbolic group in [MY06]).

The set underlying the flow space FS(X) is defined in section 3.3 and uses the idea of Mineyev in [Min05]. The definition of the metric on FS(X) is the complicated part. In section 3.4 we start the construction of this metric, but it will not be completed until section 6.3. The construction requires that the *double difference*

$$(x, x'|y, y') := d(x, y) - d(x, y') - d(x', y) + d(x', y')$$

of four points $x, x', y, y' \in X$ can be extended to allow x, x' to be points in the boundary ∂X . However, this is not necessarily possible if we start with the standard metric on \mathcal{G} and so we need to create a new metric on \mathcal{G} for which the double difference can be extended. In the original hyperbolic case covered by Mineyev, the construction of this metric was done in [MY02] building on work in [Min01].

A possible way to adapt the work from [Min01] was suggested Mineyev and Yaman in [MY06] and is done here explicitly in chapter 4, although we do not use precisely their suggested method.

The idea is to consider the vertex set $V = V(\mathcal{G})$ and edge set $E = E(\mathcal{G})$ as bases of Q-vector spaces QV and QE respectively, and then construct a map $g: \mathbb{Q}V \oplus \mathbb{Q}V \to \mathbb{Q}E$ such that g(a, b) is the end result of some kind of projection from *b* to *a*. The projection is built from many small steps in an iterative fashion. Each step is a projection by a distance μ , where μ is a fixed constant.

We start with a map $f_{\aleph} \colon \mathbb{Q}V \oplus \mathbb{Q}V \to \mathbb{Q}E$ in section 4.1 that depends on a constant \aleph . This map is the first step in the projection towards the sphere of radius \aleph around *a*. The element $f_{\aleph}(a, b) \in \mathbb{Q}E$ is in essence an average over the intersection of a sphere around *a* with a neighbourhood of the geodesics from *a* to *b*. This averaging procedure is why we need to work over edges instead of vertices, because we can use fineness to ensure we are averaging over finite sets, whereas if we worked with vertices then the analogous sets could be infinite. Hyperbolicity tells us that if *b* and *c* are close then the geodesics from *a* to *c* are close to the geodesics from *a* to *b*, in particular we get the following;

Lemma (4.6). There exists a constant $\eta_0 \in (0,1)$ such that for $a, b, c \in V$ if $d(b,c) \leq 2\mu$ and

$$d(a,b) = (m+1)\mu + \aleph = d(a,c)$$

for some $m \in \mathbb{N}$ then

$$|f_{\aleph}(a,b)-f_{\aleph}(a,c)|_1\leq 2\eta_0.$$

Section 4.2 is concerned with what happens when we iterate f_{\aleph} . Then in section 4.3 we define the map $g_{\aleph} \colon \mathbb{Q}V \oplus \mathbb{Q}V \to \mathbb{Q}E$, which is given by repeatedly applying the small step f_{\aleph} until we reach the sphere of radius \aleph . A crucial property of this map is the following proposition,

Proposition (4.18). *If* $2\mu \ge 7\delta + 4$ *then there are constants* L > 0 *and* $\lambda \in (0, 1)$ *such that for all* $a, b, b' \in V$

$$|g_{\aleph}(a,b) - g_{\aleph}(a,b')|_1 \le L\lambda^{(b|b')_a}$$

where $(b|b')_a = \frac{1}{2}(d(a,b) + d(a,b') - d(b,b'))$ is the Gromov product.

Then in section 4.4 we take an average over such g_{\aleph} using initial spheres of differing radii to get the map *g*.

The element $g(a, b) \in \mathbb{Q}E$ was the result of a projection from *b* to *a*, but turning around our point of view we can also think of it as the first step in moving *a* towards *b*. By counting how many times we have to apply *g* to move *a* to *b* we get a function $r: \mathbb{Q}V \oplus \mathbb{Q}V \rightarrow [0, \infty)$ which is our first attempt at a new metric on $V = V(\mathcal{G})$. This function is defined and studied in section 5.1. Although this work is based on ideas from [MY02] the results in chapter 5 are completely new.

The main results about the function *r* are summarised below.

Theorem. *The function* $r : \mathbb{Q}V \oplus \mathbb{Q}V \to [0, \infty)$ *satisfies*

(*i*) There exists a constant $K \ge 1$ such that for all $a, b \in V$

$$\frac{1}{K}d(a,b) \le r(a,b) \le d(a,b).$$

(*ii*) There exists a constant $N \ge 0$ such that for all $a, a' \in V$

$$|r(a,b) - r(a',b)| \le d(a,a') + N.$$

(iii) There exists a constant $M \ge 1$ such that for all $a, b, b' \in V$

$$|r(a,b) - r(a,b')| \le M d(b,b').$$

(iv) There exist constants $C > 0, \omega \in (0,1)$ such that for all $a, a', b, b' \in V$ if $d(a, a') \leq 1$ and $d(b, b') \leq 1$ then

$$|r(a,b) - r(a,b') - r(a',b) + r(a',b')| \le C\omega^{d(a,b)}.$$

This theorem is a combination of proposition 5.2, lemma 5.5, proposition 5.8, and proposition 5.10. The hardest of these is proposition 5.10 whose proof takes up all of section 5.2. From *r* we can define a function $\hat{d}: V \times V \rightarrow [0, \infty)$ which is a metric on *V*. This is done in section 5.3.

Before we can use this to define a metric on the flow space though we need to extend \hat{d} to all of X. An attempt at an extension was given by Mineyev in [Min05] and this extension was later used by Bartels-Lück-Reich in [BLR08a], but this extension is not a metric so in section 6.1 we give a framework for extending a metric from the vertex set of a simplicial complex to the whole simplicial complex. In particular this fixes the problem in [Min05] which carried over to [BLR08a].

Then in section 6.2 we show what happens in our case when we use \hat{d} as an input for this framework to get a metric \tilde{d} on X. The main results about this metric \tilde{d} are proposition 6.12 and theorem 6.13, which tell us the following two facts about the double difference

$$\langle a, a' \wr b, b' \rangle := \tilde{d}(a, b) - \tilde{d}(a, b') - \tilde{d}(a', b) + \tilde{d}(a', b')$$

with respect to \tilde{d} ;

Theorem. There are constants $C > 0, \omega \in (0,1)$ such that for all $a, a', b, b' \in V$ if $d(a, a') \le 1$ and $d(b, b') \le 1$ then

$$|\langle a,a' \wr b,b' \rangle| \leq \frac{1}{2} C \omega^{d(a,b)}$$

Moreover, the double difference extends continuously to $\overline{X} \times \overline{X} \times X \times X$ *.*

Using this extension we can construct a metric on the flow space in terms of \tilde{d} , which is done in section 6.3.

Acknowledgements

First and foremost I would like to thank my advisor, Arthur Bartels; not only for his mathematical help but also his patience and understanding, even when things weren't going quite according to plan.

Thanks to Henrik Rüping for many interesting discussions, and thanks to Philipp Kühl and Tibor Macko for providing a distraction via the means of algebraic surgery theory early on in my PhD years. Unfortunately these three were part of the mass exodus to Bonn following in Wolfgang Lück's wake. They have been missed.

Another who left Münster at that time, although for a different destination, is Clara Löh. A friend who made me feel welcome in Münster and has helped to keep me sane for the last few years. Thanks for what was surely a difficult task.

Yet more thanks to William Gollinger for his company while I was writing up this thesis. Typing can be boring and frustrating, sometimes simultaneously, so having someone around to provide the occasional distraction was more than welcome.

Finally, thanks to my parents, who almost certainly did not expect their young son to grow up to do mathematics. Despite my moving to another country they have always been supportive and I would not be the person I am today or achieved what I have without their help and support.

2 Prerequisites

We begin the thesis with some definitions and facts that are (mostly) not new but it will be helpful to have these results stated somewhere and so we do that here.

However, section 2.6 may not be a new idea but I have not seen the argument written up in such generality elsewhere.

2.1 Graphs

First of all we recall some definitions and fix some notation related to graphs.

Definition 2.1. A graph Γ is a set $V = V(\Gamma)$ of vertices together with a set $E = E(\Gamma)$ of edges and two maps $\varphi_{\pm} \colon E \to V$ such that $\varphi_{\pm}(e)$ are the end-points of the edge *e*. Two vertices are *adjacent* if there is an edge between them, an edge is *incident* to a vertex if the vertex is one of the end-points of the edge, and two edges are *coincident* if they share an end-point.

If v is a vertex of a graph Γ then define the graph $\Gamma - v$ to have vertex set $V(\Gamma - v) := V(\Gamma) \setminus \{v\}$ and whose edge set consists of all edges of Γ that do not have v as an end-point.

A *path* between two vertices u, v in a graph Γ is a sequence e_1, \ldots, e_k of edges such that for every

- u is an end-point of e_1 ,
- v is an end-point of e_2 ,
- for all i = 1, ..., k 1 the edges e_i and e_{i+1} are coincident.

A path that starts and ends at the same vertex is called a *loop*, and a loop that does not hit any vertex more than once is called a *circuit* (sometimes such a loop is also called a *cycle*).

A graph is *connected* if for any pair of vertices u, v there is a path from u to v. A connected graph Γ is 2-*vertex-connected* if for every vertex $v \in V(\Gamma)$ the graph $\Gamma - v$ is connected.

We will always assume that a graph is connected.

The *length* of a path is the number of edges in the path. Then we can define the distance $d_{\Gamma}(u, v)$ between two vertices u, v of Γ to be the length of the shortest path between u and v. This gives every edge length 1, and the metric d_{Γ} on V can naturally be extended to a path metric on all of Γ . We call d_{Γ} the *word metric* on Γ , where the terminology comes from geometric group theory and the concept of Cayley graphs (see definition 2.2).

A *geodesic* in Γ is a map α : $[t_0, t_1] \rightarrow \Gamma$ with $t_0, t_1 \in [0, \infty)$ such that for all $t, t' \in [t_0, t_1]$

$$d_{\Gamma}(\alpha(t), \alpha(t')) = |t - t'|.$$

If α is a geodesic in Γ and $x_0 = \alpha(s_0)$, $x_1 = \alpha(s_1)$ are points on the image of α with $s_0 \le s_1$ then let $\alpha_{[x_0,x_1]}$ denote the part of the geodesic α between x_0 and x_1 , i.e. $\alpha_{[x_0,x_1]} \equiv \alpha|_{[s_0,s_1]}$.

For $x, x' \in \Gamma$ we will use [x, x'] to denote a geodesic that starts at x and ends at x'. For any pair $x, x' \in V$ of vertices in Γ let Geod[x, x'] be the set of geodesics in Γ from x to x', let V[x, x'] be the set of all vertices of Γ that lie on some geodesic from Geod[x, x'], and for $k \in \mathbb{N}$ let V[x, x'; k] be the vertices from V[x, x'] whose distance from x is k.

Similarly, for any pair $x, x' \in V$ of vertices, let E[x, x'] be the set of all edges of Γ that form part of some geodesic from Geod[x, x'], and for $k \in \mathbb{N}$ let E[x, x'; k] be the edges from E[x, x'] whose distance from x is k.

We are interested in groups, in particular we want to consider a group as a metric space. We do this via Cayley graphs.

Definition 2.2. Let *G* be a group and let $S \subseteq G \setminus \{e\}$ be a generating set of *G*. The *Cayley graph of G with respect to S* is the graph Cay(G; S) whose vertex set is *G* and for every $s \in S$ and every $g \in G$ we add an edge from *g* to *gs*.

Remark 2.3. I have defined a Cayley graph such that the Cayley graph of $\mathbb{Z}/2\mathbb{Z}$ is a circle; two vertices with a pair of edges between them.

Any graph has a canonical metric as in definition 2.1, and so we get a metric on a Cayley graph of a group. By restricting this metric to the vertex set we get a metric on the group itself, which we call the *word metric* on *G* with respect to *S* and denote by d_S since it depends on the choice of generating set. If we use a different generating set then we get a different graph (and thus a different metric on *G*), but there is some notion of equivalence between the two graphs.

Definition 2.4. Let *X*, *Y* be metric spaces. Suppose $A \ge 1$ and $B \ge 0$ are fixed constants. A (not necessarily continuous) map $f: X \to Y$ is an (A, B)-quasi-isometric embedding if for all $x_1, x_2 \in X$,

$$\frac{1}{A}d_X(x_1, x_2) - B \le d_Y(f(x_1), f(x_2)) \le A d_X(x_1, x_2) + B.$$

A (not necessarily continuous) map $f: X \to Y$ is *quasi-dense* if there exists a constant $C \ge 0$ such that Y is contained in the C-neighbourhood of the image of f, i.e. for any $y \in Y$ there is some $x \in X$ with $d_Y(y, f(x)) \le C$. A (not necessarily continuous) map f is an (A, B)-quasi-isometry if it is a quasi-dense (A, B)-quasi-isometric-embedding.

The particular constants *A*, *B* are not always important and so are often omitted from the notation; a map *f* is a *quasi-isometry* if there are constants $A \ge 1$ and $B \ge 0$ such that *f* is an (A, B)-quasi-isometry.

Two metric spaces *X*, *Y* are *quasi-isometric* if there exists a quasi-isometry between them.

Remark 2.5. The property of being quasi-isometric is an equivalence relation on the class of metric spaces. In particular, the composition of two quasi-isometries is a quasi-isometry and if there exists a quasi-isometry $X \rightarrow Y$ then there exists a quasi-isometry $Y \rightarrow X$.

Proposition 2.6. Let G be a finitely generated group. For any finite generating set S the inclusion $(G, d_S) \hookrightarrow Cay(G; S)$ is a quasi-isometry.

Moreover, for any two finite generating sets S_1, S_2 , the graphs $Cay(G; S_1)$ and $Cay(G; S_2)$ are quasi-isometric.

Proof. The canonical inclusion $G \hookrightarrow Cay(G, S)$ is an isometric embedding and is also $\frac{1}{2}$ -dense, hence it is a quasi-isometry.

Given two finite generating sets S_1 , S_2 of G we can write all the elements of S_2 as words over S_1 , and vice versa. With this we can write a word in the generators from S_1 as a word in the generators from S_2 , and so we can compare d_{S_1} with d_{S_2} . See [BH99, Example I.8.17(3)] for full details.

In particular, the metric spaces (G, d_{S_1}) and (G, d_{S_2}) are quasi-isometric. Therefore we will often think of *G* as a metric space without explicitly stating which generating set is used.

For our purposes we will not always want to work with a Cayley graph associated to a group, instead we will want to use a graph on which the group acts with certain properties. A group acts freely, isometrically, and cocompactly on its Cayley graph, but we consider a slightly weaker type of group action. Recall that a group action $G \curvearrowright X$ is *free* if the stabiliser of every point is trivial, is *proper* if for every compact subset $K \subseteq X$ the number of elements $g \in G$ such that $g \cdot K \cap K \neq \emptyset$ is finite, and is *cocompact* if there is a compact subset $K \subset X$ such that $G \cdot K = X$. A metric space is a *length space* if the distance between any two points is equal to the infimum of lengths of paths between them.

Lemma 2.7 (Švarc-Milnor Lemma). *If a group G acts properly, isometrically, and cocompactly on a length space X then G is finitely generated and quasi-isometric to X.*

Proof. See [BH99, Proposition I.8.19].

2.2 Hyperbolicity

The concept of Gromov hyperbolicity has been quite extensively studied, and there exist a few books on the topic, see [GdLH90] for example. Hence here we only give a brief overview of facts we will need, omitting many details and proofs.

Definition 2.8. Let (X, d) be a geodesic metric space, i.e. for any pair of points there is a geodesic between them. Define the *Gromov product* of three points $x, x', y \in X$ to be

$$(x|x')_y = \frac{1}{2} \left(d(x,y) + d(x',y) - d(x,x') \right)$$

If Δ is a geodesic triangle in *X* with corners *x*, *y*, *z* then we can find a point ι_z on [x, y] such that $d(x, \iota_z) = (y|z)_x$ and $d(\iota_z, y) = (x|z)_y$. Similarly we can define a point ι_x on [y, z] and a point ι_y on [x, z].

Given $\delta \ge 0$, a geodesic triangle $\Delta = \Delta(x, y, z)$ is δ -*thin* if whenever y' is a point on [x, y] and z' is a point on [x, z] such that

$$d(x,y') = d(x,z') \le (y|z)_x$$

then $d(y', z') \leq \delta$.



A geodesic metric space *X* is δ -hyperbolic if every geodesic triangle is δ -thin, and a geodesic metric space is called *Gromov hyperbolic* if it is δ -hyperbolic for some $\delta \geq 0$.

Note that if a metric space is δ -hyperbolic then it is also δ' -hyperbolic for all $\delta' \geq \delta$, so we may always assume $\delta \geq 1$.

For alternative definitions see [BH99, Proposition III.H.1.17].

In the degenerate case that y = z we have $(y|y)_x = d(x, y)$, which immediately gives the following lemma.

Lemma 2.9. Suppose X is a δ -hyperbolic space. Fix $x, y \in X$. For all $k \leq d(x, y)$ and all $z, z' \in V[x, y; k]$ we have $d(z, z') \leq \delta$.

One useful alternative formulation of Gromov hyperbolicity is given by the next proposition.

Proposition 2.10. *A geodesic metric space X is Gromov hyperbolic if and only if there exists some* $\delta \ge 0$ *such that for any four points* $x, y, z, w \in X$ *,*

$$(x|z)_w \ge \min\{(x|y)_w, (y|z)_w\} - \delta.$$
 (2.1)

Proof. See [BH99, Proposition III.H.1.22].

Since we defined the Gromov product in terms of distances, we do not need *X* to be a geodesic space in order to make sense of the Gromov product $(x|z)_w$ and so we can use inequality (2.1) to define Gromov hyperbolicity for non-geodesic metric spaces.

Definition 2.11. An arbitrary metric space *X* is *Gromov hyperbolic* if there exists some $\delta \ge 0$ such that inequality (2.1) holds for all $x, y, z, w \in X$.

Remark 2.12. The condition that equation (2.1) holds for all $x, y, z, w \in X$ is equivalent to the condition that

$$d(x,y) + d(z,w) \le \max\{d(x,z) + d(y,w), d(x,w) + d(y,z)\} + 2\delta$$

holds for all $x, y, z, w \in X$.

We are interested in groups as metric spaces (via Cayley graphs as in definition 2.2). We want to say that a group is hyperbolic if its Cayley graph is Gromov hyperbolic, but the Cayley graph of a group is only defined up-to quasi-isometry, so we need that the property of being Gromov hyperbolic is preserved under quasi-isometries.

Proposition 2.13. Let X, Y be geodesic metric spaces. Suppose X and Y are quasi-isometric. If X is δ -hyperbolic then there exists some $\delta' \geq 0$ such that Y is δ' -hyperbolic.

Proof. See [BH99, Theorem III.H.1.9].

13

Proposition 2.13 needs the metric spaces to be geodesic metric spaces, so we cannot apply this to the inclusion $(G, d_S) \hookrightarrow \text{Cay}(G; S)$. However, every graph with its canonical metric is a geodesic metric space, which leads to the following corollary of proposition 2.13.

Corollary 2.14. Let G be a finitely generated group. Suppose $S_1, S_2 \subseteq G \setminus \{e\}$ are two finite generating sets of G. The graph $Cay(G; S_1)$ is Gromov hyperbolic if and only if the graph $Cay(G; S_2)$ is Gromov hyperbolic.

Thus the following definition makes sense.

Definition 2.15. A finitely generated Group *G* is *hyperbolic* if for one (and hence for all) finite generating sets $S \subseteq G \setminus \{e\}$ the Cayley graph Cay(*G*; *S*) is Gromov hyperbolic.

Remark 2.16. We could use definition 2.11 to define Gromov hyperbolicity for a group *G* with generating set *S*, but then we would have to worry about whether or not this depended on the choice of generating set.

We can formulate hyperbolicity for a group in terms of actions of the group on Gromov hyperbolic metric spaces, using the Švarc-Milnor lemma (lemma 2.7).

Proposition 2.17. A finitely generated group G is hyperbolic if and only if there exists a Gromov hyperbolic graph on which G acts properly, isometrically, and cocompactly.

Proof. If a group is hyperbolic then its Cayley graph (with respect to any finite generating set) is Gromov hyperbolic, and *G* acts on it properly, isometrically, and cocompactly.

Conversely, if there exists such a graph Γ then by the Svarc-Milnor lemma (lemma 2.7) the group *G* is quasi-isometric to Γ . Hence Γ is quasi-isometric to the Cayley graph of *G* (with respect to some finite generating set) by proposition 2.6. Then proposition 2.13 tells us that the Cayley graph of *G* is Gromov hyperbolic.

It is this formulation of a hyperbolic group that we will generalise to define a relatively hyperbolic group in section 2.4.

2.3 Fine graphs

Although proposition 2.17 only requires the graph to be Gromov hyperbolic, it follows from the existence of a proper, cocompact group action (by isometries) that the graph is locally finite.

Being locally finite is too strong of a condition for our purposes. We need to use a weaker notion for defining a relatively hyperbolic group.

Definition 2.18. A graph Γ is *fine* if for all $n \in \mathbb{N}$ and all edges $e \in E$ there are only finitely many circuits containing *e* and whose length is $\leq n$.

A graph Γ is *uniformly fine* if for all $n \in \mathbb{N}$ there is a constant $K \in \mathbb{N}$ such that for any edge $e \in E$ there are at most K circuits containing e and whose length is $\leq n$.

A locally finite graph is automatically fine, but a fine graph is not necessarily locally finite, as in the example below.

Construct a graph by starting with two vertices u, v and then for every positive integer $n \in \mathbb{N}_{>0}$ add a path of length n from u to v.



This graph is not locally finite, but it is fine.

I have given the definition of a fine graph that I have seen used the most often. However the definition of a fine graph in [MY06] is a uniformly fine graph in our terminology. When this difference is important in the statement or proof of something then it will be made clear what changes have to be made. The difference will not pose a problem for our main results since we will only consider fine graphs with a group acting cocompactly on them, and such graphs are automatically uniformly fine, due to the following lemma.

Lemma 2.19. Let *G* be a group that acts on a graph Γ such that there are only finitely many orbits of edges. If Γ is fine then Γ is uniformly fine.

Proof. Fix $n \in \mathbb{N}$. We need to bound the number of circuits of length $\leq n$ through an arbitrary edge *e* of Γ .

Pick a representative e_i of every *G*-orbit of edges of Γ . Let k_i be the number of circuits through e_i of length $\leq n$, which is finite since the graph is fine. Now set $K = \max_i k_i$, which is finite since there are only finitely many *G*-orbits of edges.

Given an arbitrary edge *e* there is an e_i and $g \in G$ such that $ge = e_i$. Then for every circuit *c* of length $\leq n$ containing *e*, the loop *gc* is a circuit of length $\leq n$ containing e_i . Thus the number of circuits of length $\leq n$ containing *e* is $\leq k_i \leq K$. Therefore the graph is uniformly fine. There are alternative formulations of the fineness condition, which give alternative ways to think of a fine graph and these will be useful for later proofs.

To state these alternative formulations we need some more terminology. An *arc* in a graph is a path that hits every vertex at most once, i.e. a path that never returns to a vertex that it has already visited. An *inner vertex of an arc* is a vertex hit by the arc but is not an end-point of the arc. Two arcs are *independent* if the they do not have any inner vertices in common. (So we allow independent arcs to share end-points.)

Lemma 2.20 ([Bow97, Proposition 2.1]). Let Γ be a graph. The following are equivalent;

- (*i*) Γ is fine.
- (*ii*) For all $x, x' \in V$ and $n \in \mathbb{N}$ the set of arcs of length n connecting x to x' is finite.
- (iii) For all $x, x' \in V$ and $n \in \mathbb{N}$ there does not exist an infinite collection of pairwise independent arcs of length n connecting x to x'.

In particular, from lemma 2.20(*ii*) by setting n = d(x, x') we immediately obtain the following corollary;

Corollary 2.21. If Γ is a fine graph then for any pair x, x' of vertices of Γ there are only finitely many geodesics from x to x'.

2.4 Relatively hyperbolic groups

As with hyperbolic groups, there are many different definitions of a relatively hyperbolic group. Each definition has its own advantages and disadvantages. The first definition we give comes from [Bow97] (cf. proposition 2.17).

Definition 2.22. A finitely generated group *G* is *hyperbolic relative* to a set \mathcal{P} of infinite subgroups of *G* if there exists a graph Γ with a *G*-action such that

- the graph Γ is connected, Gromov-hyperbolic, and fine,
- there are only finitely many *G*-orbits of edges of Γ,
- the stabiliser of any edge is finite,

- the stabiliser of any vertex is finite or a conjugate of an element of \mathcal{P} ,
- for all $P \in \mathcal{P}$ and $g \in G$ there is precisely one vertex whose stabiliser is gPg^{-1} ,
- the set \mathcal{P} is finite,
- every element of \mathcal{P} is finitely generated.

We call the elements of \mathcal{P} (and their conjugates) the *peripheral subgroups*.

This definition is combinatoric and neatly states all the properties required of the graph Γ . However, it doesn't give any clue as to how to find such a graph, nor is it easy to show that a group is *not* relatively hyperbolic to a set of subgroups. There is an alternative definition that is more explicit, which comes from [GM08].

Definition 2.23. Let *G* be a finitely generated group and let \mathcal{P} be a finite set of finitely generated subgroups of *G*. Let *S* be a finite generating set of *G* that is compatible with \mathcal{P} , i.e. for all $P \in \mathcal{P}$ the set $S \cap P$ generates *P*. Let Cay(*G*, *S*) be the Cayley graph of *G* with respect to *S*. For every coset gP of every peripheral subgroup $P \in \mathcal{P}$ add a new vertex v_{gP} to Cay(*G*; *S*) and join every element of gP to this new vertex with an edge. Call the resulting graph the *coned-off Cayley graph of G with respect to S and* \mathcal{P} . Denote this graph by $\widehat{Cay}(G, S; \mathcal{P})$.

Then we say that *G* is *hyperbolic relative to* \mathcal{P} if the coned-off Cayley graph $\widehat{Cay}(G, S; \mathcal{P})$ is Gromov hyperbolic and fine.

Remark 2.24. Farb introduced the coned-off Cayley graph in [Far98], in which a group *G* was defined to be hyperbolic relative to a finite set \mathcal{P} of subgroups if and only if the coned-off Cayley graph is Gromov hyperbolic. This is a strictly weaker condition.

For example consider the group $\mathbb{Z}^2 = \langle a, b | [a, b] \rangle$ relative to the subgroup $\langle a \rangle$. The coned-off Cayley graph is Gromov hyperbolic but is not fine.

However, Farb often uses a property he calls *bounded coset penetration*, and the coned-off Cayley graph is Gromov hyperbolic and fine if and only if it is Gromov hyperbolic and satisfies the bounded coset penetration property (see proposition 1 and lemma 5 in the appendix of [Dah03b]).

The action of the group *G* on the Cayley graph Cay(G, S) extends to an action of *G* on the coned-off Cayley graph $Cay(G, S; \mathcal{P})$, specifically *G* acts by left multiplication on non-cone vertices and $h \cdot v_{gP} := v_{hgP}$. Then the action of *G* on the coned-off Cayley graph satisfies all the conditions from definition 2.22. Thus if *G* is hyperbolic relative to \mathcal{P} in the sense of definition 2.23 then *G* is also hyperbolic relative to \mathcal{P} in the sense of definition 2.22. The converse is not so easy to show. In [Bow97] Bowditch proved that definition 2.22 is equivalent to the original definition given by Gromov in [Gro87]. Szczepański showed in [Szc98] that the Gromov definition implies the coned-off Cayley graph is Gromov hyperbolic, and then Dahmani in the appendix of his thesis ([Dah03b]) proved that the coned-off Cayley graph is also fine.

2.5 The boundary of a fine, Gromov hyperbolic graph

The ultimate aim of this thesis is to define a flow space, and this will be done using geodesics. We will often refer to the end-points of a geodesic but we want to allow geodesics of infinite length, so we need some concept of a boundary.

Definition 2.25. Let Γ be a fine, δ -hyperbolic graph. A *geodesic ray* is a geodesic $\alpha : [0, \infty) \to \Gamma$. Two geodesic rays $\alpha, \alpha' : [0, \infty) \to \Gamma$ are *asymptotic* if there is a constant $C \ge 0$ such that $d(\alpha(t), \alpha'(t)) \le C$ for all $t \in [0, \infty)$.

Given a fixed base-point $x_0 \in \Gamma$, we can define the (*visual*) boundary of Γ to be the set of geodesic rays starting at x_0 modulo the equivalence relation of being asymptotic, i.e.

 $\partial \Gamma := \{ \alpha : [0, \infty) \to \Gamma \mid \alpha \text{ is a geodesic ray with } \alpha(0) = x_0 \} / \sim$

where $\alpha \sim \alpha'$ if and only if they are asymptotic. We write $\alpha(\infty)$ for the point of $\partial \Gamma$ represented by the geodesic ray α .

Set $\overline{\Gamma} = \Gamma \cup \partial \Gamma$. We want a topology on $\overline{\Gamma}$, but it should not change the topology on Γ , i.e. we want the inclusion $\Gamma \hookrightarrow \overline{\Gamma}$ to be an embedding. Moreover, if we insist that $\Gamma \subseteq \overline{\Gamma}$ is an open subset then to define a topology on $\overline{\Gamma}$ we only need to define neighbourhoods of points on the boundary. For this, note that every point of Γ can be represented by a geodesic (of finite length) starting at x_0 . Thus we define a *generalised geodesic ray* to be a map $\alpha : [0, \infty) \to \Gamma$ that is either a geodesic ray or there exists some $l \ge 0$ such that $\alpha|_{[0,l]}$ is a geodesic and $\alpha|_{[l,\infty)}$ is constant. Denote by $\alpha(\infty)$ the end-point in $\overline{\Gamma}$ of α .

Then given a fixed constant $r > \delta$ and a point $\xi \in \partial \Gamma$ represented by a geodesic ray α_0 with $\alpha_0(0) = x_0$, we can define a base of neighbourhoods of ξ to be sets of the form

$$U_{x_0}(\alpha_0, n, r) := \left\{ \alpha(\infty) \middle| \begin{array}{c} \alpha \text{ is a generalised geodesic with } \alpha(0) = x_0 \\ \text{and } d(\alpha_0(n), \alpha(n)) \leq r \end{array} \right\}$$

for $n \in \mathbb{N}$. We need $r > \delta$ so that we can always find a neighbourhood in the intersection of two such sets. It doesn't matter if $r > \delta$ is fixed or not, since for any n, r, r' hyperbolicity gives

$$U_{x_0}(\alpha_0, n+r', r') \subseteq U_{x_0}(\alpha_0, n, \delta) \subseteq U_{x_0}(\alpha_0, n, r).$$

The boundary of a δ -hyperbolic geodesic metric space *X* has been oft studied, but many facts about the boundary use the assumption that *X* is a proper metric space, see for example [BH99, Section III.H.3]. A fine graph is not necessarily a proper metric space. However, properness is only used to construct the limit of a sequence of geodesic rays, by invoking Arzelà-Ascoli. So if we have an appropriate method for constructing geodesic rays then facts about proper Gromov hyperbolic spaces will also hold for fine graphs.

The corollary of Arzelà-Ascoli we need is the following.

Lemma 2.26. Let Γ be a fine, δ -hyperbolic graph. Given any point $x_1 \in \Gamma$ and any sequence $(y_n)_{n \in \mathbb{N}}$ of points in Γ , let β_n be a geodesic from x_1 to y_n . If $(y_n|y_m)_{x_1} \to \infty$ as $n, m \to \infty$ then there is a subsequence of $(\beta_n)_{n \in \mathbb{N}}$ that converges to a geodesic ray α_1 .

Proof. We construct the geodesic ray α_1 one edge at a time, showing that at every step there will be a next edge that is contained in infinitely many of the β_n . Formally we use a kind of inductive proof by contradiction.

Suppose $(\beta_{n_k})_{k \in \mathbb{N}}$ is a subsequence of $(\beta_n)_{n \in \mathbb{N}}$ such that the first *l* edges of all the β_{n_k} coincide but that the (l + 1)-th edges are distinct, i.e. for all $k \neq k'$ the (l + 1)-th edge of β_{n_k} is not the same as the (l + 1)-th edge of $\beta_{n_{k'}}$. Let x_2 be the vertex where all the β_{n_k} split.



We may assume that all the geodesics β_{n_k} start at x_2 .

Since $(y_n|y_m)_{x_2} \ge (y_n|y_m)_{x_1} - d(x_1, x_2) \to \infty$ we can pick $N \in \mathbb{N}$ large enough such that for all $k, l \ge N$ we have $(y_{n_k}|y_{n_l})_{x_2} \ge [\delta + 1] =: d$, where $[\delta + 1]$ denotes the integer part of $\delta + 1$. For $k \ge N$ set $z_k = \beta_{n_k}(d)$. Then $d(z_N, z_k) \le \delta$ by hyperbolicity and the choice of N. Let γ_k be a geodesic from z_k to z_N . The concatenation of γ_k with $\beta_{n_k}|_{[0,d]}$ is a path from x_2 to z_N whose length is $\le d + \delta$, from which we get an arc a_k from x_2 to z_N whose length is $\le d + \delta$. Since $d > \delta$ we know that each geodesic γ_k cannot pass through x_2 . Therefore the first edge of a_k is the same as the first edge of β_{n_k} . Hence we have a sequence $(a_k)_{k \in \mathbb{N}}$ of distinct arcs from x_2 to z_N , all of which have length $\leq d + \delta$. This contradicts the fineness of the graph, using lemma 2.20.

This means there cannot exist such a subsequence $(\beta_{n_k})_{k \in \mathbb{N}}$ that coincide on the first *l* edges but all take a different (l + 1)-th edge.

So we can pass to a subsequence of $(\beta_n)_{n \in \mathbb{N}}$ consisting of geodesics that all agree on the first edge. Then we can pass to a further subsequence of geodesics that agree on the first two edges. Iterating this gives a sequence of subsequences $((\beta_{k,n})_{n \in \mathbb{N}})_{k \in \mathbb{N}}$ such that the geodesics $\{\beta_{k,n} \mid n \in \mathbb{N}\}$ coincide on the first *k* edges. Then the diagonal sequence $(\beta_{n,n})_{n \in \mathbb{N}}$ converges to a geodesic ray α_1 .

Now that we have this lemma, we can use arguments as in the proper case to deduce facts about the boundary of a fine graph. In particular, the boundary $\partial\Gamma$ and the topology on $\overline{\Gamma}$ are independent of the choice of basepoint. (See [BH99, Proposition III.H 3.7(1) on page 429].) Note however, that the space $\overline{\Gamma}$ need not be compact, unlike for a proper Gromov hyperbolic metric space.

The Cayley graph of a group is only defined up to quasi-isometry; using different generating sets of the group yields quasi-isometric graphs. Hence it is important to look at what a quasi-isometry does to the boundary. For this, it is helpful to give an alternative definition of the boundary in terms of quasi-geodesic rays. But first we recall the definition of the Hausdorff distance; Let *X* be a metric space. The *Hausdorff distance* of two subsets *A*, *B* \subseteq *X* is given by

$$d_H(A,B) = \max\left\{\sup_{a\in A}\inf_{b\in B}d_X(a,b), \sup_{b\in B}\inf_{a\in A}d_X(a,b)\right\}.$$

Definition 2.27. Let Γ be a δ -hyperbolic graph. A *quasi-geodesic* in Γ is a quasi-isometric embedding of the form $\alpha \colon [0, l] \to \Gamma$. A *quasi-geodesic ray* in Γ is a quasi-isometric embedding of the form $\alpha \colon [0, \infty) \to \Gamma$.

Two quasi-geodesic rays $\alpha, \alpha' \colon [0, \infty) \to \Gamma$ are *asymptotic* if the Hausdorff distance between their images is finite.

Then given any choice of base-point $x_0 \in \Gamma$ we can define the boundary $\partial_q \Gamma$ of Γ to be the set of quasi-geodesic rays starting at x_0 up to asymptoty.

Proposition 2.28. If Γ is a fine δ -hyperbolic graph with base-point x_0 , then there is a natural bijection $\partial\Gamma \rightarrow \partial_q\Gamma$.

Proof. Any geodesic ray is a quasi-geodesic ray. Two geodesic rays are asymptotic as geodesic rays if and only if they are asymptotic as quasi-geodesic rays, so there is a natural map $\partial \Gamma \rightarrow \partial_q \Gamma$, which is injective.

For surjectivity; Given a quasi-geodesic ray α we can use lemma 2.26 to construct a geodesic ray β that is asymptotic to α .

Using quasi-geodesic rays we can show that a quasi-isometry $f \colon \Gamma \to \Gamma'$ induces a homeomorphism $f_{\partial} \colon \partial \Gamma \to \partial \Gamma$. (See [BH99, Theorem III.H 3.9].)

The definition of the boundary in terms of geodesic rays may be intuitive but it is not always the most convenient to work with. There is an alternative formulation in terms of sequences of points and the Gromov product.

Definition 2.29. Let Γ be a fine, δ -hyperbolic graph and let x_0 be a fixed base-point. Define the *boundary* of Γ to be

$$\partial_s \Gamma := \{ (x_n)_{n \in \mathbb{N}} \mid (x_n | x_m)_{x_0} \to \infty \text{ as } n, m \to \infty \} / \sim$$

where $(x_n)_{n \in \mathbb{N}} \sim (y_n)_{n \in \mathbb{N}}$ if and only if $(x_n|y_n)_{x_0} \to \infty$ as $n \to \infty$. To see that \sim is indeed an equivalence relation on the set of such sequences we need to use proposition 2.10. If a sequence $(x_n)_{n \in \mathbb{N}}$ in Γ represents a point $\xi \in \partial_s \Gamma$ then we write $\xi = \lim x_n$ and $x_n \to \xi$.

This does not depend on the choice of base-point x_0 since for any other choice y_0

$$(x_n|x_m)_{x_0} = \frac{1}{2} (d(x_n, x_0) + d(x_m, x_0) - d(x_n, x_m))$$

$$\leq \frac{1}{2} (d(x_n, y_0) + d(y_0, x_0) + d(x_m, y_0) + d(y_0, x_0) - d(x_m, x_n))$$

$$= (x_n|x_m)_{y_0} + d(x_0, y_0)$$

and similarly $(x_n|x_m)_{y_0} \leq (x_n|x_m)_{x_0} + d(x_0, y_0)$.

Remark 2.30. Given any geodesic ray $\alpha : [0, \infty) \to \Gamma$ with $\alpha(0) = x_0$ we can set $x_n = \alpha(n)$ to get a point in $\partial_s \Gamma$. Asymptotic rays yield the same point in $\partial_s \Gamma$ so we have a map $\partial \Gamma \to \partial_s \Gamma$.

Conversely, given any sequence of points $(x_n)_{n \in \mathbb{N}}$ representing a point in $\partial_s \Gamma$ we can use lemma 2.26 to construct a geodesic ray α such that

$$\lim x_n = \lim \alpha(n) \in \partial_s \Gamma$$

and so we have an inverse $\partial_s \Gamma \rightarrow \partial \Gamma$.

Therefore we may choose to represent a point in the boundary by a geodesic ray or a sequence of points whose Gromov product tends to infinity.

The concept of a Gromov product can be extended to the boundary using the $\partial_s \Gamma$ definition.

Definition 2.31. Let Γ be a fine, Gromov hyperbolic graph. For any two points $x, y \in \Gamma \cup \partial_s \Gamma$ and any point $z \in \Gamma$ define the *Gromov product* of x, y at z to be

$$(x|y)_z := \sup \liminf_{n \to \infty} (x_n|y_n)_z$$

where the supremum is taken over all sequences $(x_n)_{n \in \mathbb{N}}$ and $(y_n)_{n \in \mathbb{N}}$ in Γ such that $x_n \to x$ and $y_n \to y$.

Remark 2.32. Since the Gromov product is continuous on Γ , if x lies in Γ then we may take the sequence $(x_n)_{n \in \mathbb{N}}$ to be constantly x, i.e.

$$(x|y)_z = \sup \liminf_{n \to \infty} (x|y_n)_z$$

where the supremum is taken over all sequences $(y_n)_{n \in \mathbb{N}}$ such that $y_n \to y$ as $n \to \infty$. In particular, if x and y both lie in Γ then this supremum just gives back the original Gromov product.

Furthermore, for arbitrary $x, y \in \overline{\Gamma}$ and $z \in \Gamma$ we can use a diagonal argument to find two sequences $(x_n)_{n \in \mathbb{N}}$ and $(y_n)_{n \in \mathbb{N}}$ converging to x and y respectively such that

$$(x|y)_z = \lim_{n \to \infty} (x_n|y_n)_z.$$

Proposition 2.33. Let Γ is a fine, δ -hyperbolic graph. For any $x, y, z \in \Gamma \cup \partial_s \Gamma$ and $w \in \Gamma$ the Gromov product $(x|y)_w$ has the following properties.

- (*i*) $(x|y)_w = \infty$ *if and only if* $x = y \in \partial_s \Gamma$;
- (*ii*) $(x|z)_w \ge \min\{(x|y)_w, (y|z)_w\} 2\delta;$
- (*iii*) Suppose $x, y \in \partial_s \Gamma$. If $(x_n)_{n \in \mathbb{N}}$ and $(y_n)_{n \in \mathbb{N}}$ are sequences in Γ that tend to x and y respectively then

$$(x|y)_w - 2\delta \leq \liminf_{n \to \infty} (x_n|y_n)_w \leq (x|y)_w.$$

Proof. Let $(x_n)_{n \in \mathbb{N}}, (x'_n)_{n \in \mathbb{N}}, (y_n)_{n \in \mathbb{N}}, (y'_n)_{n \in \mathbb{N}}, (z_n)_{n \in \mathbb{N}}$ be sequences in Γ that converge to x, x, y, y and z respectively.

(*i*) If $x \in \partial_s \Gamma$ then $(x_n | x'_n)_w \to \infty$ by definition, and so $(x | x)_w = \infty$. So we just need to show the converse.

If *x* or *y* are in Γ then the Gromov product is finite, since it is bounded by the distance to *w*. So assume $x, y \in \partial_s \Gamma$. If $x \neq y$ then by definition of the equivalence relation there is some constant *K* such that $(x_n|y_n)_w \leq K$ for infinitely many *n*. Hence $\liminf_{n\to\infty} (x_n|y_n)_w \leq K$. Then by proposition 2.10

$$\liminf_{n \to \infty} (x_n | y_n)_w \ge \liminf_{n \to \infty} \left(\min\{ (x_n | x'_n)_w, (x'_n | y'_n)_w, (y'_n | y_n)_w \} - 2\delta \right)$$
$$= \liminf_{n \to \infty} (x'_n | y'_n)_w - 2\delta$$
(2.2)

since $(x_n|x'_n)_w$ and $(y_n|y'_n)_w$ both tend to ∞ . Therefore $(x|y)_w \leq K + 2\delta$ which is finite.

(*ii*) The argument again uses proposition 2.10. If $y \in \Gamma$ then we may assume $y_n = y = y'_n$ and use inequality (2.1) for x_n, y, z_n . If $y \in \partial_s \Gamma$ then $(y_n|y'_n)_w \to \infty$ and

$$\begin{split} \liminf_{n \to \infty} (x_n | z_n)_w &\geq \liminf_{n \to \infty} \left(\min\{ (x_n | y_n)_w, (y_n | y'_n)_w, (y'_n | z_n)_w \} - 2\delta \right) \\ &= \min\left\{ \liminf_{n \to \infty} (x_n | y_n)_w, \liminf_{n \to \infty} (y'_n | z_n)_w \right\} - 2\delta. \end{split}$$

This holds for arbitrary such sequences so it must also hold when taking the supremum.

(*iii*) The upper bound is immediate since $(x|y)_w$ is defined to be the supremum over all such values. The lower bound follows from inequality (2.2).

Remark 2.34. We can use the (extended) Gromov product to define a topology on $\Gamma \cup \partial_s \Gamma$, by saying a base for the neighbourhoods of a point $x \in \Gamma$ is given by metric balls $B_r(x)$, and a base for the neighbourhoods of a point $\xi \in \partial_s \Gamma$ is given by the sets

$$W_R(\xi) := \{ x \in \Gamma \cup \partial_s \Gamma \mid (x|\xi)_{x_0} \ge R \}.$$
(2.3)

with $R \ge 0$.

The advantage of this method is that it allows us to define the boundary $\partial_s X$ of an arbitrary Gromov hyperbolic metric space X, where we use definition 2.11 to define hyperbolicity for a metric space when it is not a geodesic metric space. We can still extend the Gromov product as in definition 2.31, and the statements in proposition 2.33 still hold. Hence we can define a topology on $\overline{X} := X \cup \partial_s X$.

Proposition 2.35. Let Γ be a fine, Gromov hyperbolic graph. The natural map $\Gamma \cup \partial \Gamma \rightarrow \Gamma \cup \partial_s \Gamma$, using the identity on Γ and the natural bijection $\partial \Gamma \rightarrow \partial_s \Gamma$ from remark 2.30, is a homeomorphism.

Proof. It is automatically a bijection, so we only need to show it is continuous and its inverse is continuous. The topology on Γ is unchanged, and it is an open subset, hence we only need to look at neighbourhoods of points at infinity.

Fix a base-point $x_0 \in \Gamma$ and a point $\xi \in \partial \Gamma$. By definition there exists a geodesic ray $\alpha_0 \colon [0, \infty) \to \Gamma$ with $\alpha_0(0) = x_0$ and $\alpha_0(\infty) = \xi$. Then a base of neighbourhoods for ξ in $\Gamma \cup \partial \Gamma$ is comprised of sets of the form

$$U_{x_0}(\alpha_0, n, r) := \begin{cases} \alpha(\infty) & \text{α is a generalised geodesic with $\alpha(0) = x_0$} \\ & \text{and $d(\alpha_0(n), \alpha(n)) \le r$} \end{cases}$$

for $n \in \mathbb{N}$, where $r > \delta$ is some fixed constant.

Furthermore, a base of neighbourhoods for ξ in $\Gamma \cup \partial_s \Gamma$ is comprised of sets of the form

$$W_R(\xi) := \{ x \in \Gamma \cup \partial_s \Gamma \mid (x|\xi)_{x_0} \ge R \}$$

for R > 0. We need to show that for every $n \in \mathbb{N}$ we can find some R' > 0such that $W_{R'}(\xi) \subseteq U_{x_0}(\alpha_0, n, r)$, and conversely for every R > 0 there is some $n' \in \mathbb{N}$ such that $U_{x_0}(\alpha_0, n', r) \subseteq W_R(\xi)$.

Given $n \in \mathbb{N}$ take $R' > n + 2\delta$. For any $x \in W_{R'}(\xi)$ we can find a generalised geodesic ray α from x_0 to x. Proposition 2.33(*iii*) tells us that

$$n < (x|\xi)_{x_0} - 2\delta \le \liminf_{m \to \infty} (\alpha(m)|\alpha_0(m))_{x_0} \le (x|\xi)_{x_0}$$

and in particular, for some large $m \in \mathbb{N}$ we have $n \leq (\alpha(m)|\alpha_0(m))_{x_0}$. Then hyperbolicity gives $d(\alpha(n), \alpha_0(n)) \leq \delta$, and therefore $x \in U_{x_0}(\alpha_0, n, r)$. Since x was arbitrary, we conclude $W_{R'}(\xi) \subseteq U_{x_0}(\alpha_0, n, r)$.

Conversely, suppose we are given R > 0, so that we need to find $n' \in \mathbb{N}$ such that $U_{x_0}(\alpha_0, n', r) \subseteq W_R(\xi)$. If $x \in U_{x_0}(\alpha_0, n', r)$ then there is a generalised geodesic ray α from x_0 to x such that $d(\alpha(n'), \alpha_0(n')) \leq r$. Hence

$$(\alpha(n')|\alpha_0(n'))_{x_0} = \frac{1}{2} \left(d(\alpha(n'), x_0) + d(\alpha_0(n'), x_0) - d(\alpha(n'), \alpha_0(n')) \right)$$

$$\geq n' - \frac{1}{2}r.$$
(2.4)

Now we aim to show that $(x|\xi)_{x_0} \ge (\alpha(n')|\alpha_0(n'))_{x_0}$. To see this, note that the Gromov product doesn't decrease as we travel along the geodesic rays α , α_0 , i.e. for all $m \ge n'$;

$$\begin{aligned} (\alpha(m)|\alpha_0(m))_{x_0} &= \frac{1}{2} \left(d(\alpha(m), x_0) + d(\alpha_0(m), x_0) - d(\alpha(m), \alpha_0(m)) \right) \\ &\geq m - \frac{1}{2} \left(d(\alpha(m), \alpha(n')) + d(\alpha(n'), \alpha_0(n')) + d(\alpha_0(n'), \alpha_0(m)) \right) \\ &= n' - \frac{1}{2} d(\alpha(n'), \alpha_0(n')) \\ &= (\alpha(n')|\alpha_0(n'))_{x_0}. \end{aligned}$$

So we can use proposition 2.33(*iii*) to get

$$(\alpha(n')|\alpha_0(n'))_{x_0} \leq \liminf_{m \to \infty} (\alpha(m)|\alpha_0(m))_{x_0} \leq (x|\xi)_{x_0}$$

Combining this with inequality (2.4) gives $(x|\xi)_{x_0} \ge n' - \frac{1}{2}r$. Therefore, if we take $n' \ge R + \frac{1}{2}r$ then $U_{x_0}(\alpha_0, n', r) \subseteq W_R(\xi)$.

This proves that we get the same topology around ξ regardless of which base we use, and so the map $\Gamma \cup \partial \Gamma \rightarrow \Gamma \cup \partial_s \Gamma$ is a homeomorphism. \Box

This allows us to talk about the space $\overline{\Gamma}$ without needing to specify if we are using $\partial\Gamma$ or $\partial_s\Gamma$ to define the boundary and the topology.

Remark 2.36. If *X* is a proper geodesic Gromov hyperbolic metric space then the space \overline{X} is compact. However, it is not necessarily true that $\overline{\Gamma}$ is compact for an arbitrary fine, δ -hyperbolic graph Γ . There is a way to define $\overline{\Gamma}$ to make it compact (see [Bow97, Proposition 8.6], where as sets his $\Delta(K)$ is equivalent to our $\overline{\Gamma}$) but this changes the topology on Γ , which we want to avoid doing here.

2.6 Metrising \overline{X}

In section 2.5 we defined a boundary of a fine, Gromov hyperbolic graph Γ , and used this to define a space $\overline{\Gamma}$. It is known that if Γ is a locally finite, Gromov hyperbolic graph then the space $\overline{\Gamma}$ is metrisable. (See for example, [BH99, Exercise 3.18(4)].)

We want to use a similar argument to show that the space Γ is still metrisable, even under the weaker assumptions that the graph Γ is fine and Gromov hyperbolic. We can actually do this more generally for any Gromov hyperbolic metric space *X*, where we use definition 2.11 to define hyperbolicity when the space is not a geodesic metric space, and remark 2.34 to define the topological space \overline{X} .

To show \overline{X} is metrisable we use uniform structures;

Definition 2.37. Let *S* be a set. A family \mathcal{B} of subsets of $S \times S$ is a *base of a uniform structure* on *S* if

- (*i*) Every $B \in \mathcal{B}$ is symmetric;
- (*ii*) The diagonal $\Delta(S)$ is contained in every $B \in \mathcal{B}$;
- (*iii*) For any $B_1, B_2 \in \mathcal{B}$ there exists a $B \in \mathcal{B}$ such that $B \subseteq B_1 \cap B_2$;
- (*iv*) For all $B \in \mathcal{B}$ there exists a $B' \in \mathcal{B}$ such that for all $x, y, z \in S$

$$(x,y), (y,z) \in B' \Rightarrow (x,z) \in B.$$

Moreover, a uniform structure is *separated* if for all $x \neq y \in S$ there exists some $B \in \mathcal{B}$ such that $(x, y) \notin B$.

A metric on a set *Y* determines a separated uniform structure whose base consists of elements of the form $B_r := \{(x, y) \mid d_Y(x, y) < r\}$.

Conversely, a base \mathcal{B} of a uniform structure on a set S induces a topology on S where for any point $x \in S$ the sets $B[x] := \{y \in S \mid (x, y) \in B\}$ form a base for the neighbourhoods of x (see [Kel55, Theorem 6.5]). We will refer to the set S with this induced topology as a *uniform space*. We can use the following theorem to say when a uniform space is metrisable.

Theorem 2.38. *A uniform space S is metrisable if and only if the uniform structure is separated and has a countable base.*

Proof. See [Kel55, Theorem 6.13], and note that a uniform space is Hausdorff if and only if the uniform structure is separated. \Box

So to prove a topological space is metrisable, it suffices to find a countable base for a separated uniform structure that induces that topology on the set. This is how we will show that \overline{X} is metrisable when X is a Gromov hyperbolic metric space.

The topology induced on *X* by the uniform structure should coincide with the original topology induced by the metric d_X , so for any $\epsilon \in \mathbb{Q}_+$ set $U_{\epsilon} = \{(x, y) \in X | d_X(x, y) < \epsilon\}$. For neighbourhoods at infinity, fix a base-pt $x_0 \in X$ and consider the sets $W_r := \{(x, y) \in \overline{X} | (x|y)_{x_0} \ge r\}$ with $r \in \mathbb{Q}_+$. Then set $B_{\epsilon,r} = U_{\epsilon} \cup W_r$ and $\mathcal{B} = \{B_{\epsilon,r} | \epsilon, r \in \mathbb{Q}_+\}$.

The goal is to show that \mathcal{B} is the base of a separated uniform structure on \overline{X} and that the topology induced by \mathcal{B} coincides with the topology given in remark 2.34.

Proposition 2.39. Let X be a δ -hyperbolic metric space. The family \mathcal{B} is a base of a separated uniform structure on \overline{X} .

Proof. We need to show that the family \mathcal{B} satisfies properties (*i*)-(*iv*) from definition 2.37, and that \mathcal{B} is separated.

(*i*) All of the sets U_{ϵ} and W_r are symmetric, hence every $B_{\epsilon,r}$ is symmetric.

(*ii*) For any $\epsilon > 0$ we know $\Delta(X) \subseteq U_{\epsilon}$, and by proposition 2.33(*i*) we know $(x|x)_{x_0} = \infty$ for all $x \in \partial_s X$, thus $\Delta(\partial_s X) \subseteq W_r$ for any $r \in \mathbb{Q}_+$. Combining the two tells us that $\Delta(\overline{X}) \subseteq B_{\epsilon,r}$.

(*iii*) Given B_{ϵ_1,r_1} and B_{ϵ_2,r_2} take $\epsilon = \min\{\epsilon_1, \epsilon_2\}$ and $r = \max\{r_1, r_2\}$. Then $B_{\epsilon,r} \subseteq B_{\epsilon_1,r_1} \cap B_{\epsilon_2,r_2}$.

(*iv*) Given $B_{\epsilon,r} \in \mathcal{B}$ we know $B_{\epsilon,r} = U_{\epsilon} \cup W_r$. We need to find $\epsilon', r' \in \mathbb{Q}_+$ such that

$$(x,y), (y,z) \in B_{\epsilon',r'} \Rightarrow (x,z) \in B_{\epsilon,r}.$$
(2.5)

We have different cases depending on whether the points are in X or $\partial_s X$. If $x, y, z \in X$ then we must have $(x, y), (y, z) \in U_{\epsilon'}$ and then $(x, z) \in U_{2\epsilon'}$ by the triangle inequality. If $y \in \partial_s X$ or $x, z \in \partial_s X$ then $(x, y), (y, z) \in W_{r'}$ and so $(x, z) \in W_{r'-2\delta}$ by proposition 2.33(*ii*). This leaves the case $y \in X$ and exactly one of x, z is in the boundary. Without loss of generality $z \in \partial_s X$. Hence $(x, y) \in U_{\epsilon'}$ and $(y, z) \in W_{r'}$ and applying proposition 2.33(*ii*) gives

$$\begin{aligned} &(x|z)_{x_0} \ge \min\{(x|y)_{x_0}, (y|z)_{x_0}\} - 2\delta \\ &\ge \min\{d(y, x_0) - d(x, y), (y|z)_{x_0}\} - 2\delta \\ &\ge \min\{(y|z)_{x_0} - \epsilon', (y|z)_{x_0}\} - 2\delta \\ &\ge r' - \epsilon' - 2\delta \end{aligned}$$

and so $(x, z) \in W_{r'-\epsilon'-2\delta}$. Therefore if we take $\epsilon' = \frac{1}{2}\epsilon$ and $r' = r + \frac{1}{2}\epsilon + 2\delta$ we know that (2.5) holds.

So we have proven that \mathcal{B} is a base of a uniform structure on \overline{X} . It remains to prove that the base is separated.

Given $x, y \in \overline{X}$ with $x \neq y$, if $x, y \notin \partial_s X$ then $d_X(x, y)$ is finite and for $\epsilon < d_X(x, y)$ we know that $(x, y) \notin U_{\epsilon}$. Otherwise, at least one of the points x, y lies in the boundary. Since $x \neq y$ we know $(x|y)_{x_0} < \infty$ (see proposition 2.33(*i*)) and so for $r > (x|y)_{x_0}$ we must have $(x, y) \notin W_r$. Therefore the base is separated.

The family \mathcal{B} is also countable, since we took the parameters ϵ , r to be in \mathbb{Q}_+ , and thus the topology induced by this uniform structure is metrisable by theorem 2.38. However, we have to check that this induced topology is the topology we want on \overline{X} .

Proposition 2.40. Let X be a Gromov hyperbolic metric space. The topology on X induced by the separated uniform structure \mathcal{B} coincides with the topology given in remark 2.34.

Proof. Fix $x \in \overline{X}$ and consider neighbourhoods of the point x in the two topologies. Let τ denote the topology given in remark 2.34 and let $\tau_{\mathcal{B}}$ denote the topology induced by the uniform structure.

We know that the sets $B_{\epsilon,r}[x] := \{y \in \overline{X} \mid (x, y) \in B_{\epsilon,r}\}$ define a base for the neighbourhoods of x in $\tau_{\mathcal{B}}$.

If $x \in X$ then the sets $U_{\epsilon}[x]$ form a base for the neighbourhoods of x in τ , and for any $R > d_X(x, x_0)$ the set $W_R[x]$ is empty. Hence for any $\epsilon, r \in \mathbb{Q}_+$ we have $B_{\epsilon,R}[x] = U_{\epsilon}[x] \subseteq B_{\epsilon,r}[x]$ and so $\tau_{\mathcal{B}}$ coincides with τ on X.

It remains to consider neighbourhoods of $x \in \partial_s X$. By definition the sets $W_r[x]$ form a base of open neighbourhoods of x in τ , and we also know that the sets $B_{\epsilon,r}[x]$ form a base of neighbourhoods of x in τ_B . But $U_{\epsilon}[x] = \emptyset$ for any $\epsilon \in \mathbb{Q}_+$ and so $W_r[x] = B_{\epsilon,r}[x]$.

Therefore the topologies $\tau_{\mathcal{B}}$ and τ coincide on all of \overline{X} .

Combining the results of this section leads to the following theorem.

Theorem 2.41. If X is a Gromov hyperbolic metric space then the space \overline{X} as defined in remark 2.34 is metrisable.

Proof. The family $\mathcal{B} = \{U_{\epsilon} \cup W_r | \epsilon, r \in \mathbb{Q}_+\}$ is a countable base of a separated uniform structure, using proposition 2.39, and by proposition 2.40 the topology induced on \overline{X} by \mathcal{B} coincides with the topology defined in remark 2.34. Then theorem 2.38 says the space is metrisable.

2.7 Angles in a graph

The concept of angles in a graph was used by Dahmani in his Ph.D. thesis ([Dah03b]) and expanded upon by Mineyev and Yaman in [MY06]. In this section we give definitions and state the results we will need from that article.

Definition 2.42. Let Γ be a graph. Given two coincident edges e = (a, b) and e' = (a, b') that are both incident to a vertex *a* in Γ define the *angle between e and e' at a* to be the distance between *b* and *b'* in the graph $\Gamma - a$, i.e.

$$\operatorname{ang}_{a}(e, e') := d_{\Gamma-a}(b, b')$$

where we define $ang_a(e, e') = \infty$ if *b* and *b'* are in different components of $\Gamma - a$.

Given any geodesic α in Γ and an internal vertex *a* of α we define the *angle of* α *at a* to be the angle between the pair of edges in α that are incident to *a*, and we denote this by $ang_a(\alpha)$. Set $maxang(\alpha)$ to be the maximum of all the angles of α .



In order to try to motivate why angles in a graph may be useful, recall the coned-off Cayley graph from definition 2.23, and here we use the degenerate case $\mathcal{P} = \{G\}$, where *G* is a finitely generated group. If *S* is a finite generating set of *G* then the coned-off Cayley graph $\widehat{Cay}(G, S; \{G\})$ is formed by taking the standard Cayley graph Cay(G, S) and adding a new vertex v_G to Cay(G, S) with edges joining it to every element of G = V(Cay(G, S)).

If *g* and *g*' are two non-adjacent elements in the Cayley graph Cay(*G*, *S*) then in the coned-off Cayley graph the path given by the edges (g, v_G) and (v_G, g') is a geodesic (of length 2). So by attaching the cone-vertex
we have shortened all distances to at most 2, which loses the information given by the word metric d_S on G with respect to the finite generating set S. However, this information is retained via angles, namely for any $g, g' \in G$

$$\operatorname{ang}_{v_G}((g,v_G),(v_G,g')) = d_S(g,g').$$

Thus we can think of angles in the coned-off Cayley graph $\widehat{Cay}(G, S; \mathcal{P})$ as a way of extracting information about the geometry of the original Cayley graph Cay(G, S).

The following lemma shows what angles in a Gromov hyperbolic graph can tell us about the behaviour of geodesics and geodesic triangles.

Lemma 2.43 ([MY06, Lemma 3]). Let \mathcal{G} be a graph with the path metric d having δ -thin triangles. There exists a constant κ depending only on δ such that given vertices a, b, c, and geodesics $\alpha \in \text{Geod}[b, c]$, $\beta \in \text{Geod}[a, c]$ and $\gamma \in \text{Geod}[a, b]$ we have the following:

- (*i*) If $ang_{z}(\alpha) > \kappa$ for some $z \in \alpha$ distinct from b and c, then $z \in \beta$, or $z \in \gamma$.
- (*ii*) If $z \in \alpha$, $d(c, z) < (a|b)_c$ and $ang_z(\alpha) > \kappa$, then $z \in \beta$.
- (*iii*) If $ang_c(\alpha, \beta) > \kappa$, then $c \in \gamma$.
- (*iv*) If b = a i.e. γ is a null geodesic, then $ang_c(\alpha, \beta) \leq \kappa$.

Sketch Proof. We give a quick sketch of why the first part is true. The other parts can be proven analogously.

Let z_{\pm} be the two vertices of α that are adjacent to z. If we assume that the point z does not lie on either β or γ then we can use the triangle to find a path from z_{-} to z_{+} that does not pass through z, and this is how we can bound the angle of α at z.

There are two possibilities; maybe we have to use all sides of the triangle to create the path z_{-} to z_{+} , or perhaps we only need to jump to one other side. The two pictures below indicate the idea behind how to construct a path z_{-} to z_{+} that bypasses z.



For details see [MY02, Lemma 3].

For working with angles it is helpful to consider a metric on edges defined in terms of angles, but first we show that the angle at a vertex $a \in V$ defines a metric on the edges incident to a, which follows from the next lemma.

Lemma 2.44 ([MY06, Proposition 1]). *Given three coincident edges* e_1, e_2, e_3 *that are all incident to a vertex a in a graph G one has*

- $ang_a(e_1, e_2) = ang_a(e_2, e_1).$
- $ang_a(e_1, e_3) \le ang_a(e_1, e_2) + ang_a(e_2, e_3).$

Definition 2.45. Let Γ be a graph and let e, e' be a pair of arbitrary edges of Γ . An *admissable sequence of edges from from e to e'* is a sequence e_0, e_1, \ldots, e_k of edges of Γ such that $e_0 = e, e_k = e'$, and for every $i = 1, \ldots, k$ the edges e_{i-1} and e_i are coincident.

Then the *snake distance* d^{ζ} *from e to e'* is defined by

$$d^{\zeta}(e,e') := \inf \sum_{i=1}^{k} \operatorname{ang}(e_{i-1},e_i)$$

where the infimum is taken over all admissable sequences of edges from e to e'.

Remark 2.46. If Γ is a connected graph then the function $d^{\zeta} : E \times E \to [0, \infty]$ is an extended metric on the set of edges of Γ . Recall that an extended metric is a function satisfying all the properties of being a metric except it is allowed to take the value ∞ .

Furthermore, if Γ is 2-vertex-connected then $d^{\zeta} \colon E \times E \to [0, \infty)$ is a metric on the set of edges, and we call this metric the *snake metric*.

This metric on *E* is invariant under the action of $\text{Isom}(\Gamma)$ on *E*, meaning for any isometry ψ of Γ and any edges $e, e' \in E$

$$d^{\zeta}(\psi(e),\psi(e')) = d^{\zeta}(e,e').$$

As noted earlier, the fineness condition for a graph is weaker than being locally finite. In particular, the balls around an arbitrary vertex in a fine graph need not be finite. However, we do have a kind of local finiteness for edges;

Lemma 2.47. Let Γ be a graph. If Γ is fine then any ball in the metric space (E, d^{ζ}) is finite, i.e.

 $\forall R > 0, \forall e \in E, \exists k \in \mathbb{N}, |B_R^{\zeta}(e)| \leq k.$

where $B_R^{\zeta}(e) := \{ e' \in E \mid d^{\zeta}(e, e') \leq R \}.$

This is a version of [MY06, Lemma 11] using our terminology. In that paper their definition of a fine graph coincides with our definition of a uniformly fine graph. With this stronger condition they can show that the balls in (E, d^{ζ}) are uniformly finite, so the parameter *k* would not depend on *e* (although it would still depend on *R*).

We can think of an edge as a subset of a graph, and so we can set

$$d(e, e') = \inf\{d(e, e') \mid x \in e, x' \in e'\}.$$

The snake metric is defined using admisssable sequences of edges, and such a sequence determines a path in the graph, so we can compare d(e, e') to $d^{\zeta}(e, e')$.

Lemma 2.48 ([MY06, Lemma 6]). *Let* Γ *be a hyperbolic graph. For any arbitrary edges e*, $e' \in E$,

$$d(e,e') \le d^{\zeta}(e,e').$$

Corollary 2.49. Let Γ be a hyperbolic graph. Let $e, e' \in E$ be arbitray edges. Let $x \in V$ be an end-point of e and let $x' \in V$ be an end-point of e'. Then

$$d(x, x') \le d^{\zeta}(e, e') + 2$$

We have already explained how fineness of a graph tells us something about d^{ζ} , but for a relatively hyperbolic group as in definition 2.22 we also require the graph to be Gromov hyperbolic, and then we can use the following lemma to get a version of hyperbolicity for d^{ζ} (cf. definition 2.8).

Lemma 2.50 ([MY02, Proposition 10]). A connected graph Γ is δ -hyperbolic if and only if there exists a constant $\delta' \geq 0$ such that for every geodesic triangle $\Delta(x, y, z)$ in Γ , if e_y is an edge on [x, y] and e_z is an edge on [x, z] with

$$d(x, e_y) = d(x, e_z) \le (y|z)_x$$

then

$$d^{\zeta}(e_{y},e_{z}) \leq \delta'.$$

Sometimes a picture can be very illustrative;



A corollary to lemma 2.50 is the degenerate case where y = z;

Corollary 2.51. Let Γ be a δ -hyperbolic graph. Fix two distinct points $x, y \in \Gamma$. For any $k \leq d(x, y) - 1$ and for all $e, e' \in E[x, y; k]$ we have $d^{\zeta}(e, e') \leq \delta$.

Note that lemma 2.50 is *not* claiming that the space (E, d^{ζ}) is Gromov hyperbolic (in the sense of definition 2.11).

Remark 2.52. Although the edge-hyperbolicity uses a different constant δ' we may always assume δ is large enough for edge-hyperbolicity, i.e. we take our new δ to be the maximum of the old δ and the δ' .

3 Construction of the Flow Space

The purpose of this thesis is to construct a flow space associated to a relatively hyperbolic group *G*. We already have an associated graph Γ (as per definition 2.22), but for a flow space we would like to start with a simplicial complex associated to the group, so in section 3.1 we show how to turn the graph Γ into a simplicial complex which has nice properties, eg is contractible.

Section 3.2 is a short detour into what a flow space over a tree would look like in order to motivate the definition of the flow space more generally in section 3.3. Then in section 3.4 we begin to outline how to construct a metric on the flow space.

3.1 The MY-space

Associated to a hyperbolic group *G* we have a Cayley graph Γ and over this we can build a Rips complex $P_r(\Gamma)$ with parameter *r* by adding an edge (u, v) to Γ whenever $d_{\Gamma}(u, v) \leq r$ and then taking the flag complex over the resulting graph, i.e. we add as many simplices as the resulting graph allows.

In this case if the parameter r is sufficiently large then the Rips complex $P_r(\Gamma)$ is finite-dimensional and contractible simplicial complex (see [BH99, Theorem III. Γ .3.21]). Moreover, it is a universal space for proper actions of the group G (see [MS02]).

If the group *G* is hyperbolic relative to a finite set \mathcal{P} of subgroups then we use the coned-off Cayley graph instead, but we want to retain some of the information about distances in the original Cayley graph so we do

not take the Rips complex over the coned-off Cayley graph but perform a similar construction that also takes angles into account (since angles are a way of retaining information about distances in the original Cayley graph, see the comments after definition 2.42).

Definition 3.1. Let Γ be a graph and fix a parameter $\eta \ge 1$. For vertices v, v' of Γ , an η -geodesic is a geodesic α in Γ from v to v' of length $\le \eta$ such that maxang(α) $\le \eta$.

The *MY*-space over Γ with parameter η is the simplicial complex $X_{\eta}(\Gamma)$ with vertex set $V(X_{\eta}(\Gamma)) = V(\Gamma)$ such that $\{v_0, \ldots, v_k\}$ spans a simplex in $X_{\eta}(\Gamma)$ if and only if for every pair v_i, v_j there is an η -geodesic in Γ from v_i to v_j .

I call this simplicial complex the "MY-space" to reflect who first defined it, namely Mineyev and Yaman in [MY06].

Note that an edge has no inner vertices, so its maximum angle is 0. Therefore, if $\eta \ge 1$ then the graph Γ embeds into the 1-skeleton of $X_{\eta}(\Gamma)$.

The MY-space will be our replacement of the Rips complex. To help motivate why the MY-space is a good replacement we quote two properties about it in the next two propositions.

Proposition 3.2 ([MY06, Corollary 17]). Let Γ be a uniformly fine, δ -hyperbolic graph. If $\eta \geq 1$ then the MY-space $X_{\eta}(\Gamma)$ is finite-dimensional.

For proposition 3.2 it is important that the graph is uniformly fine. To see this, let K_n denote the complete graph on n vertices and consider the graph $\Gamma := \bigvee_{n \in \mathbb{N}} K_n$, where the base-point of each K_n is a vertex. Then for any $\eta \ge 1$ the space $X_{\eta}(\Gamma)$ is $\bigvee_{n \in \mathbb{N}} \Delta^n$, a wedge of simplices. The graph Γ is fine and 2-hyperbolic but the space $X_{\eta}(\Gamma)$ is infinite-dimensional.

Proposition 3.3 ([MY06, Theorem 19]). Let Γ be a fine, δ -hyperbolic graph. Let κ be the constant from in lemma 2.43. Then for $\eta \geq 3\kappa$ the MY-space $X_{\eta}(\Gamma)$ is contractible.

By definition if *G* is a group that is hyperbolic relative to \mathcal{P} then there is a fine, δ -hyperbolic graph Γ on which *G* acts with finitely many orbits of vertices and edges and such that the stabiliser of any vertex is finite or an element of \mathcal{P} . So for κ as in lemma 2.43 and $\eta \ge 3\kappa$ we know that $X_{\eta}(\Gamma)$ is contractible. Moreover, the graph Γ is uniformly fine by lemma 2.19 so we also know that $X_{\eta}(\Gamma)$ is finite-dimensional.

Ergo, associated to a relatively hyperbolic group *G* we have a finitedimensional, contractible simplicial complex $X_{\eta}(\Gamma)$ on which the group acts. It would be nice if this action were cocompact. To prove this we need another lemma from [MY06].

Lemma 3.4 ([MY06, Lemma 15]). Let Γ be a fine, δ -hyperbolic graph. Fix $\eta \geq 1$. For any edge e = (a, b) of $X_{\eta}(\Gamma)$ there are only finitely many vertices c of $X_{\eta}(\Gamma)$ that are connected to both a and b by edges in $X_{\eta}(\Gamma)$. **Corollary 3.5.** If Γ is a fine, δ -hyperbolic graph and $\eta \geq 1$ then every edge of $X_{\eta}(\Gamma)$ is contained in only finitely many simplices.

Now we prove that the action of *G* on $X_{\eta}(\Gamma)$ is cocompact.

Proposition 3.6. Let G be a group and let Γ be a fine graph on which G acts cocompactly. Fix a parameter $\eta \ge 1$. For all $k \in \mathbb{N}$ there are only finitely many G-orbits of k-simplices in the simplicial complex $X_{\eta}(\Gamma)$.

Proof. It suffices to prove that there are only finitely many *G*-orbits of edges in $X_{\eta}(\Gamma)$ because then corollary 3.5 says there are only finitely many *G*-orbits of *k*-simplices.

To show that there are only finitely many *G*-orbits of edges in $X_{\eta}(\Gamma)$ it is enough to show that there are only finitely many *G*-orbits of η -geodesics in Γ . We prove this by showing that for any edge *e* in Γ there are only finitely many η -geodesics whose first edge is *e*, and then use the fact that there are only finitely many *G*-orbits of edges in Γ . So it is enough to show that for a fixed edge e_1 of Γ there are only finitely many η -geodesics in Γ whose first edge is e_1 .

Let α be an η -geodesic in Γ whose first edge is e_1 and let e_k be the k-th edge of α . The geodesic α gives an admissable sequence of edges from e_1 to e_k (recall from definition 2.45 that a sequence of edges is admissable if consecutive edges are coincident). The angle between consecutive edges of α is $\leq \eta$, hence

$$d^{\zeta}(e_1, e_k) \leq \sum_{i=2}^k \operatorname{ang}(e_{i-1}, e_i) \leq (k-1)\eta.$$

So for any edge *e* of α , we have $d^{\zeta}(e_1, e) \leq \eta(\eta - 1)$, since the length of α is bounded by η . By lemma 2.47 there are only finitely many such edges, therefore there are only finitely many η -geodesics whose first edge is e_1 . \Box

The 1-skeleton of any simplicial complex is a graph. We will often work with the 1-skeleton of $X_{\eta}(\Gamma)$, which we denote by $\mathcal{G}_{\eta}(\Gamma)$. Hence we want to know how graph properties of Γ can be translated to $\mathcal{G}_{\eta}(\Gamma)$.

Lemma 3.7. Let Γ be a graph and let d_{Γ} denote the canonical metric on Γ . Fix the parameter $\eta \geq 1$ and let $\mathcal{G} = \mathcal{G}_{\eta}(\Gamma)$ denote the 1-skeleton of $X_{\eta}(\Gamma)$. If $d_{\mathcal{G}}$ denotes the canonical metric on the graph \mathcal{G} then the inclusion map $(\Gamma, d_{\Gamma}) \hookrightarrow (\mathcal{G}, d_{\mathcal{G}})$ is a quasi-isometry.

Proof. The vertex set of a graph is quasi-dense and $V(\Gamma) = V(\mathcal{G}) =: V$, hence it is enough to show that the identity map $id_V : (V, d_{\Gamma}) \to (V, d_{\mathcal{G}})$ is a quasi-isometry. It follows from the definition of \mathcal{G} that for any $x, y \in V$

$$d_{\mathcal{G}}(x,y) \le d_{\Gamma}(x,y) \le \eta \, d_{\mathcal{G}}(x,y)$$

and therefore the identity map is a quasi-isometry.

Gromov hyperbolicity is preserved under quasi-isometries (see proposition 2.13) so lemma 3.7 yields the following corollary.

Corollary 3.8. *If* Γ *is a Gromov hyperbolic graph then the graph* $\mathcal{G}_{\eta}(\Gamma)$ *is Gromov hyperbolic (for any* $\eta \geq 1$ *).*

The graph property fineness is not preserved by quasi-isometries, but we can use still say something due to a lemma by Bowditch;

Lemma 3.9 ([Bow97, Lemma 4.5]). Let G be a group. Let V be a set on which G acts with finite pair stabilisers, i.e. for all $u, v \in V$ there are only finitely many elements of G that fix both u and v. Let K, L be connected G-invariant graphs with vertex set V and with finite quotient under G, i.e. there are only finitely many G-orbits of vertices and finitely many G-orbits of edges.

If K is fine then L is fine.

Remark 3.10. If we start with a relatively hyperbolic group *G* and have an associated graph Γ , as in definition 2.22, then for any $\eta \ge 1$ the 1-skeleton of $X_{\eta}(\Gamma)$ is a uniformly fine, Gromov hyperbolic graph, by corollary 3.8 and lemma 3.9, where the uniformity of the fineness uses lemma 2.19 and proposition 3.6.

The group *G* acts on *X* via the action of *G* on $V = V(\Gamma) = V(X)$. There are only finitely many orbits of simplices, by proposition 3.6.

We end this section by saying something about the stabilisers of the *G*-action on *X*.

Before we can do that though, we need a result from Osin about the intersection of peripheral subgroups.

Lemma 3.11 ([Osi06, Proposition 2.36]). *If a group G is hyperbolic relative to a finite set* \mathcal{P} *of subgroups then for all* $P, P' \in \mathcal{P}$ *and for all* $g, g' \in G$ *either*

$$gPg^{-1} = g'P'g'^{-1}$$

or

$$\left|gPg^{-1}\cap g'P'g'^{-1}\right|<\infty.$$

Osin uses yet another definition of relatively hyperbolicity in terms of (finite) relative presentations and relative Dehn functions, but Theorem 6.10 in the appendix of [Osi06] proves the equivalence of his definition to the definition given in section 2.4 here.

Proposition 3.12. Let G be a group that is hyperbolic relative to a finite set \mathcal{P} of subgroups and let Γ be an associated graph (as in definition 2.22).

For any $x \in X := X_{\eta}(\Gamma)$ either the stabiliser of x is finite or $x \in V$.

Proof. Suppose $x \in X$ has an infinite stabiliser H. Write $x = \sum_{v \in V} x_v v$ in barycentric coordinates. The group H must permute the support of x, but the support of x is finite thus the stabiliser of every $v \in \text{supp}(x)$ is infinite.

So for every $v \in \text{supp}(x)$ we can find some $g_v \in G$ and $P_v \in \mathcal{P}$ such that the stabiliser of v is $g_v P_v g_v^{-1}$.

Furthermore, by considering the diagonal action of H on $(\operatorname{supp}(x))^2$ we see that the pair stabilisers are infinite, i.e. the intersection of the stabilisers of any two vertices in the support of x is infinite. But by lemma 3.11 the intersection of two distinct conjugates of elements of \mathcal{P} is always finite. Therefore there cannot be any pair stabilisers, and the support of x is a single vertex v, in which case x = v.

3.2 Motivation

The motivation for the flow space comes from considering the a tree T, since a tree is the special case of a 0-hyperbolic graph. Consider the space \overline{T} as defined in section 2.5. Between any pair of points in \overline{T} there is a unique geodesic. Then we say that a map $c \colon \mathbb{R} \to T$ is a *generalised geodesic* if there exists $t_-, t_+ \in \overline{\mathbb{R}} := \mathbb{R} \cup \{-\infty, \infty\}$ such that

- $c|_{(-\infty,t_{-}]}$ is constant;
- $c|_{[t_-,t_+]}$ is a geodesic;
- $c|_{[t_+,\infty)}$ is constant.

So intuitively, we have a geodesic *c* which we extend to a map $\mathbb{R} \to T$ by making it constant at either end (if it is not already an infinite geodesic).

Then the *flow space* is defined to be the set of generalised geodesics in *T*. We make this a metric space via

$$d(c,c') := \int_{t \in \mathbb{R}} e^{-\frac{1}{2}|t|^2} d_T(c(t),c'(t)) dt.$$

Finally, we can define a flow Φ on this space via $\Phi_{\tau}(c)(t) := c(t + \tau)$.

This doesn't work for a general Gromov hyperbolic space because the geodesics are not necessarily unique. Hence we need to formally replicate the idea.

3.3 The flow space

Let *X* be a simplicial complex whose 1-skeleton is a fine, Gromov hyperbolic graph. In our motivating example of a tree *T*, the image of a generalised geodesic $c \colon \mathbb{R} \to T$ is determined by its end-points, hence our first attempt at a flow space is $\overline{X} \times \overline{X}$. It is possible for our generalised geodesic to be stationary (i.e. $c(-\infty) = c(\infty)$) but if one end-point lies in the boundary then the generalised geodesic cannot be stationary. Thus we consider the set $(\overline{X} \times \overline{X}) \setminus \Delta(\partial X)$, where $\Delta(\partial X)$ is the diagonal of $\partial X \times \partial X$.

This determines the 'image' of a generalised geodesic but we also need a way of encoding the parametrisation. Going back to our motivating example, fix a base-point $x_0 \in T$. For any geodesic *c* in *T* there is a unique point on *c* that is closest to x_0 . Then we can determine the parametrisation of *c* by what time it hits this point (or leaves the point if this unique point is the starting point). This gives us a value $t \in \mathbb{R}$. Therefore, we consider the space $((\overline{X} \times \overline{X}) \setminus \Delta(\partial X)) \times \mathbb{R}$.

However, there are stationary geodesics, for which the time coordinate plays no role. So we quotient out by the relation $(x, x, t) \sim (x, x, t')$.

We are not quite finished. In a tree *T*, if a generalised geodesic *c* starts at a point $x \in T$ then the flow $\Phi_{\tau}(c)$ should converge to the stationary geodesic at x as $\tau \to -\infty$. To replicate this we use $\overline{\mathbb{R}}$ instead of \mathbb{R} in our definition of the flow space, i.e. we consider $((\overline{X} \times \overline{X}) \setminus \Delta(\partial X)) \times \overline{\mathbb{R}}$, where we want the point $(x, y, -\infty)$ to correspond to the stationary geodesic at x. This is only possible if $x \notin \partial X$ so we need to remove such points beforehand. Then we know that the stationary geodesic at x is given by the point (x, x, t)(for any $t \in \overline{\mathbb{R}}$) so we need to identify $(x, y, -\infty)$ with $(x, x, -\infty)$. Similarly we need to identify (x, y, ∞) with (y, y, ∞) .

Putting everything together, we obtain the following definition.

Definition 3.13. Let *X* be a simplicial complex whose 1-skeleton is a fine, Gromov hyperbolic graph. First set

$$X' := \left\{ (x, y, t) \in \overline{X} \times \overline{X} \times \overline{\mathbb{R}} \middle| \begin{array}{l} x, y \in \partial X \Rightarrow x \neq y, \\ t = -\infty \Rightarrow x \notin \partial X, \\ t = \infty \Rightarrow y \notin \partial X \end{array} \right\}$$
(3.1)

and then define the *flow space* as the quotient $FS(X) := X' / \sim$ where

$$\begin{array}{ll} \forall x \in \overline{X}, \forall t, t' \in \overline{\mathbb{R}}, & (x, x, t) \sim (x, x, t'); \\ \forall x, y, y' \in \overline{X}, & (x, y, -\infty) \sim (x, y', -\infty); \\ \forall x, x', y \in \overline{X}, & (x, y, \infty) \sim (x', y, \infty). \end{array}$$

The topology on FS(X) is defined to be the quotient topology, where we give $\overline{X} \times \overline{X} \times \overline{\mathbb{R}}$ the product topology and X' the subspace topology. The

flow on FS(X) is given by

$$\Phi_{\tau}(x, y, t) := (x, y, t + \tau).$$
(3.2)

Note that we do have $\Phi_{\tau}(x, y, t) \rightarrow (x, x, -\infty)$ as $\tau \rightarrow -\infty$ with this definition, and analogously for $\tau \rightarrow \infty$.

This formulation of a flow space was first given by Mineyev in [Min05], where he used the notation $*\overline{X}$ for the flow space FS(X).

A quick remark about this definition. In the motivating example of a tree *T* we thought of the flow space as the generalised geodesics $c \colon \mathbb{R} \to T$ and then said that given a base-point $x_0 \in T$ we could always get a parameter *t*. This gives an identification of the space of generalised geodesics with the abstract flow space as in definition 3.13, but this identification depends on the choice of base-point. Using a different base-point would still yield an identification of the two versions of a flow space but it would not necessarily give a generalised geodesic *c* the same parameter $t \in \mathbb{R}$.

Therefore, although we do not use a choice of base-point in this definition we will often think of one as being implicit. Sometimes we will want to explicitly use a base-point, in which case we will use the following definition.

Definition 3.14. Let *X* be a finite-dimensional simplicial complex whose 1-skeleton \mathcal{G} is a fine, Gromov hyperbolic graph. Suppose \tilde{d} is a metric on *X* whose restriction to $X^{(0)}$ coincides with $d_{\mathcal{G}}$. Given any fixed basepoint $x_0 \in X$, for any $x, y \in \overline{X}$ define the map $\theta_{x,y}^{x_0} \colon \overline{\mathbb{R}} \to [-(y|x_0)_x, (x|x_0)_y]$ by

$$\theta_{x,y}^{x_0}(t) := \begin{cases} -(y|x_0)_x & \text{if } t \le -(y|x_0)_x \\ t & \text{if } -(y|x_0)_x \le t \le (x|x_0)_y \\ (x|x_0)_y & \text{if } (x|x_0)_y \le t \end{cases}$$

and observe that for the case x = y the map $\theta_{x,x}^{x_0}$ is identically zero.

The idea is to replicate what happens in a tree. If we have a fixed base-point $x_0 \in X$ then we can think of (x, y, 0) as the formal generalised geodesic whose 'closest' point to x_0 is at time 0. Then for arbitrary *t* we just use the flow relation $(x, y, t) = \Phi_t(x, y, 0)$.

So the map $\theta_{x,y}^{x_0}$ can be thought of as evaluation at time 0; starting at the point on [x, y] that is 'closest' to x_0 we move a distance $|\theta_{x,y}^{x_0}(t)|$ along [x, y] and the sign says which direction, namely towards x if negative and towards y if positive.

3.4 Constructing a metric

If *X* is a Gromov hyperbolic metric space then we would like to define a metric on FS(X). In our motivating example, we defined the distance between $c, c' \in FS(T)$ by integrating over the image of the generalised geodesics c, c'. Note that $c(s) = \Phi_s(c)(0)$ for all $s \in \mathbb{R}$ so it is enough to know distances between the points c(0) and c'(0) for all $c, c' \in FS(X)$.

Back in the general case the idea is to think of a point $(x, y, t) \in FS(X)$ as a formal generalised geodesic. Then there should be a point $(x, y, t)_0 \in X$ that corresponds to the formal generalised geodesic evaluated at time t = 0. However, we have no such evaluation map $FS(X) \to X$ so we need a way of calculating the distance without using an evaluation map.

We start by defining a function $X \times FS(X) \rightarrow \mathbb{R}$ that in a sense gives a distance from a point $z \in X$ to the 'point' $(x, y, t)_0$.

Then for any two points $(x, y, t), (x', y', t') \in FS(X)$ we can compare how close $(x, y, t)_0$ and $(x', y', t')_0$ are to a point $z \in X$. If we do this for all points $z \in X$ then we should be able to say something about how close the 'points' $(x, y, t)_0$ and $(x', y', t')_0$ are.

To define a distance from $(x, y, t)_0$ to a point $z \in X$ we want to find a point $(x, y)_z$ on the formal generalised geodesic [x, y] that is 'closest' to z and then the distance from $(x, y, t)_0$ to z is the distance along the formal generalised geodesic from $(x, y, t)_0$ to $(x, y)_z$ plus the distance from $(x, y)_z$ to z. The distance from z to the point $(x, y)_z$ is the distance from z to the formal generalised geodesic [x, y], which is given by the Gromov product $(x|y)_z$.

To make sense of $(x, y, t)_0$ we need a base-point $x_0 \in X$ (see definition 3.14). We need to know what the distance between $(x, y)_z$ and $(x, y, t)_0$ is in terms of x_0, x, y, t and z. If $(x, y)_0$ denotes the point of [x, y] that is 'closest' to x_0 then by definition the distance between $(x, y)_0$ and $(x, y, t)_0$ is $|\theta_{x,y}^{x_0}(t)|$, where $\theta_{x,y}^{x_0}(t)$ is positive if $(x, y)_0$ lies between $(x, y, t)_0$ and x, and is negative if $(x, y)_0$ lies between $(x, y, t)_0$ and y. So for $x, y \in X$ we can write the distance between $(x, y, t)_0$ and $(x, y)_z$ as the following expression in terms of Gromov products;

However, if *x* and *y* are in the boundary then all of the Gromov products are infinite and so the expression does not make sense. We get around this problem by looking specifically at the value $(y|z)_x - (y|x_0)_x$;

Definition 3.15. Let *X* be an arbitrary metric space. The *double difference* of four points $x, y, z, w \in X$ is

$$(x, y|z, w) := (y|z)_x - (y|w)_x.$$
(3.3)

Before returning to the problem of defining a metric on our flow space, we list some properties of the double difference.

Proposition 3.16. Let X be an arbitrary metric space. The double difference satisfies the following properties for any points in X;

- (0) $(x, x'|y, y') = \frac{1}{2} (d_X(x, y) d_X(x, y') d_X(x', y) + d_X(x', y'));$
- (1) (x, x'|y, y') = (y, y'|x, x');
- (2) (x, x'|y, y') = -(x', x|y, y') = -(x, x'|y', y);
- (3) (x, x|y, y') = 0 = (x, x'|y, y);
- (4) (x, x'|y, y') + (x', x''|y, y') = (x, x''|y, y');
- (5) (x, y|z, w) + (z, x|y, w) + (y, z|x, w) = 0.

Now we want to define the distance from a point $z \in X$ to $(x, y, t)_0$ for $(x, y, t) \in FS(X)$ to be $(x|y)_z + |\theta_{x,y}^{x_0}(t) - (x, y|z, x_0)|$ but for this we need to know that the double difference is defined for $x, y \in \partial X$.

This is not necessarily true for an arbitrary metric space *X*. Moreover, it is not true for an arbitrary Gromov hyperbolic graph. To solve this we construct a new metric on a uniformly fine, δ -hyperbolic graph such that the double difference with respect to this new metric can be extended to the boundary. Then we can use this new metric to define a metric on *FS*(*X*) using the argument laid out above, which is done formally in section 6.3.

4 Projecting Along Geodesics

In section 3.4 we started to sketch how to define a metric on the flow space FS(X) for a Gromov hyperbolic metric space X, but we encountered a problem that the double difference (as in definition 3.15) is not well-defined for points on the boundary of X.

In this section we will use the work of [Min01] together with the modification suggested in [MY06] to define a projection map g that will be used in chapter 5 to construct the metric \hat{d} on the vertex set of X.

Our projection map g is essentially the map \overline{f} as in [MY06, Proposition 45], where several facts about the map \overline{f} are stated but not explicitly proven. Instead it is said that the proofs are analogous to proofs given in [Min01]. In this section we will define the projection map g, dependent upon three constants I, J, and R, and prove the following theorem.

Theorem 4.1. Let \mathcal{G} be a uniformly fine, Gromov hyperbolic graph. There is a map $g: \mathbb{Q}V \oplus \mathbb{Q}V \to \mathbb{Q}E$ that satisfies the following properties;

- (*i*) It is invariant under isometries of \mathcal{G} (where $\text{Isom}(\mathcal{G})$ acts on $\mathcal{G} \times \mathcal{G}$ diagonally).
- (*ii*) For all $a, b \in V$, for every integer \aleph with $I \leq \aleph \leq J$ the support of g(a, b) contains $E[a, b; \aleph]$.
- (iii) For all $a, b \in V$ and all $x \in supp(\varphi g(a, b)), d(x, b) \leq d(a, b) 1$.
- (*iv*) For all $a, b \in V$ and all $x \in supp(\varphi g(a, b))$ there is an \aleph with $I \leq \aleph \leq J$ such that for all $u_{\aleph} \in V[a, b; \aleph]$ we have $d(x, u_{\aleph}) \leq 3\delta + R + 2 =: C_0$.
- (v) For all $a, b \in V$, $diam(supp(\varphi g(a, b))) \le 2C_0 + J I$.
- (vi) There exist constants L > 0 and $\lambda \in (0, 1)$ such that for all $a, b, b' \in V$

$$|g(a,b) - g(a,b')|_1 \le L\lambda^{(b|b')_a}$$

(vii) There exists a constant $v \in (0, 1)$ such that for all $a, a', b \in V$ if

- $d(a,b) \ge J + 1$ and $d(a',b) \ge J + 1$
- $|d(a,b) d(a',b)| \leq J I$
- $(a|b)_{a'} \leq J$ and $(a'|b)_a \leq J$ then

$$|g(a,b)-g(a',b)|_1\leq 2\nu.$$

where the map $\varphi \colon \mathbb{Q}E \to \mathbb{Q}V$ picks out the end-points of every edge.

Many of the objects in the statement of theorem 4.1 are yet to be defined but all will be explained in this section. At the very end of chapter 4 we will give the proof of the theorem, which will just show where in the section to look for the proof of the relevant facts.

4.1 The first projection

On our uniformly fine δ -hyperbolic graph we want to define a new metric that has better convergence properties than the word metric. To do this we modify the ideas used in [Min01].

In the following picture we consider two points b, b' and a third point a that is comparatively far away. The upper arc perpendicular to [a, b] (blue if you have colour) is the intersection of a sphere around a with the δ -neighbourhood of the geodesic [a, b], whereas the lower arc perpendicular to [a, b'] (red) is the intersection of the sphere around a with the δ -neighbourhood of the geodesic [a, b'].

Fig. 4.1: Overlapping Neighbourhoods



The geodesics may not overlap but by hyperbolicity their δ -neighbourhoods will overlap close to *a*. The idea behind the new metric therefore is to average over the δ -neighbourhoods of the geodesics between the two points. We will need to use the fineness of the graph to ensure we are averaging over something finite. So we average over edges.

For this average to make sense, we work in QE, the Q-vector space with basis *E*. For any finite subset $S \subseteq E$ define the *average of S* to be

$$\operatorname{av}(S) := \frac{1}{|S|} \sum_{s \in S} s \in \mathbb{Q}E.$$

Then in Q*E* we can measure the size of the overlap using the l^1 -norm, which we denote by $|..|_1$, i.e.

$$\left\|\sum_{s\in S}q_s s\right\|_1 := \sum_{s\in S}|q_s|.$$

1

In lemma 2.50 we needed d(a, e) = d(a, e') so we want to keep the distance to *a* constant. Therefore, given $e \in E[a, V]$ we define the *flower of radius r around e with centre a* to be the set

$$\operatorname{Fl}[a,e;r] := E[a,V;d(a,e)] \cap B_r^{\zeta}(e).$$

Similarly, for a vertex $b \in V[a, V]$ we can define the *flower of radius r around b* with centre *a* to be $Fl[a, b; r] := V[a, V; d(a, b)] \cap B_r(b)$.

By taking the average we can turn the flower of radius *r* around *e* with centre *a* into an element of QE[a, V], and so for a fixed vertex *a* and flower radius *r* we get a map $av_{Fl(a;r)} \colon E[a, V] \to QE[a, V]$, which we can extend Q-linearly to get a map $av_{Fl(a;r)} \colon QE[a, V] \to QE[a, V]$.

We do not want to project all the way to the vertex *a* so we need a constant $\aleph > 0$ such that inside the ball of radius \aleph around *a* we should not change *b* much. We will only spread out at certain distances away from *a*, so we need a second constant $\mu > 0$ such that we only spread out when d(a, b) is the initial constant \aleph plus a multiple of the moving constant μ . First, we define a function $f_{\aleph} \colon \mathbb{Q}V \oplus \mathbb{Q}V \to \mathbb{Q}E$ which is moving one step before spreading out.

Definition 4.2. Let \mathcal{G} be a fine, δ -hyperbolic graph. Define a \mathbb{Q} -bilinear map $f_{\mathbb{N}} \colon \mathbb{Q}V \oplus \mathbb{Q}V \to \mathbb{Q}E$ on vertices as follows; Given two vertices a, b of \mathcal{G} either $d(a, b) \leq \mathbb{N}$ or there is an $m \in \mathbb{N}$ such that

$$m\mu + \aleph < d(a,b) \le (m+1)\mu + \aleph$$

and then f_{\aleph} is given by

$$f_{\aleph}(a,b) := \begin{cases} 0 & \text{if } a = b \\ \operatorname{av}(E[a,b;d(a,b)-1]) & \text{if } 1 \le d(a,b) \le \aleph \\ \operatorname{av}(E[a,b;m\mu+\aleph]) & \text{if } m\mu+\aleph < d(a,b) < (m+1)\mu+\aleph \\ \operatorname{av}_{\operatorname{Fl}(a;\delta)}(\operatorname{av}(E[a,b;m\mu+\aleph])) & \text{if } d(a,b) = (m+1)\mu+\aleph \end{cases}$$

The function f_{\aleph} depends on both \aleph and μ . We will keep μ fixed so it is suppressed in the notation. In section 4.4 we will allow \aleph to vary so we want to keep track of it in the notation.

For understanding what the map does it is more intuitive to consider the end-points closest to *a*, namely let $\iota_a : \mathbb{Q}E[a, V] \to \mathbb{Q}V[a, V]$ be the \mathbb{Q} linear extension of the map that given an edge $e \in E[a, V]$ picks out the end-point of *e* that is closest to *a* and consider $\iota_a f_{\otimes}(a, b) \in \mathbb{Q}V$.



If the vertex b does not lie on one of the spheres then we just project to the next sphere by averaging along all of the geodesics from b to a. So b_1 in the picture is sent to the (blue) dots.

The interesting part happens when b lies on one of the spheres, in which case we take the projection as in the previous case but then we also spread out when we reach the next sphere. This is represented in the picture by b_2 being moved to the (orange) bars on the next-smallest sphere.

Remark 4.3. If $a \neq b$ then the element $f_{\aleph}(a, b) \in \mathbb{Q}E$ is a convex combination, i.e. all coefficients are non-negative and their sum is 1. Furthermore, the map f_{\aleph} is invariant under the diagonal Isom(\mathcal{G})-action on $\mathbb{Q}V \times \mathbb{Q}V$.

The map f_{\aleph} is only interesting when $d(a, b) > \aleph$, and the next lemma gives some properties of the support of $f_{\aleph}(a, b)$ in this case.

Lemma 4.4. Let \mathcal{G} be a fine, δ -hyperbolic graph. Suppose $a, b \in V$ have distance $d(a, b) > \aleph$. Pick $m \in \mathbb{N}$ such that

$$n\mu + \aleph < d(a,b) \le (m+1)\mu + \aleph.$$

Then the following statements hold;

- (*i*) $E[a,b;m\mu + \aleph] \subseteq supp(f_{\aleph}(a,b));$
- (*ii*) For any $e \in E[a, b; m\mu + \aleph]$, $supp(f_{\aleph}(a, b)) \subseteq Fl[a, e; 2\delta]$.

Proof. (*i*) Follows immediately from the definition.

(*ii*) Corollary 2.51 gives $d^{\zeta}(e, e') \leq \delta$ for all $e, e' \in E[a, b; m\mu + \aleph]$. In particular, the triangle inequality shows $Fl[a, e'; \delta] \subseteq Fl[a, e; 2\delta]$.

If $d(a,b) < (m+1)\mu + \aleph$ then $f_{\aleph}(a,b) = \operatorname{av}(E[a,b;m\mu + \aleph])$ and so

$$\operatorname{supp}(f_{\aleph}(a,b)) = E[a,b;m\mu + \aleph] \subseteq \operatorname{Fl}[a,e;2\delta].$$

If $d(a,b) = (m+1)\mu + \aleph$ then $f_{\aleph}(a,b) = \operatorname{av}_{\operatorname{Fl}(a;\delta)}(\operatorname{av}(E[a,b;m\mu+\aleph]))$. So

$$\operatorname{supp}(f_{\aleph}(a,b)) = \bigcup_{\bar{e} \in E[a,b;m\mu+\aleph]} \operatorname{Fl}[a,\bar{e};\delta] \subseteq \operatorname{Fl}[a,e;2\delta]. \qquad \Box$$

We will iterate this projection via ι_a so it will be useful to also know what the support of $\iota_a f_{\aleph}(a, b)$ is like.

Corollary 4.5. Suppose we have the same set-up as in lemma 4.4. Then

- (*i*) $V[a, b; m\mu + \aleph] \subseteq supp(\iota_a f_{\aleph}(a, b));$
- (*ii*) For any $v \in V[a, b; m\mu + \aleph]$, $supp(\iota_a f_\aleph(a, b)) \subseteq Fl[a, v; 2\delta + 2]$.

Proof. Use lemma 4.4 and corollary 2.49.

So we know that the support is never too far away from any geodesic from *a* to *b*.

The spreading out in the fourth case in the definition is done to create an overlap, as in figure 4.1. If two vertices *b*, *c* are close then we want the supports of $\iota_a f_{\aleph}(a, b)$ and $\iota_a f_{\aleph}(a, c)$ to remain close, and furthermore we want the intersection of the supports to be non-empty.

For there to be a global bound we assume G is uniformly fine, in which case the balls in the edge-metric d^{ζ} are uniformly finite, by lemma 2.47, and we set

$$B_{\delta}^{\zeta} = \max\left\{\left|B_{\delta}^{\zeta}(e)\right| : e \in E\right\}.$$

Lemma 4.6. Let G be a fine, δ -hyperbolic graph. Let $a, b, c \in V$ be three vertices satisfying

- $\exists m \in \mathbb{N}$ such that $d(a,b) = (m+1)\mu + \aleph = d(a,c);$
- $d(b,c) \leq 2\mu$.

Then for all $e_b \in supp(f_{\aleph}(a, b))$ and for all $e_c \in supp(f_{\aleph}(a, c))$,

$$d^{\zeta}(e_b,e_c)\leq 3\delta$$

In particular $d(\iota_a e_b, \iota_a e_c) \leq 3\delta + 2$. Moreover, if \mathcal{G} is uniformly fine then there exists a constant $\eta_0 \in (0, 1)$ such that

$$|f_{\aleph}(a,b) - f_{\aleph}(a,c)|_1 \le 2\eta_0.$$

Proof. By the definition of f_{\aleph} , there is an edge $\bar{e}_b \in E[a, b; m\mu + \aleph]$ such that $e_b \in \operatorname{Fl}[a, \bar{e}_b; \delta]$. Pick \bar{e}_c analogously. Under our assumptions on the distances between a, b, c we know that $(b|c)_a \ge m\mu + \aleph$. Hence lemma 2.50 tells us that $d^{\zeta}(\bar{e}_b, \bar{e}_c) \le \delta$. Thus

$$d^{\zeta}(e_b,e_c) \leq d^{\zeta}(e_b,\bar{e}_b) + d^{\zeta}(\bar{e}_b,\bar{e}_c) + d^{\zeta}(\bar{e}_c,e_c) \leq 3\delta.$$

In particular corollary 2.49 gives $d(\iota_a e_b, \iota_a e_c) \leq 3\delta + 2$.

Now suppose that \mathcal{G} is uniformly fine (hence B_{δ}^{ζ} is finite). To simplify the notation a bit, set $E_{ab} = E[a, b; m\mu + \aleph]$, and for $e \in E_{ab}$ set $Fl_e = Fl[a, e; \delta]$. Similarly we can define E_{ac} and $Fl_{\overline{e}}$ for $\overline{e} \in E_{ac}$.

Then by definition we have

$$f_{\aleph}(a,b) = \frac{1}{|E_{ab}|} \sum_{e \in E_{ab}} \frac{1}{|\mathrm{Fl}_e|} \sum_{e' \in \mathrm{Fl}_e} e' \quad , \quad f_{\aleph}(a,c) = \frac{1}{|E_{ac}|} \sum_{\bar{e} \in E_{ac}} \frac{1}{|\mathrm{Fl}_{\bar{e}}|} \sum_{\bar{e}' \in \mathrm{Fl}_{\bar{e}}} \bar{e'}.$$

For any $e \in E_{ab}$ and $\bar{e} \in E_{ac}$ we have $d^{\zeta}(e, \bar{e}) \leq \delta$ since $(b|c)_a \geq m\mu + \aleph$. So $\operatorname{Fl}_{\bar{e}} \cap \operatorname{Fl}_{\bar{e}} \neq \emptyset$ and therefore

$$\begin{split} |f_{\aleph}(a,b) - f_{\aleph}(a,c)|_{1} \\ &\leq \frac{1}{|E_{ab}|} \sum_{e \in E_{ab}} \frac{1}{|FI_{e}|} \sum_{e' \in FI_{e}} \frac{1}{|E_{ac}|} \sum_{\bar{e} \in E_{ac}} \frac{1}{|FI_{\bar{e}}|} \sum_{\bar{e}' \in FI_{\bar{e}}} |e' - \bar{e}'|_{1} \quad (4.1) \\ &= \frac{1}{|E_{ab}|} \sum_{e \in E_{ab}} \frac{1}{|E_{ac}|} \sum_{\bar{e} \in E_{ac}} 2\frac{|FI_{e}||FI_{\bar{e}}| - |FI_{e} \cap FI_{\bar{e}}|}{|FI_{e}||FI_{\bar{e}}|} \\ &\leq \frac{1}{|E_{ab}|} \sum_{e \in E_{ab}} \frac{1}{|E_{ac}|} \sum_{\bar{e} \in E_{ac}} 2\left(1 - \frac{1}{(B_{\delta}^{\zeta})^{2}}\right) \\ &= 2\left(1 - \frac{1}{(B_{\delta}^{\zeta})^{2}}\right) \end{split}$$

and the proof is finished by setting $\eta_0 = 1 - \frac{1}{(B_{\lambda}^{\xi})^2} \in (0, 1)$.

4.2 Iterated projections

Using f_{\aleph} we can project to the next sphere of radius $m\mu + \aleph$ around *a*. We can iterate these projections in the natural way.

Definition 4.7. Let \mathcal{G} be a fine, δ -hyperbolic graph. Given $k \in \mathbb{N}$ define $f_{\mathbb{N}}^{(k)} : \mathbb{Q}V \oplus \mathbb{Q}V \to \mathbb{Q}E$ to be the Q-bilinear function that is defined inductively on vertices by

- $f_{\aleph}^{(0)}(a,b) := f_{\aleph}(a,b);$
- $f_{\aleph}^{(k+1)}(a,b) := f_{\aleph}^{(k)}(a,\iota_a f_{\aleph}(a,b)).$

Remark 4.8. As with $f_{\aleph}(a, b)$, for $a \neq b$ the element $f_{\aleph}^{(k)}(a, b) \in \mathbb{Q}E$ is a convex combination, in fact it is a convex combination in $\mathbb{Q}E[a, V]$ so we can always consider $\iota_a f_{\aleph}^{(k)}(a, b)$. Furthermore, the map $f_{\aleph}^{(k)}$ is invariant under the action of Isom(\mathcal{G}).

This recursive definition has an equivalent form, given in the following lemma.

Lemma 4.9. *Let* G *be a fine,* δ *-hyperbolic graph. For all* $a, b \in V$ *and for all* $k \in \mathbb{N}$

$$f_{\aleph}^{(k+1)}(a,b) = f_{\aleph}(a,\iota_a f_{\aleph}^{(k)}(a,b)).$$

Proof. Induct on *k*.

The case k = 0 follows straight from the definition of the map $f_{\aleph}^{(0)}$. For the inductive step;

$$f_{\aleph}^{(k+1)}(a,b) = f_{\aleph}^{(k)}(a,\iota_a f_{\aleph}(a,b))$$

= $f_{\aleph}(a,\iota_a f_{\aleph}^{(k-1)}(a,\iota_a f_{\aleph}(a,b)))$
= $f_{\aleph}(a,\iota_a f_{\aleph}^{(k)}(a,b))$

where the first equality uses the definition of $f_{\aleph}^{(k+1)}$, the second equality is the inductive hypothesis and the third equality is the definition of $f_{\aleph}^{(k)}$. \Box

From now on we will have to make an assumption about the constant μ , which is how far we move in each step of the recursive function $f_{\aleph}^{(k)}$. Since we will later introduce more constants and more conditions we will start a list of all the assumptions we are making which we can add to later on.

Assumptions 4.10. We assume that the constant μ always satisfies the inequality

$$2\mu \ge 3\delta + 2. \tag{4.2}$$

The right-hand side of inequality (4.2) is chosen so that we can iterate lemma 4.6. We know that $3\delta + 2$ bounds $d(\iota_a e_b, \iota_a e_c)$ with e_b in the support of $f_{\aleph}(a, b)$ and e_c in the support of $f_{\aleph}(a, c)$ (for suitable vertices $a, b, c \in V$), but in lemma 4.6 we assume $d(b, c) \leq 2\mu$, so to iterate the lemma we have to assume μ satisfies inequality (4.2).

As with f_{\aleph} it is helpful to have an idea what the support of $f_{\aleph}^{(k)}(a, b)$ is like. This leads to the following lemma (cf lemma 4.4).

Lemma 4.11. Let \mathcal{G} be a fine, δ -hyperbolic graph. Suppose $a, b \in V$ have distance $d(a, b) > \aleph$. Pick $m \in \mathbb{N}$ such that

$$m\mu + \aleph < d(a, b) \le (m+1)\mu + \aleph$$

Then for all $k \in \{0, \ldots, m\}$;

- (*i*) $E[a,b;(m-k)\mu + \aleph] \subseteq supp(f_{\aleph}^{(k)}(a,b));$
- (*ii*) For any $e \in E[a, b; (m k)\mu + \aleph]$, $supp(f_{\aleph}^{(k)}(a, b)) \subseteq Fl[a, e; 3\delta]$.

Proof. We induct on *k*.

The case k = 0 is lemma 4.4.

Now for the inductive step.

(*i*) Assume $E[a, b; (m - k)\mu + \aleph] \subseteq \operatorname{supp}(f_{\aleph}^{(k)}(a, b))$. We want to look at the support of $f_{\aleph}^{(k+1)}(a, b)$. For any $e \in E[a, b; (m - k)\mu + \aleph]$ we know that any geodesic from *a* to *b* restricts to a geodesic from *a* to $\iota_a e$. In particular

$$E[a,b;(m-(k+1))\mu+\aleph] \subseteq E[a,\iota_ae;(m-(k+1))\mu+\aleph].$$

Lemma 4.4 shows

$$E[a,\iota_a e; (m - (k+1))\mu + \aleph] \subseteq \operatorname{supp}(f_{\aleph}(a,\iota_a e))$$

but by the alternative inductive definition of $f_{\aleph}^{(k+1)}$ given in lemma 4.9 we know that $\operatorname{supp}(f_{\aleph}(a,\iota_a e)) \subseteq \operatorname{supp}(f_{\aleph}^{(k+1)}(a,b))$ and bringing this altogether we obtain

$$E[a, \iota_a e; (m - (k+1))\mu + \aleph] \subseteq \operatorname{supp}(f_{\aleph}^{(k+1)}(a, b)).$$

(*ii*) Fix $e \in E[a, b; (m - (k + 1))\mu + \aleph]$. We need to show that for any edge $e_1 \in \text{supp}(f_{\aleph}^{(k+1)}(a, b))$ we have $d^{\zeta}(e, e_1) \leq 3\delta$. By the definition of $f_{\aleph}^{(k+1)}$ there is some $e'_1 \in \text{supp}(f_{\aleph}^{(k)}(a, b))$ with $e_1 \in \text{supp}(f_{\aleph}(a, \iota_a e'_1))$. Pick $e' \in E[a, b; (m - k)\mu + \aleph]$ such that there is a geodesic from *a* to *b* that contains both *e* and *e'*. Then by part (*i*) we know $e' \in \text{supp}(f_{\aleph}^{(k)}(a, b))$.

contains both *e* and *e'*. Then by part (*i*) we know $e' \in \text{supp}(f_{\aleph}^{(k)}(a, b))$. The inductive hypothesis says $d^{\zeta}(e', e'_1) \leq 3\delta$, and then corollary 2.49 gives $d(\iota_a e', \iota_a e'_1) \leq 3\delta + 2 \leq 2\mu$. So lemma 4.6 tells us that $d^{\zeta}(e, e_1) \leq 3\delta$, as desired.

If we start with two vertices b, c then we already know that f_{\aleph} creates an overlap if d(b, c) is small, and we hope that by iterating this overlap becomes larger, in other words we want $|f_{\aleph}^{(k)}(a, b) - f_{\aleph}^{(k)}(a, c)|_1$ to get smaller. This is formalised in the following lemma (cf lemma 4.6).

Lemma 4.12. Let G be a fine, δ -hyperbolic graph. Let $a, b, c \in V$ be three vertices satisfying

- $\exists m \in \mathbb{N}$ such that $d(a,b) = (m+1)\mu + \aleph = d(a,c);$
- $d(b,c) \leq 2\mu$.

For any
$$k \in \{0, ..., m\}$$
, $x \in supp\left(\iota_a f_{\aleph}^{(k)}(a, b)\right)$, and $y \in supp\left(\iota_a f_{\aleph}^{(k)}(a, c)\right)$,
 $d(x, y) < 3\delta + 2.$

Moreover, if G is uniformly fine then

$$|f_{\aleph}^{(k)}(a,b) - f_{\aleph}^{(k)}(a,c)|_1 \le 2\eta_0^k$$

where $\eta_0 \in (0, 1)$ is the constant from lemma 4.6.

Proof. We induct on *k*.

The case k = 0 is exactly lemma 4.6.

For the inductive step, use the alternative recursive definition of $f_{\aleph}^{(k+1)}$ given in lemma 4.9, which says $f_{\aleph}^{(k+1)}(a,b) = f_{\aleph}(a,\iota_a f_{\aleph}^{(k)}(a,b))$. So for any $x \in \operatorname{supp}(\iota_a f_{\aleph}^{(k+1)}(a,b))$ there is some $b_x \in \operatorname{supp}(\iota_a f_{\aleph}^{(k)}(a,b))$ such that $x \in \text{supp}(\iota_a f_{\aleph}^{(k+1)}(a, b_x))$. Similarly for $y \in \text{supp}(\iota_a f_{\aleph}^{(k+1)}(a, c))$ there is a corresponding $c_y \in \text{supp}(\iota_a f_{\aleph}^{(k)}(a, c))$. By the inductive hypothesis $d(b_x, c_y) \leq 3\delta + 2 \leq 2\mu$. Thus we can apply

lemma 4.6 to deduce that $d(x, y) \leq 3\delta + 2$.

Moreover, suppose G is uniformly fine. We keep the notation as in the proof of lemma 4.6. By definition $f_{\aleph}^{(k+1)}(a,b) = f_{\aleph}^{(k)}(a,\iota_a f_{\aleph}(a,b))$. Using the Q-linearity of $f_{\aleph}^{(k)}$ we can proceed analogously to the proof of lemma 4.6 and we get

$$|f_{\aleph}^{(k+1)}(a,b) - f_{\aleph}^{(k+1)}(a,c)|_{1} \leq \frac{1}{|E_{ab}|} \sum_{e \in E_{ab}} \frac{1}{|FI_{e}|} \sum_{e' \in FI_{e}} \frac{1}{|E_{ac}|} \sum_{\bar{e} \in E_{ac}} \frac{1}{|FI_{\bar{e}}|} \sum_{\bar{e}' \in FI_{\bar{e}}} |f_{\aleph}^{(k)}(a,\iota_{a}e') - f_{\aleph}^{(k)}(a,\iota_{a}\bar{e}')|_{1}.$$

Lemma 4.6 also tells us that $d(\iota_a e', \iota_a \bar{e}') \leq 3\delta + 2 \leq 2\mu$, hence the inductive hypothesis gives

$$|f_{\aleph}^{(k)}(a,\iota_a e') - f_{\aleph}^{(k)}(a,\iota_a \bar{e}')|_1 \le \eta_0^k$$

and therefore

$$\begin{aligned} f_{\aleph}^{(k+1)}(a,b) - f_{\aleph}^{(k+1)}(a,c)|_{1} \\ &\leq \frac{1}{|E_{ab}|} \sum_{e \in E_{ab}} \frac{1}{|\mathrm{Fl}_{e}|} \sum_{e' \in \mathrm{Fl}_{e}} \frac{1}{|E_{ac}|} \sum_{\bar{e} \in E_{ac}} \frac{1}{|\mathrm{Fl}_{\bar{e}}|} \sum_{\bar{e}' \in \mathrm{Fl}_{\bar{e}}} |e' - \bar{e}'|_{1} \eta_{0}^{k} \end{aligned}$$

but this is just the sum (4.1) multiplied by the constant η_0^k . Hence we conclude

$$|f_{\aleph}^{(k+1)}(a,b) - f_{\aleph}^{(k+1)}(a,c)|_1 \le 2\eta_0^{k+1}$$

which finishes the induction.

4.3 The total projection

Now we consider what happens when we repeatedly project until we reach the ball of radius \aleph around *a*.

Definition 4.13. Let \mathcal{G} be a fine, δ -hyperbolic graph. Define a Q-linear map $g_{\aleph} : \mathbb{Q}V \oplus \mathbb{Q}V \to \mathbb{Q}E$ on vertices as follows;

For all $a, b \in V$, if $d(a, b) > \aleph$ then there exists an $m \in \mathbb{N}$ such that

$$m\mu + \aleph < d(a,b) \le (m+1)\mu + \aleph$$

and then

$$g_{\aleph}(a,b) := \left\{ egin{array}{ll} f_{\aleph}(a,b) & ext{if } d(a,b) \leq \aleph \ f_{\aleph}^{(m)}(a,b) & ext{if } d(a,b) > \aleph \end{array}
ight.$$

Remark 4.14. The element $g_{\aleph}(a, b) \in \mathbb{Q}E$ is a convex combination if $a \neq b$, and the map g_{\aleph} is invariant under the action of $\text{Isom}(\mathcal{G})$.

Previous results about $f_{\aleph}^{(k)}(a, b)$ immediately carry over to $g_{\aleph}(a, b)$. In particular lemma 4.11 gives;

Lemma 4.15. Let \mathcal{G} be a fine, δ -hyperbolic graph. Suppose $a, b \in V$ have distance $d(a, b) > \aleph$. Pick $m \in \mathbb{N}$ such that

$$m\mu + \aleph < d(a, b) \le (m+1)\mu + \aleph.$$

Then the following statements hold;

- (*i*) $E[a,b;\aleph] \subseteq supp(g_{\aleph}(a,b));$
- (*ii*) For any $e \in E[a, b; \aleph]$, supp $(g_{\aleph}(a, b)) \subseteq Fl[a, e; 3\delta]$.

Furthermore, lemma 4.12 yields the following lemma;

Lemma 4.16. Let G be a fine, δ -hyperbolic graph. Let $a, b, c \in V$ be three vertices satisfying

- $\exists m \in \mathbb{N}$ such that $d(a, b) = (m+1)\mu + \aleph = d(a, c)$;
- $d(b,c) \leq 2\mu$.

Then for all $x \in supp(\iota_a g_{\aleph}(a, b))$, and all $y \in supp(\iota_a g_{\aleph}(a, c))$,

$$d(x,y) \le 3\delta + 2.$$

Moreover, if G is uniformly fine then

$$|g_{\aleph}(a,b) - g_{\aleph}(a,c)|_1 \le 2\eta_0^m$$

where $\eta_0 \in (0, 1)$ is the constant from lemma 4.6.

Up to now we have been assuming that $d(b, c) \leq 2\mu$, and that they both lie on a sphere of radius $(m + 1)\mu + \aleph$ around *a*. We want to drop these conditions on the vertices *b*, *c* to get a general result about what happens when the second variable of g_{\aleph} is changed.

However, we will have to assume a bit more about the constant μ . So we have to update the assumption 4.10 to the following;

Assumptions 4.17. We assume that the constant μ always satisfies the inequality

$$2\mu \ge 7\delta + 4. \tag{4.3}$$

If we consider lemma 4.12 with b = c then we know that the diameter of the support of $\iota_a f_{\aleph}^{(k)}(a, b)$ is bounded by $3\delta + 2$. Hence if we have three vertices $a, b, b' \in V$ and know that there is some vertex x_0 in the support of $\iota_a f_{\aleph}^{(k)}(a, b)$ and some x'_0 in the support of $\iota_a f_{\aleph}^{(k)}(a, b')$ with $d(x_0, x'_0) \leq \delta$ then we can bound the distance between an arbitrary element in the support of $\iota_a f_{\aleph}^{(k)}(a, b)$ and an arbitrary element in the support of $\iota_a f_{\aleph}^{(k)}(a, b)$ and an arbitrary element in the support of $\iota_a f_{\aleph}^{(k)}(a, b)$ and an arbitrary element in the support of $\iota_a f_{\aleph}^{(k)}(a, b)$ by $7\delta + 4$. This is why we now assume that inequality (4.3) holds.

Proposition 4.18. Let \mathcal{G} be a uniformly fine, δ -hyperbolic graph. There are constants L > 0, $\lambda \in (0, 1)$ such that for all $a, b, b' \in V$,

$$|g_{\aleph}(a,b) - g_{\aleph}(a,b')|_1 \leq L\lambda^{(b|b')_a}.$$

Proof. Set $L = \max\left\{2\eta_0^{-\frac{\aleph}{\mu}}, \eta_0^{-1-\frac{\aleph}{\mu}}\right\}$ and $\lambda = \eta_0^{\frac{1}{\mu}} \in (0,1)$ where η_0 is the constant from lemma 4.6.

First consider the case $(b|b')_a \leq \aleph$. Here we have

$$|g_{\aleph}(a,b) - g_{\aleph}(a,b')|_{1} \leq |g_{\aleph}(a,b)|_{1} + |g_{\aleph}(a,b')|_{1}$$
$$\leq 2$$
$$= 2\eta_{0}^{-\frac{\aleph}{\mu}}\eta_{0}^{\frac{\aleph}{\mu}}$$
$$\leq 2\eta_{0}^{-\frac{\aleph}{\mu}}\eta_{0}^{\frac{(b|b')a}{\mu}}$$
$$\leq I \lambda^{(b|b')a}$$

and so the proposition is true in this special case.

Now consider the case $(b|b')_a > \aleph$. Then there is some $n \in \mathbb{N}$ such that $n\mu + \aleph < (b|b')_a \le (n+1)\mu + \aleph$.

There is also some $m \in \mathbb{N}$ such that $m\mu + \aleph < d(a,b) \le (m+1)\mu + \aleph$. So $m \ge n$ since $d(a,b) \ge (b|b')_a$. Set k = m - n and $S = \operatorname{supp}(\iota_a f_{\aleph}^{(k)}(a,b))$. Write $\iota_a f_{\aleph}^{(k)}(a,b) = \sum_{x \in S} \alpha_x x$ and then $g_{\aleph}(a,b) = \sum_{x \in S} \alpha_x g_{\aleph}(a,x)$. Observe that for $x \in S$ we have $d(a,x) = n\mu + \aleph$. Similarly for b' we find $m' \ge n$, and analogously define k', S', and $\alpha'_{x'}$ for $x' \in S'$. Again, observe that $d(a, x') = n\mu + \aleph$ for all $x' \in S'$.

We wish to apply lemma 4.16 to a, x, x' with $x \in S$ and $x' \in S'$ but we need a bound on the distance between x and x'.

Fix a vertex $x_0 \in V[a, b; n\mu + \aleph]$ and a vertex $x'_0 \in V[a, b'; n\mu + \aleph]$. Then $x_0 \in S$ and $x'_0 \in S'$ using lemma 4.11(*i*). Moreover, $d(x_0, x'_0) \leq \delta$ by considering a geodesic triangle with corners a, b, b' and that contains the two vertices x_0, x'_0 (this uses the fact $d(a, x) = d(a, x') \leq (b|b')_a$). So for any $x \in S$ and any $x' \in S'$, using the triangle-inequality together with lemma 4.12 gives

$$d(x, x') = d(x, x_0) + d(x_0, x'_0) + d(x'_0, x')$$

$$\leq 3\delta + 2 + \delta + 3\delta + 2$$

$$= 7\delta + 4$$

$$\leq 2u.$$

.

Thus we can apply lemma 4.16 to get

$$|g_{\aleph}(a,x) - g_{\aleph}(a,x')|_1 \le \eta_0^n.$$

Therefore

$$\begin{aligned} |g_{\aleph}(a,b) - g_{\aleph}(a,b')|_{1} &= \left| \sum_{x \in S} \alpha_{x} g_{\aleph}(a,x) - \sum_{x' \in S'} \alpha'_{x'} g_{\aleph}(a,x') \right|_{1} \\ &\leq \sum_{x \in S} \sum_{x' \in S'} \alpha_{x} \alpha'_{x'} |g_{\aleph}(a,x) - g_{\aleph}(a,x')|_{1} \\ &\leq \sum_{x \in S} \sum_{x' \in S'} \alpha_{x} \alpha'_{x'} \eta_{0}^{n} \\ &= \eta_{0}^{n} \\ &\leq \eta_{0}^{-1 - \frac{\aleph}{\mu}} \eta_{0}^{\frac{(b|b')_{a}}{\mu}} \\ &\leq L \lambda^{(b|b')_{a}}. \end{aligned}$$

4.4 Varying the initial sphere

Proposition 4.18 gives us some control over what happens when we change the second variable of g_{\aleph} , but we still have no control over what happens when we change the first variable. The support of $g_{\aleph}(a,b)$ (for d(a,b) large) is contained in the sphere of radius \aleph around a. Even if a' is

close to *a* there need not be a large overlap between $E[a, V; \aleph]$ and $E[a', V; \aleph]$ To compensate for this, we take an average over many different initial spheres.

So we introduce two new constants to designate the start and end of the range over which we will average. We will use *I* for the radius of the smallest initial sphere to be considered, and *J* for the largest. We also need a third constant *R* that will be the radius of the balls in the edge metric over which we average each $g_{\aleph}(a, b)$. This is made formal in the following definition.

Definition 4.19. Let \mathcal{G} be a fine, δ -hyperbolic graph. Define the Q-linear map $g: \mathbb{Q}V \oplus \mathbb{Q}V \to \mathbb{Q}E$ on vertices $a, b \in V$ by

$$g(a,b) := \frac{1}{1+J-I} \sum_{\aleph=I}^{J} \operatorname{av}_{B_{\mathcal{R}}^{\zeta}}(g_{\aleph}(a,b)).$$

Some facts about $g_{\aleph}(a, b)$ carry straight to g(a, b). For example, g(a, b) is a convex combination if $a \neq b$, the map g is invariant under the action of Isom(\mathcal{G}), and lemma 4.15 gives the following lemma about g(a, b).

Lemma 4.20. Let \mathcal{G} be a uniformly fine, δ -hyperbolic graph. Let $a, b \in V$ be two distinct vertices of \mathcal{G} with d(a,b) > J. For every \aleph with $I \leq \aleph \leq J$ pick an edge $e_{\aleph} \in E[a,b;\aleph]$. Then

- (*i*) Every e_{\aleph} is contained in the support of g(a, b);
- (*ii*) For any $e \in supp(g(a, b))$ there exists some \aleph between I and J such that $d^{\zeta}(e, e_{\aleph}) \leq 3\delta + R$.

The support of g(a, b) is not necessarily contained in E[a, V] so we cannot apply ι_a to it. Recall that in the definition of a graph (see definition 2.1) we included the existence of two maps $\varphi_{\pm} : E \to V$ that pick out the endpoints of an edge. These maps can be Q-linearly extended and then we can take the average of the two end-points of an edge, to wit; let $\varphi : QE \to QV$ be the Q-linear map defined by

$$arphi(e):=rac{1}{2}arphi_+(e)+rac{1}{2}arphi_-(e).$$

Then the composition φg is a Q-bilinear map $\mathbb{Q}V \oplus \mathbb{Q}V \to \mathbb{Q}V$.

Corollary 4.21. Let \mathcal{G} be a uniformly fine, δ -hyperbolic graph. Let $a, b \in V$ be two distinct vertices of \mathcal{G} with d(a, b) > J. For every \aleph with $I \leq \aleph \leq J$ pick a vertex $u_{\aleph} \in V[a, b; \aleph]$. Then

- (*i*) Every u_{\aleph} is in the support of $\varphi g(a, b)$;
- (*ii*) For every $x \in supp(\varphi g(a, b))$ there exists some \aleph between I and J such that $d(x, u_{\aleph}) \leq 3\delta + R + 2 =: C_0$;

(iii) $diam(supp(\varphi g(a, b))) \le 2C_0 + J - I.$

Proof. The first two statements follow from lemma 4.20. The third statement follows from the second statement. \Box

And since $\operatorname{av}_{B_R^{\zeta}}$ is Q-linear we can use proposition 4.18 to bound how much g(a, b) changes with b.

Proposition 4.22. *Let* G *be a uniformly fine,* δ *-hyperbolic graph. There are constants* L > 0, $\lambda \in (0, 1)$ *such that for all a, b, b'* $\in V$,

$$|g(a,b) - g(a,b')|_1 \le L\lambda^{(b|b')_a}$$

Proof. Proposition 4.18 tell us that $|g_{\aleph}(a, b) - g_{\aleph}(a, b')|_1 \leq L\lambda^{(b|b')_a}$ for any \aleph and then, using the Q-linearity of $av_{B_{\lambda}^{\zeta}}$ we get

$$\begin{aligned} \left| \operatorname{av}_{B_{R}^{\zeta}} g_{\aleph}(a,b) - \operatorname{av}_{B_{R}^{\zeta}} g_{\aleph}(a,b') \right|_{1} &= \left| \operatorname{av}_{B_{R}^{\zeta}} (g_{\aleph}(a,b) - g_{\aleph}(a,b')) \right|_{1} \\ &= |g_{\aleph}(a,b) - g_{\aleph}(a,b')|_{1} \\ &\leq L \lambda^{(b|b')_{a}} \end{aligned}$$

Then g(a, b) - g(a, b') is an average of the $\operatorname{av}_{B_R^{\zeta}} g_{\aleph}(a, b) - \operatorname{av}_{B_R^{\zeta}} g_{\aleph}(a, b')$ so its l^1 -norm is also bounded by $L\lambda^{(b|b')_a}$.

For what follows we will need assumptions on the constants R and I. Hence we add to the assumption 4.17 to now assume the following.

Assumptions 4.23. We assume the constants μ , R, and I are chosen to satisfy the following inequalities;

$$2\mu \ge 7\delta + 4$$
$$R \ge 4\delta$$
$$I \ge C_0 + 1$$

where $C_0 = 3\delta + R + 2$ is the constant that appears in corollary 4.21(*ii*). We will use the lower bound on R to show that there is an overlap in proposition 4.25. The proof of that proposition will be via induction, for which we need to know that the distance from b to any element in the support of $\varphi g(a, b)$ is strictly less than d(a, b). This fact is the focus of the next lemma, and needs the lower bound on I.

Lemma 4.24. Let G be a uniformly fine, δ -hyperbolic graph. If $a, b \in V$ satisfy d(a, b) > J + 1 then for all $x \in supp(\varphi g(a, b))$,

$$d(x,b) \le d(a,b) - 1.$$

Proof. Given $x \in \text{supp}(\varphi g(a, b))$ for any $u_{\aleph} \in V[a, b; \aleph]$ corollary 4.21 tells us that there is some \aleph with $d(x, u_{\aleph}) \leq C_0$ where $C_0 = 3\delta + R + 2$ as in corollary 4.21(*ii*). Then

$$d(x,b) \le d(x,u_{\aleph}) + d(u_{\aleph},b)$$

$$\le C_0 + d(a,b) - d(a,u_{\aleph})$$

$$\le C_0 + d(a,b) - I$$

$$\le d(a,b) - 1.$$

Next we turn to look at what happens when we change the first variable of *g*. We consider a special case (as we did when first considering the effect of changing the second variable). We will need to use hyperbolicity of the triangle $\Delta(a, a', b)$ around the corner *b*, so we assume that the vertices $a, a', b \in V$ satisfy $|d(a, b) - d(a', b)| \leq J - I$ to ensure that we can find some \aleph, \aleph' with $I \leq \aleph, \aleph' \leq J$ such that $d(a, b) - \aleph = d(a', b) - \aleph'$ and we also assume that both $(a|b)_{a'}$ and $(a'|b)_a$ are bounded by *J* so that we may apply hyperbolicity to move from the geodesic [a, b] to the geodesic [a', b].

Proposition 4.25. *Let* G *be a uniformly fine,* δ *-hyperbolic graph. There is a constant* $v \in (0,1)$ *such that for all* $a, a', b \in V$ *, if*

- $d(a,b) \ge J + 1$ and $d(a',b) \ge J + 1$
- $|d(a,b) d(a',b)| \le J I$
- $(a|b)_{a'} \leq J$ and $(a'|b)_a \leq J$

then

$$|g(a,b) - g(a',b)|_1 \le 2\nu.$$

Proof. Without loss of generality $d(a', b) \leq d(a, b)$. So the first assumption becomes $0 \leq d(a, b) - d(a', b) \leq J - I$. Pick an edge $e_0 \in E[a, b; J]$. Then $\operatorname{supp}(g_I(a, b)) \subseteq \operatorname{Fl}[a, e_0; 3\delta]$ by lemma 4.15. The idea behind the proof is to show that there is a corresponding edge $e'_0 \in E[a', b]$ such that $d(e'_0, b) = d(e_0, b) \leq (a|a')_b$ and $I \leq d(a', e'_0) \leq J$. The first condition allows us to bound $d^{\zeta}(e_0, e'_0)$ and the second condition ensures that e'_0 contributes to g(a', b). These two facts lead to a bound on the size of the overlap between g(a, b) and g(a', b).

So pick an edge $e'_0 \in E[b, a'; d(b, e_0)]$. Then $d' := d(a', e'_0) \leq J$ and from the first assumption we also obtain

$$d' = d(a', b) - 1 - d(b, e'_0)$$

= $d(a', b) - 1 - d(b, e_0)$
= $d(a', b) - d(a, b) + d(a, e_0)$
 $\geq -(J - I) + J$
= I

Thus $I \leq d' \leq J$ and $\operatorname{av}_{B_{p}^{\zeta}}(g_{d'}(a', b))$ contributes to g(a', b).

The picture below shows a geodesic triangle with vertices a, a', b.



The (yellow) bars represent the union of the supports of the $g_{\aleph}(a, b)$ and $g_{\aleph}(a', b)$ for $I \leq \aleph \leq J$. The dashed (blue) line is part of a sphere around *b* that intersects both (yellow) bars. The thick (green) dots are the inner points of the triangle.

We know that $(a'|b)_a \leq J$ and so $d(e_0, b) \leq (a|a')_b$, thus we can apply hyperbolicity (see lemma 2.50) to get $d^{\zeta}(e_0, e'_0) \leq \delta$.

Also, by lemma 4.15, the support of $g_J(a, b)$ is contained in $Fl[a, e_0; 3\delta]$. Hence for all $e \in supp(g_J(a, b))$

$$egin{array}{ll} d^{\zeta}(e,e_0') &\leq d^{\zeta}(e,e_0) + d^{\zeta}(e_0,e_0') \ &\leq & 3\delta &+ & \delta \ &= & 4\delta \end{array}$$

Then for any $e' \in \text{supp}(g_{d'}(a', b))$ we have $e' \in \text{Fl}[a', e'_0; 3\delta]$ (by lemma 4.15) and so $e'_0 \in B^{\zeta}_R(e) \cap B^{\zeta}_R(e')$. Thus there is an overlap between $\text{av}_{B^{\zeta}_R}(g_J(a, b))$ and $\text{av}_{B^{\zeta}_R}(g_{d'}(a', b))$. We bound this analogously to the proof of lemma 4.6.

$$\begin{aligned} \left| \operatorname{av}_{B_{R}^{\zeta}}(g_{J}(a,b)) - \operatorname{av}_{B_{R}^{\zeta}}(g_{d'}(a',b)) \right|_{1} \\ &\leq \sum_{e \in S_{J}} \sum_{e' \in S'_{d'}} \frac{\alpha_{e}}{|B_{R}^{\zeta}(e)|} \frac{\alpha'_{e'}}{|B_{R}^{\zeta}(e')|} \sum_{\bar{e} \in B_{R}^{\zeta}(e)} \sum_{\bar{e}' \in B_{R}^{\zeta}(e')} |\bar{e} - \bar{e}'|_{1} \\ &\leq 2 \left(1 - \frac{1}{(B_{R}^{\zeta})^{2}} \right) \end{aligned}$$
(4.4)

where $S_J := \text{supp}(g_J(a, b)), S'_{d'} := \text{supp}(g_{d'}(a', b))$ and we express $g_J(a, b)$ and $g_{d'}(a', b)$ as the convex combinations

$$g_J(a,b) = \sum_{e \in S_J} \alpha_e e$$
, $g_{d'}(a',b) = \sum_{e' \in S'_{d'}} \alpha'_{e'} e'$.

$$\begin{split} \left|g(a,b) - g(a',b)\right|_{1} \\ &= \left|\frac{1}{1+J-I}\sum_{\aleph=I}^{J}\operatorname{av}_{B_{R}^{\zeta}}(g_{\aleph}(a,b)) - \frac{1}{1+J-I}\sum_{\aleph'=I}^{J}\operatorname{av}_{B_{R}^{\zeta}}(g_{\aleph'}(a',b))\right|_{1} \\ &\leq \frac{1}{(1+J-I)^{2}}\sum_{\aleph,\aleph'=I}^{J}\left|\operatorname{av}_{B_{R}^{\zeta}}g_{\aleph}(a,b) - \operatorname{av}_{B_{R}^{\zeta}}(g_{\aleph'}(a',b))\right|_{1}. \end{split}$$

All these summands are bounded by 2, but the particular summand with $\aleph = J$ and $\aleph' = d'$ can be bounded as in inequality (4.4). Thus

$$\begin{split} \left| g(a,b) - g(a',b) \right|_1 &\leq \frac{1}{(1+J-I)^2} \left(2 \left((1+J-I)^2 - 1 \right) + 2 \left(1 - \frac{1}{(B_R^{\zeta})^2} \right) \right) \\ &= 2 \left(1 - \frac{1}{(1+J-I)^2 (B_R^{\zeta})^2} \right). \end{split}$$

Therefore we set

So

$$\nu = 1 - \frac{1}{(1+J-I)^2 (B_R^{\zeta})^2} \in (0,1)$$

to conclude $|g(a,b) - g(a',b)|_1 \le 2\nu$.

We end the section by proving theorem 4.1, which summarises what has been done in this section.

Proof of Theorem 4.1. (*i*) The map g is invariant by construction, this was explicitly stated after the definition of g in definition 4.19.

(*ii*) The support of g(a, b) contains the sets $E[a, b; \aleph]$ for $I \leq \aleph \leq J$ as stated in lemma 4.20, which followed immediately from lemma 4.15, which itself followed straight from lemma 4.11.

(*iii*) The support of $\varphi g(a, b)$ has distance strictly less than d(a, b) to b by lemma 4.24.

(*v*) The diameter of supp($\varphi g(a, b)$) was bounded in corollary 4.21(*iii*), which is an easy corollary of lemma 4.20.

(*vi*) The bound when varying the second coordinate of *g* was obtained in proposition 4.22, but the bulk of the work behind the proposition came in proposition 4.18 where the bound was obtained for the map g_\aleph with arbitrary initial constant \aleph .

(*vii*) The bound when varying the first coordinate of *g* was obtained in proposition 4.25. \Box

58

5 A New Metric on the Vertex Set

Given a uniformly fine, Gromov hyperbolic graph \mathcal{G} we need to use the map $g: \mathbb{Q}V \oplus \mathbb{Q}V \to \mathbb{Q}E$ from theorem 4.1 to construct a metric on V. In section 5.1 we will define a function $r: \mathbb{Q}V \oplus \mathbb{Q}V \to [0, \infty)$, and prove some properties about this function, which will help for the triangle inequality. Section 5.2 is about the double difference of this function r. Then in section 5.3 we can define the metric \hat{d} on the vertex set V of the graph \mathcal{G} , and state some properties that this metric satisfies.

5.1 A first attempt at a metric

In chapter 4 we constructed the element $\varphi g(a, b) \in \mathbb{Q}V$ for vertices a, b in a fine, Gromov hyperbolic graph \mathcal{G} , by moving b in small steps towards a. This can also be thought of as moving a one step closer to b, and then we can count how many steps it takes to move a to b.

Definition 5.1. Let \mathcal{G} be a uniformly fine, δ -hyperbolic graph. As in chapter 4 we need to fix the constants μ , R, I, and J satisfying the inequalities given as assumptions 4.23. The map $r: \mathbb{Q}V \oplus \mathbb{Q}V \to [0, \infty)$ is defined recursively, given on vertices $a, b \in V$ by

$$r(a,b) := \begin{cases} 0 & \text{if } a = b \\ 1 & \text{if } 0 < d(a,b) \le J+1 \\ 1 + r(\varphi g(a,b),b) & \text{if } d(a,b) > J+1 \end{cases}$$

and then Q-linearly extending. To see that the map is well-defined we need to use theorem 4.1(iii) to say that the distance from *b* to any vertex in the

support of $\varphi g(a, b)$ is strictly less than d(a, b), thus the recursion must stop in $\leq d(a, b)$ steps.

The map $r: \mathbb{Q}V \oplus \mathbb{Q}V \to [0, \infty)$ is our first attempt at a metric. It is positive-definite but not symmetric. However we can symmetrise it later. We want something resembling the triangle-inequality. Since r is not symmetric, we will consider |r(a, b) - r(a', b)| and |r(a, b) - r(a, b')| separately. But before that, we compare the function r to the original metric d on V.

Proposition 5.2. Let G be a uniformly fine, δ -hyperbolic graph. For all vertices $a, b \in V$ we have

$$\frac{1}{J+1}d(a,b) \le r(a,b) \le d(a,b).$$

Proof. Induct on d(a, b).

If a = b then by definition r(a, b) = 0. If $0 < d(a, b) \le J + 1$ then $r(a, b) = 1 \le d(a, b)$ but also

$$\frac{1}{J+1}\,d(a,b) \le \frac{1}{J+1}\,(J+1) = 1.$$

If d(a,b) > J + 1 then $r(a,b) = 1 + r(\varphi g(a,b),b)$. For any x in the support of $\varphi g(a,b)$ theorem 4.1(*iii*) tells us that $d(x,b) \le d(a,b) - 1$, so the inductive hypothesis gives

$$\frac{1}{J+1}d(x,b) \le r(x,b) \le d(x,b).$$

If we write $\varphi g(a, b) = \sum_{x \in V} \alpha_x x$ where $\alpha_x \in [0, 1]$ and $\sum_{x \in V} \alpha_x = 1$, which can be done since $\varphi g(a, b)$ is a convex combination, then

$$r(a,b) = 1 + \sum_{x \in V} \alpha_x r(x,b)$$

$$\leq 1 + \sum_{x \in V} \alpha_x d(x,b)$$

$$\leq 1 + \sum_{x \in V} \alpha_x (d(a,b) - 1)$$

$$= d(a,b).$$

Moreover, for any $x \in \text{supp}(\varphi g(a, b))$,

$$d(x,b) \ge d(a,b) - d(a,x) \ge d(a,b) - J$$

and so

$$\begin{aligned} r(a,b) &= 1 + \sum_{x \in V} \alpha_x r(x,b) \\ &\geq 1 + \sum_{x \in V} \frac{\alpha_x}{J+1} d(x,b) \\ &\geq 1 + \sum_{x \in V} \frac{\alpha_x}{J+1} \left(d(a,b) - J \right) \\ &\geq \frac{1}{J+1} d(a,b). \end{aligned}$$

Remark 5.3. The function $r: \mathbb{Q}V \oplus \mathbb{Q}V \to [0,\infty)$ is invariant under the action of $\operatorname{Isom}(\mathcal{G})$, i.e. if ψ is an isometry of \mathcal{G} (with respect to the word metric *d*) then for all $a, b \in V$

$$r(\psi(a),\psi(b)) = r(a,b).$$

For what follows we need to make more assumptions about the constants *I* and *J*, so we add to assumptions 4.23 to get;

Assumptions 5.4. We assume the constants μ , *R*, *I*, and *J* are chosen to satisfy the following inequalities;

$$2\mu \ge 7\delta + 4$$

 $R \ge 4\delta$
 $I \ge C_0 + \delta + 1$
 $J - I \ge C_0 + \delta + 1 + I$

where C_0 is the constant from theorem 4.1(iv). We have increased the lower bound on *I* and introduced a lower bound on *J*. These new assumptions are there to allow us to show that the distance from *a'* to an element in the support of $\varphi g(a, b)$ is strictly less than d(a, a'), under certain conditions on the vertices *a*, *a'*, *b*. This is for the induction in the proof of lemma 5.5.

Now we turn to the triangle inequality. The next lemma gives a bound for |r(a, b) - r(a', b)| in terms of d(a, a').

Lemma 5.5. Let G be a uniformly fine, δ -hyperbolic graph. There exists a constant $N \ge 0$ such that for all vertices $a, a', b \in V$ we have

$$|r(a,b) - r(a',b)| \le d(a,a') + N.$$

Proof. Without loss of generality, $d(a', b) \le d(a, b)$. Pick $N \ge J + 2$ large enough to satisfy the equation

$$(1 - \nu)N \ge 2\nu(C_0 + J) \tag{5.1}$$

where $\nu \in (0, 1)$ is the constant from theorem 4.1(vii) and C_0 is the constant from theorem 4.1(iv).

The idea behind the proof is to move at least one of a, a' closer to b and use the inductive definition of r, hence we induct over d(a, b) + d(a', b). But first we need a start point for the induction.

If $d(a', b) \leq J + 1$ then $r(a', b) \in \{0, 1\}$ by definition. Thus

$$\begin{aligned} |r(a,b) - r(a',b)| &\leq r(a,b) + r(a',b) \\ &\leq d(a,b) + 1 \\ &\leq d(a,a') + d(a',b) + 1 \\ &\leq d(a,a') + (J+1) + 1 \\ &\leq d(a,a') + N. \end{aligned}$$

Furthermore, if $d(a,b) + d(a',b) \le 2(J+1)$ then $d(a',b) \le J+1$, so for the inductive step we may assume $J + 1 \le d(a',b) \le d(a,b)$. Then by the inductive definition $r(a,b) = 1 + r(\varphi g(a,b),b)$ and similarly for r(a',b).

The idea is to move *a* to $\varphi g(a, b)$, which is closer to *b*, by theorem 4.1(*iii*). So moving *a* to $\varphi g(a, b)$ reduces d(a, b) + d(a', b) and we can apply the inductive hypothesis. However, $\varphi g(a, b)$ need not be closer to *a'*, and may even be further away. Hence we split the proof into three cases;

- *Case* 1 When *a* and *a*' are about the same distance away from *b* and this distance is large in comparison to d(a, a').
- Case 2 When *a* and *a*' are about the same distance away from *b* but d(a, a') is comparatively large.

Case 3 When *a* and *a*′ are not roughly the same distance away from *b*.

In the first case we will move *a* to $\varphi g(a, b)$ and *a'* to $\varphi g(a', b)$ to get closer to *b*. This may increase d(a, a') but we can use theorem 4.1(vii) to compensate for this possibility.

In the second case we will move only *a* to $\varphi g(a, b)$, but using the hyperbolic nature of the graph we can show that we have moved closer to *a'* too.

In the third case, we move only a, and this moves us closer to a' too because a was much further away from b than a' is.

Before tackling these three cases we fix some notation.

Let $\Delta = \Delta([a, b], [a', b], [a, a'])$ be a geodesic triangle with vertices a, a', b. For $I \leq \aleph \leq J$ let u_{\aleph} be the vertex of [a, b] whose distance from a is \aleph , and let u'_{\aleph} be the vertex of [a', b] whose distance from a' is \aleph .

Case 1: Assume the following;

• $d(a,b) - d(a',b) \le J - I$

• $(a|b)_{a'} \leq J$ and $(a'|b)_a \leq J$

i.e., assume that *a* and *a*' are about the same distance away from *b* and that this distance is large in comparison to d(a, a'). We are interested in the expression

$$|r(a,b) - r(a',b)| = |r(\varphi g(a,b),b) - r(\varphi g(a',b),b)| = |r(\varphi (g(a,b) - g(a',b)),b)|.$$

There is some cancellation between g(a, b) and g(a', b), which is forumlated in theorem 4.1(*vii*). To use this cancellation we define two new elements $g_+, g_- \in \mathbb{Q}V$ by:

- $g_+ g_- = g(a, b) g(a', b)$
- all coefficients of g_{\pm} are positive
- $\operatorname{supp}(g_+) \cap \operatorname{supp}(g_-) = \emptyset$

These three conditions determine g_+ and g_- . All we have done is cancel out as much as possible from g(a, b) - g(a', b). In particular, since the support of g_+ is disjoint from the support of g_- we have

$$egin{aligned} |g_+|_1 + |g_-|_1 &= |g_+ - g_-|_1 \ &= |g(a,b) - g(a',b)|_1 \ &\leq 2
u. \end{aligned}$$

Moreover, if we let $\epsilon : \mathbb{Q}V \to \mathbb{Q}$ be the augmentation homomorphism then the non-negativity of the coefficients of g_{\pm} yields $|g_{\pm}|_1 = \epsilon(g_{\pm})$. Then

$$|g_{+}|_{1} - |g_{-}|_{1} = \epsilon(g_{+}) - \epsilon(g_{-})$$

= $\epsilon(g_{+} - g_{-})$
= $\epsilon(g(a, b) - g(a', b))$
= $\epsilon(g(a, b)) - \epsilon(g(a', b))$
= $|g(a, b)|_{1} - |g(a', b)|_{1}$
= 0

Therefore $|g_+|_1 = |g_-|_1$.

Combining the previous two facts gives

$$|g_{\pm}|_1 = \frac{1}{2}(|g_{+}|_1 + |g_{-}|_1) \le \nu.$$

Write $\varphi g_+ = \sum_{x \in V} \alpha_x x$ and $g_- = \sum_{x' \in V} \alpha'_{x'} x'$.

Observe that the support of φg_+ is contained in the support of $\varphi g(a, b)$ and the support of φg_- is contained in the support of $\varphi g(a', b)$ and thus

theorem 4.1(*iii*) tells us that for all $x \in \text{supp}(\varphi g_+)$ and all $x' \in \text{supp}(\varphi g_-)$ we have

$$d(x,b) + d(x',b) \le d(a,b) - 1 + d(a',b) - 1.$$

Hence we can apply the inductive hypothesis to x, x', b to obtain

$$|r(x,b) - r(x',b)| \le d(x,x') + N.$$

Moreover, we know from theorem 4.1(*iv*) that there is some \aleph and some \aleph' between *I* and *J* such that $d^{\zeta}(x, u_{\aleph}) \leq C_0$ and $d^{\zeta}(x', u'_{\aleph'}) \leq C_0$. Therefore

$$|r(x,b) - r(x',b)| \le d(x,u_{\aleph}) + d(u_{\aleph},u'_{\aleph'}) + d(u'_{\aleph'},x') + N \le 2C_0 + (J + d(a,a') + J) + N.$$

Now we have

$$\begin{aligned} |r(a,b) - r(a',b)| &= |r(\varphi g_+,b) - r(\varphi g_-,b)| \\ &= \left| \sum_{x \in V} \alpha_x r(x,b) - \sum_{x' \in V} \alpha'_{x'} r(x',b) \right| \\ &\leq \sum_{x,x' \in V} \frac{\alpha_x \alpha'_{x'}}{|g_{\pm}|_1} |r(x,b) - r(x',b)| \\ &\leq \sum_{x,x' \in V} \frac{\alpha_x \alpha'_{x'}}{|g_{\pm}|_1} (2C_0 + 2J + d(a,a') + N) \\ &= |g_{\pm}|_1 (2C_0 + 2J + d(a,a') + N) \\ &\leq \nu (2C_0 + 2J + d(a,a') + N) \\ &\leq d(a,a') + N \end{aligned}$$

since we chose *N* large enough to satisfy equation (5.1). This finishes case 1.

Case 2: Assume that $J < (a'|b)_a$.

Here the Gromov product is large and so any element in the support of $\varphi g(a, b)$ should be closer to a' than a is, and so we only move a.

Write $\varphi g(a, b) = \sum_{x \in V} \alpha_x x$ with $\sum_{x \in V} \alpha_x = 1$. For any x in the support of $\varphi g(a, b)$ we can find an \aleph such that $d(x, u_\aleph) \leq C_0$. Let v_\aleph be the point on the geodesic [a, a'] with $d(a, v_\aleph) = \aleph = d(a, u_\aleph)$. Since $\aleph \leq J < (a'|b)_a$ hyperbolicity gives $d(u_\aleph, v_\aleph) \leq \delta$. Then the distance from x to a' is;

$$d(x,a') \leq d(x,u_{\aleph}) + d(u_{\aleph},v_{\aleph}) + d(v_{\aleph},a')$$

$$\leq C_0 + \delta + (d(a,a') - d(a,v_{\aleph}))$$

$$= C_0 + \delta + d(a,a') - \aleph$$

$$\leq C_0 + \delta + d(a,a') - I$$

$$\leq d(a,a') - 1.$$
Using theorem 4.1(*iii*) we can apply the inductive hypothesis to x, a', b. Hence;

$$|r(a,b) - r(a',b)| = |1 + r(\varphi g(a,b),b) - r(a',b)|$$

$$\leq 1 + \sum_{x \in V} \alpha_x |r(x,b) - r(a',b)|$$

$$\leq 1 + \sum_{x \in V} \alpha_x (d(x,a') + N))$$

$$\leq 1 + N + \sum_{x \in V} \alpha_x (d(a,a') - 1)$$

$$= d(a,a') + N.$$

This finishes case 2.

Case 3: Assume that |d(a,b) - d(a',b)| > J - I, i.e. assume that *a* is much further away from *b* than *a'* is. Since we have already covered the possibility $J < (a'|b)_a$ in case 2, we may also assume that $(a'|b)_a \le J$.

Write $\varphi g(a, b) = \sum_{x \in V} \alpha_x x$ with $\sum_{x \in V} \alpha_x = 1$. For any x in the support of $\varphi g(a, b)$ take \aleph with $d(x, u_{\aleph}) \leq C_0$. We aim to show that d(x, a') is smaller than d(a, a').

If $\aleph = d(u_\aleph, a) \le (a'|b)_a$ then we can repeat the argument in case 2 to get $d(x, a') \le d(a, a') - 1$.

If $d(u_{\aleph}, a) > (a'|b)_a$ then $d(b, u_{\aleph}) \le (a|a')_b$ and we use hyperbolicity to move to [a', b] and bound how far we need to travel along [a', b] to a'. Let $w_{\aleph} \in [a', b]$ satisfy $d(b, w_{\aleph}) = d(b, u_{\aleph})$. Then hyperbolicity tells us that $d(w_{\aleph}, u_{\aleph}) \le \delta$. Thus

$$d(u_{\aleph}, a') \leq d(u_{\aleph}, w_{\aleph}) + d(w_{\aleph}, a')$$

$$\leq \delta + (d(a', b) - d(b, w_{\aleph}))$$

$$= \delta + d(a', b) - d(b, u_{\aleph})$$

$$= \delta + d(a', b) - (d(a, b) - d(a, u_{\aleph}))$$

$$\leq \delta + d(a', b) - d(a, b) + J$$

$$\leq \delta - (J - I) + J$$

$$= \delta + I$$

Moreover we know

$$d(a, a') \ge |d(a, b) - d(a', b)| > J - I$$

and so

$$d(x,a') \leq d(x,u_{\aleph}) + d(u_{\aleph},a')$$

$$\leq C_0 + (\delta + I)$$

$$\leq (d(a,a') - (J - I)) + C_0 + \delta + I$$

$$\leq d(a,a') - 1.$$

Then we proceed as in case 2 and conclude

$$|r(a,b) - r(a',b)| = d(a,a') + N.$$

This finishes case 3.

That covers all possible cases and therefore we have finished the inductive proof. $\hfill \Box$

The following corollary can be thought of as a means for bounding the size of error terms in later proofs.

Corollary 5.6. Let \mathcal{G} be a uniformly fine, δ -hyperbolic graph. Let $\epsilon : \mathbb{Q}V \to \mathbb{Q}$ be the augmentation homomorphism. There is a constant D > 0 such that for any $b \in V$ and any $z \in \mathbb{Q}V$ if $\epsilon(z) = 0$ then

$$|r(z,b)| \leq D|z|_1 diam(supp(z)).$$

Proof. Let $z = \sum_{v \in V} \alpha_v v$. Then $\sum_{v \in V} \alpha_v = 0$ since z is a cycle. So we can write z in the form $z = \sum_{i \in I} \beta_i (u_i - w_i)$ where $\beta_i > 0$ and $\sum_{i \in I} \beta_i = \frac{1}{2} |z|_1$. Set $D = \frac{1}{2}(1 + N)$, where N is the constant from lemma 5.5. Applying lemma 5.5 gives

$$\begin{split} r(z,b)| &\leq \sum_{i \in I} \beta_i |r(u_i,b) - r(w_i,b)| \\ &\leq \sum_{i \in I} \beta_i (d(u_i,w_i) + N) \\ &\leq \frac{1}{2} |z|_1 (\operatorname{diam}(\operatorname{supp}(z)) + N) \\ &\leq D|z|_1 \operatorname{diam}(\operatorname{supp}(z)). \end{split}$$

The next lemma goes towards proving that geodesics with respect to d will not be not far away from being geodesics with respect to the new metric (which is defined in section 5.3).

Lemma 5.7. Let \mathcal{G} be a uniformly fine, δ -hyperbolic graph. There exists a constant $C_1 \geq 0$ such that if α is a geodesic in \mathcal{G} from $a \in V$ to $b \in V$ then for all vertices b_0 on α and all vertices $\tilde{a} \in N(\alpha_{[a,b_0]}, C_0)$

Proof. We induct on $d(\tilde{a}, b_0)$. Let L > 0 and $\lambda \in (0, 1)$ be the constants from proposition 4.18, N > 0 be the constant from lemma 5.5, and D > 0 be the constant from corollary 5.6. In the induction we will acquire error terms but these error terms become exponentially small as $d(\tilde{a}, b)$ increases, so we take as our inductive hypothesis that

$$|r(\tilde{a},b) - r(\tilde{a},b_0) - r(b_0,b)| \le C_1' \sum_{k=0}^{d(\tilde{a},b_0)} \lambda^k$$

where $C'_1 := \max\{2(J+1) + 4C_0 + N, DL\lambda^{-2C_0}(2C_0 + 2J)\}$. If we can prove this then setting $C_1 = C'_1 \sum_{k \in \mathbb{N}} \lambda^k = C'_1/1 - \lambda$ finishes the proof.

To start the induction, assume $d(\tilde{a}, b_0) \leq J + 1 + 2C_0$. Then

$$\begin{aligned} |r(\tilde{a}, b) - r(\tilde{a}, b_0) - r(b_0, b)| &\leq |r(\tilde{a}, b) - r(b_0, b)| + r(\tilde{a}, b_0) \\ &\leq (d(\tilde{a}, b_0) + N) + d(\tilde{a}, b_0) \\ &\leq 2(J + 1 + 2C_0) + N \\ &\leq C_1' \sum_{k=0}^{d(\tilde{a}, b_0)} \lambda^k \end{aligned}$$

using lemma 5.5, proposition 5.2 and the definition of C'_1 .

For the inductive step assume $d(\tilde{a}, b_0) > J + 1 + 2C_0$. By the inductive definition $r(\tilde{a}, b_0) = 1 + r(\varphi g(\tilde{a}, b_0), b_0)$. Also, there is some vertex \tilde{a}' on $\alpha_{[a,b_0]}$ such that $d(\tilde{a}, \tilde{a}') \leq C_0$. Then $d(\tilde{a}', b) = d(\tilde{a}', b_0) + d(b_0, b)$ and

$$d(\tilde{a}, b) \ge d(\tilde{a}', b) - d(\tilde{a}, \tilde{a}') \ge d(\tilde{a}', b_0) - d(\tilde{a}, \tilde{a}') \ge d(\tilde{a}, b_0) - 2d(\tilde{a}, \tilde{a}')$$

> $J + 1 + 2C_0 - 2C_0 = J + 1$

so $r(\tilde{a}, b) = 1 + r(\varphi g(\tilde{a}, b), b)$. Thus

$$|r(\tilde{a}, b) - r(\tilde{a}, b_0) - r(b_0, b)| = |r(\varphi g(\tilde{a}, b), b) - r(\varphi g(\tilde{a}, b_0), b_0) - r(b_0, b)|$$

$$\leq |r(\varphi g(\tilde{a}, b_0), b) - r(\varphi g(\tilde{a}, b_0), b_0) - r(b_0, b)|$$

$$+ |r(\varphi g(\tilde{a}, b) - \varphi g(\tilde{a}, b_0), b)|.$$

We will bound these two terms separately, using the inductive hypothesis for the first term (the inductive term) and corollary 5.6 for the second term (the error term). We start by considering the inductive term. Write $\varphi g(\tilde{a}, b_0) = \sum_{y \in V} \alpha_y y$ with $\sum_{y \in V} \alpha_y y = 1$. Then

$$|r(\varphi g(\tilde{a}, b_0), b) - r(\varphi g(\tilde{a}, b_0), b_0) - r(b_0, b)| \le \sum_{y \in V} \alpha_y |r(y, b) - r(y, b_0) - r(b_0, b)|$$

Theorem 4.1(*iii*) gives $d(y, b_0) \le d(\tilde{a}, b_0) - 1$ for any $y \in \text{supp}(\varphi g(\tilde{a}, b_0))$ and so we can apply the inductive hypothesis to get

$$\begin{aligned} |r(y,b) - r(y,b_0) - r(b_0,b)| &\leq C_1' \sum_{k=0}^{d(y,b_0)} \lambda^k \\ &\leq C_1' \sum_{k=0}^{d(\tilde{a},b_0)-1} \lambda^k. \end{aligned}$$

We still need to bound the error term $|r(\varphi g(\tilde{a}, b) - \varphi g(\tilde{a}, b_0), b)|$. Applying corollary 5.6 gives

$$\begin{aligned} |r(\varphi g(\tilde{a}, b) - \varphi g(\tilde{a}, b_0), b)| \\ &\leq D|g(\tilde{a}, b) - g(\tilde{a}, b_0)|_1 \operatorname{diam}(\operatorname{supp}(\varphi g(\tilde{a}, b) - \varphi g(\tilde{a}, b_0))). \end{aligned}$$

and by theorem 4.1(vi) we have

$$|g(\tilde{a},b) - g(\tilde{a},b_0)|_1 \le L\lambda^{(b|b_0)_{\tilde{a}}}$$

Hence we need a lower bound for $(b|b_0)_{\tilde{a}}$ and an upper bound for the diameter of the support of $\varphi g(\tilde{a}, b) - \varphi g(\tilde{a}, b_0)$. We consider the Gromov product first.

$$\begin{split} (b|b_0)_{\tilde{a}} &= \frac{1}{2} \left(d(\tilde{a}, b) + d(\tilde{a}, b_0) - d(b, b_0) \right) \\ &\geq \frac{1}{2} \left(d(\tilde{a}', b) - C_0 + d(\tilde{a}', b_0) - C_0 - d(b, b_0) \right) \\ &= d(\tilde{a}', b_0) - C_0 \\ &\geq d(\tilde{a}, b_0) - 2C_0. \end{split}$$

So $\lambda^{(b|b_0)\tilde{a}} \leq \lambda^{d(\tilde{a},b_0)-2C_0}$. We still need an upper bound for the diameter of the support. Theorem 4.1(v) tells us that the diameter of the support of $\varphi g(\tilde{a}, b)$ is bounded by $2C_0 + J - I$, as is the diameter of the support of $\varphi g(\tilde{a}, b_0)$. We can bound the distance from an element in the support of $\varphi g(\tilde{a}, b)$ to an element in the support of $\varphi g(\tilde{a}, b_0)$ by going via the vertex \tilde{a} , hence we get a bound of $2C_0 + 2J$. Therefore

$$|r(\varphi g(\tilde{a},b)-\varphi g(\tilde{a},b_0),b)| \leq DL\lambda^{d(\tilde{a},b_0)-2C_0}(2C_0+2J).$$

Bringing everything together gives

$$\begin{aligned} |r(\tilde{a},b) - r(\tilde{a},b_0) - r(b_0,b)| &\leq C_1' \sum_{k=0}^{d(\tilde{a},b_0)-1} \lambda^k + DL\lambda^{d(\tilde{a},b_0)-2C_0} (2C_0 + 2J) \\ &\leq C_1' \sum_{k=0}^{d(\tilde{a},b_0)} \lambda^k. \end{aligned}$$

This finishes the inductive proof and the lemma follows as given at the start of this proof. $\hfill \Box$

For the triangle inequality we also need to look at |r(a, b) - r(a, b')|.

Proposition 5.8. Let G be a uniformly fine, δ -hyperbolic graph. There exists a constant M > 0 such that for all $a, b, b' \in V$

$$|r(a,b) - r(a,b')| \le M d(b,b').$$

The proof will be by induction but in this induction there will be error terms so it is easier to use a different inductive hypothesis, which is given in the next lemma.

Lemma 5.9. Let \mathcal{G} be a uniformly fine, δ -hyperbolic graph. For any $a, b, b' \in V$ we have

$$|r(a,b) - r(a,b')| \le d(b,b') + 2(J+1) + \frac{1}{2}L\sum_{j=1}^{(b|b')_a} \lambda^j (2j + d(b,b')).$$

The term 2(J + 1) is for the start of the induction, and the sum bounds the size of the error terms that we pick up. Before proving lemma 5.9 we will show how it implies proposition 5.8.

Proof of Proposition 5.8. If d(b,b') = 0 then b = b' and the inequality is trivially satisfied (both sides equal zero). So assume that d(b,b') > 0. In particular $d(b,b') \ge 1$. Set $M = 1 + 2(J+1) + \frac{1}{2}L\sum_{j=1}^{\infty} \lambda^j (2j+1)$, where the infinite sum $\sum_{j=1}^{\infty} \lambda^j (2j+1)$ converges by using the ratio test. Then by lemma 5.9;

$$\begin{aligned} |r(a,b) - r(a,b')| &\leq d(b,b') + 2(J+1) + \frac{1}{2}L\sum_{j=1}^{(b|b')a}\lambda^{j}(2j+d(b,b')) \\ &\leq \left(1 + 2(J+1) + \frac{1}{2}L\sum_{j=1}^{\infty}\lambda^{j}(2j+1)\right)d(b,b') \\ &= Md(b,b'). \end{aligned}$$

We still need to prove lemma 5.9.

Proof of Lemma 5.9. Without loss of generality $d(a,b) \ge d(a,b')$. The proof is by induction on $(b|b')_a$.

If $(b|b')_a \leq J + 1$ then we use proposition 5.2;

$$\begin{aligned} |r(a,b) - r(a,b')| &\leq r(a,b) + r(a,b') \\ &\leq d(a,b) + d(a,b') \\ &= d(b,b') + 2(b|b')_a \\ &\leq d(b,b') + 2(J+1). \end{aligned}$$

This is the starting point for our induction.

So assume that $(b|b')_a > J + 1$. Then $d(a,b) \ge d(a,b') \ge (b|b')_a > J + 1$ and so

$$|r(a,b) - r(a,b')| = |1 + r(\varphi g(a,b),b) - 1 - r(\varphi g(a,b'),b')|$$

= |r(\varphi g(a,b),b) - r(\varphi g(a,b'),b')|.

The idea is to say that since the Gromov product $(b|b')_a$ is large there is a large overlap between g(a, b) and g(a, b'), and for any element in the support of both g(a, b) and g(a, b') we can apply the inductive hypothesis. We still need to worry about the elements in the support of one but not both, however these elements should not contribute much to the final total, and will be swallowed in the sum.

Formally, define new chains g_+ , g_- , g_0 by saying that all their coefficients are positive, and that they satisfy;

- $\operatorname{supp}(g_+) \cap \operatorname{supp}(g_-) = \emptyset$
- $g(a,b) g(a,b') = g_+ g_-$
- $g(a,b) = g_+ + g_0$
- $g(a,b') = g_- + g_0$.

Then g_0 represents the overlap and g_{\pm} represent the parts outside the overlap. Using the linearity of *r* we have

$$\begin{aligned} |r(\varphi g(a,b),b) - r(\varphi g(a,b'),b')| \\ &= |r(\varphi g_{+},b) + r(\varphi g_{0},b) - r(\varphi g_{0},b') - r(\varphi g_{-},b')| \\ &\leq |r(\varphi g_{+},b)| + |r(\varphi g_{0},b) - r(\varphi g_{0},b')| + |r(\varphi g_{-},b')|. \end{aligned}$$

We want to bound these three terms. The middle term is the inductive term. The first and third terms are error terms.

We start with the error terms $|r(\varphi g_+, b)|$ and $|r(\varphi g_-, b')|$. The contribution from these terms should be small because $|g_+|_1$ and $|g_-|_1$ are small. More precisely, by considering the augmentation homomorphism (as in case 1 of the proof of lemma 5.5) we know

$$|g_{+}|_{1} - |g_{-}|_{1} = |g(a, b)|_{1} - |g(a, b')|_{1} = 0$$

and thus $|g_+|_1 = |g_-|_1$. Moreover it follows from theorem 4.1(*vi*) that

$$2|g_{\pm}|_{1} = |g_{+}|_{1} + |g_{-}|_{1} = |g_{+} - g_{-}|_{1} = |g(a, b) - g(a, b')|_{1} \le L\lambda^{(b|b')_{a}}$$

Write $\varphi g_+ = \sum_{x \in V} \alpha_x^+ x$. Theorem 4.1(*iii*) tells us that $d(x, b) \le d(a, b) - 1$ for all $x \in \text{supp}(\varphi g_+) \subseteq \text{supp}(\varphi g(a, b))$. Hence proposition 5.2 gives;

$$|r(\varphi g_+, b)| \le \sum_{x \in V} \alpha_x^+ |r(x, b)|$$
$$\le \sum_{x \in V} \alpha_x^+ d(x, b)$$
$$\le \sum_{x \in V} \alpha_x^+ d(a, b)$$
$$= |g_+|_1 d(a, b)$$

Similarly $|r(\varphi g_{-}, b')| \leq |g_{-}|_{1}d(a, b')$. Therefore

$$\begin{aligned} |r(\varphi g_+, b)| + |r(\varphi g_-, b')| &\leq |g_+|_1 d(a, b) + |g_-|_1 d(a, b') \\ &\leq \frac{1}{2} L \lambda^{(b|b')_a} (d(a, b) + d(a, b')) \\ &= \frac{1}{2} L \lambda^{(b|b')_a} (d(b, b') + 2(b|b')_a). \end{aligned}$$

Now consider the inductive term $|r(\varphi g_0, b) - r(\varphi g_0, b')|$. In barycentric coordinate write $\varphi g_0 = \sum_{y \in V} \alpha_y^0 y$ so

$$|r(\varphi g_0, b) - r(\varphi g_0, b')| \le \sum_{y \in V} \alpha_y^0 |r(y, b) - r(y, b')|.$$

We want to apply the inductive hypothesis to |r(y,b) - r(y,b')|, so we need to know that $(b|b')_y < (b|b')_a$. The support of φg_0 is contained in both the support of $\varphi g(a,b)$ and the support of $\varphi g(a,b')$ then theorem 4.1(*iii*) tells us that $d(y,b) \le d(a,b) - 1$ and $d(y,b') \le d(a,b') - 1$. Thus

$$(b|b')_{y} = \frac{1}{2} (d(y,b) + d(y,b') - d(b,b'))$$

$$\leq \frac{1}{2} (d(a,b) - 1 + d(a,b') - 1 - d(b,b'))$$

$$= (b|b')_{a} - 1.$$

Hence I can apply the inductive hypothesis to get

$$\begin{aligned} |r(y,b) - r(y,b')| &\leq d(b,b') + 2(J+1) + \frac{1}{2}L\sum_{j=1}^{(b|b')_y} \lambda^j (2j + d(b,b')) \\ &\leq d(b,b') + 2(J+1) + \frac{1}{2}L\sum_{j=1}^{(b|b')_a - 1} \lambda^j (2j + d(b,b')) \end{aligned}$$

and so

$$|r(\varphi g_0, b) - r(\varphi g_0, b')| \le |g_0|_1 \left(d(b, b') + 2(J+1) + \frac{1}{2}L \sum_{j=1}^{(b|b')_a - 1} \lambda^j (2j + d(b, b')) \right).$$

Combining this with the bound on the other two terms together with the fact $|g_0|_1 \le 1$ gives

$$\begin{aligned} |r(a,b) - r(a,b')| &\leq |r(g_+,b)| + |r(g_0,b) - r(g_0,b')| + |r(g_-,b')| \\ &\leq \frac{1}{2}L\lambda^{(b|b')_a} (d(b,b') + 2(b|b')_a) \\ &\quad + d(b,b') + 2(J+1) + \frac{1}{2}L \sum_{j=1}^{(b|b')_a - 1} \lambda^j (2j + d(b,b')) \\ &= d(b,b') + 2(J+1) + \frac{1}{2}L \sum_{j=1}^{(b|b')_a} \lambda^j (2j + d(b,b')) \end{aligned}$$

as claimed.

5.2 Convergence of the double difference

The reason for altering the metric on \mathcal{G} was to be have better convergence properties at the boundary, namely we wanted to extend the double difference to the boundary (see section 3.4). Recall from proposition 3.16 that the double difference can be written as

$$(a,a'|b,b') = \frac{1}{2} \left(d(a,b) - d(a,b') - d(a',b) + d(a',b') \right).$$

Hence we look at R(a, a', b, b') := r(a, b) - r(a, b') - r(a', b) + r(a', b'). Our goal for this section is the following proposition about the behaviour of *R*.

Proposition 5.10. Let G be a uniformly fine, δ -hyperbolic graph. There exists constants $C \ge 0$ and $\omega \in (0,1)$ such that for all $a, a', b, b' \in V$, if $d(a, a') \le 1$ and $d(b, b') \le 1$ then

$$|R(a,a',b,b')| \leq C\omega^{d(a,b)}.$$

We will use induction on d(a, b) + d(a', b) to prove proposition 5.10, for which we will move *a* and/or *a'* towards *b* and *b'*. However, when we do this we will not always be able to assume $d(x, x') \le 1$ to apply the inductive hypothesis to R(x, x', b, b'). Therefore we have to allow d(a, a') to be larger in the inductive hypothesis. This forces us to take account of d(a, a') in the right-hand side of the inequality.

To make life easier, we fix some notation:

$$D(a, a', b) := d(a, b) + d(a', b)$$

$$\zeta := 2C_0 + 2(J - I) + \delta.$$

Then the inductive hypothesis we will actually use is the following.

Proposition 5.11. Let G be a uniformly fine, δ -hyperbolic graph. There are constants A, B > 0 and $\rho \in (0, 1)$ such that for any $a, a', b, b' \in V$, if $d(a, a') \leq \zeta$ and $d(b, b') \leq 1$ then

$$|R(a, a', b, b')| \le (A d(a, a') + B) \rho^{D(a, a', b)}$$

Before proving the more induction-friendly proposition 5.11, we show how it implies proposition 5.10

Proof of Proposition 5.10. In proposition 5.10 we assume that $d(a, a') \le 1$, so proposition 5.11 tells us

$$|R(a, a', b, b')| \le (A + B)\rho^{D(a, a', b)}$$

Moreover $D(a, a', b) \ge 2d(a, b) - 1$ since $d(a', b) \ge d(a, b) - 1$ by the triangle inequality. Hence

$$|R(a, a', b, b')| \le (A + B)\rho^{2d(a,b)-1}$$

and proposition 5.10 follows from setting the constants $C = (A + B)\rho^{-1}$ and $\omega = \rho^2$.

Throughout the rest of this section we keep the notation and assumptions as in proposition 5.11, namely we assume the following;

Assumptions. Suppose \mathcal{G} is a uniformly fine, δ -hyperbolic graph and assume $a, a', b, b' \in V$ are vertices such that $d(a, a') \leq \zeta$ and $d(b, b') \leq 1$. Moreover assume, without loss of generality, that $d(a, b) \geq d(a', b)$.

We are also still assuming that the constants μ , R, I and J satisfy the inequalities as laid out as assumptions 5.4.

The proof of proposition 5.11 will be an induction on D(a, a', b). We begin with a lemma which will start the induction.

Lemma 5.12. For any $\xi > 0$, if $D(a, a', b) \le \xi$ then

$$|R(a,a',b,b')| \leq 2\xi + 2.$$

Proof. We use proposition 5.2 as follows:

$$\begin{aligned} |R(a,a',b,b')| &\leq r(a,b) + r(a,b') + r(a',b) + r(a',b') \\ &\leq d(a,b) + d(a,b') + d(a',b) + d(a',b') \\ &\leq 2d(a,b) + d(b,b') + 2d(a',b) + d(b,b') \\ &\leq 2\xi + 2. \end{aligned}$$

Now we need to prove the inductive step.

We shall split into two cases and get an inequality in each case.

The first case is when the distance from *a* to *b* is much larger than the distance from a' to *b*. This is covered in lemma 5.13.

The second case is when a and a' have roughly the same distance to b. This is covered in lemma 5.14.

After this has been done we will need to show that the constants A, B, ρ can be chosen such that these inequalities are always satisfied, which is done in lemma 5.24.

Note that $(a'|b)_a \ge (a|b)_{a'}$ since $d(a, b) \ge d(a', b)$ (by assumption). Also, by the definition of ζ and the assumptions 5.4

$$d(a, a') \leq \zeta$$

= 2C₀ + 2(J - I) + δ
 \leq 2J - δ - 2
 $<$ J (5.2)

In particular, both $(a|b)_{a'} < J$ and $(a'|b)_a < J$.

First we consider the case where d(a, b) is much larger than d(a', b).

Lemma 5.13. If D(a, a', b) > 2(J + 2) and d(a, b) - d(a', b) > J - I then

$$R(a, a', b, b')| \le (A d(a, a') - A + B)\rho^{D(a, a', b) - J - C_0} + 2DL\lambda^{\frac{1}{2}(D(a, a', b) - \zeta - 2)}(2C_0 + J - I).$$

Proof. The idea is use φg to move *a* towards *b* and *b'*, and by doing so we will also have moved closer to *a'*. There will be a large overlap between $\varphi g(a, b)$ and $\varphi g(a, b')$, on which we can apply the inductive hypothesis and this will give the first term on the right-hand side of the inequality. Then there are two small error terms given by the part of $\varphi g(a, b)$ that is not also in $\varphi g(a, b')$ and vice versa. These two error terms will be bounded using corollary 5.6 and yield the second term of the inequality.

That is the rough idea. Now we have to go through the argument formally. We are assuming that $d(a, b) \ge d(a', b)$. Hence

$$2d(a,b) \ge d(a,b) + d(a',b) = D(a,a',b) > 2(J+2)$$

and so d(a,b) > J + 2. Then, by definition, $r(a,b) = 1 + r(\varphi g(a,b),b)$. Since $d(b,b') \le 1$ the triangle inequality yields

$$d(a,b') \ge d(a,b) - d(b,b') > J + 1$$

and so $r(a, b') = 1 + r(\varphi g(a, b'), b')$. Therefore

$$|R(a, a', b, b')| = |r(\varphi_{\mathcal{B}}(a, b), b) - r(\varphi_{\mathcal{B}}(a, b'), b') - r(a', b) + r(a', b')|.$$

As it is, the right-hand side cannot be written in terms of summands of the form R(x, a', b, b'). However, there will be an overlap in the support of g(a, b) and the support of g(a, b') and on this overlap we could try to use the inductive hypothesis. We make this formal, as in the proof of lemma 5.9, by defining new 0-chains g_+ , g_- , g_0 to have the following conditions;

- all coefficients of *g*₊, *g*₋, *g*₀ are positive
- $g(a,b) g(a,b') = g_+ g_-$
- $g(a,b) g_+ = g_0 = g(a,b') g_-$
- $\operatorname{supp}(g_+) \cap \operatorname{supp}(g_-) = \emptyset$.

It follows, using a similar argument as the one in the proof of lemma 5.9, that

$$\alpha := |g_+|_1 = |g_-|_1 \le \frac{1}{2} L \lambda^{(b|b')_a}.$$

Here the chain g_0 represents the overlap. Now

$$\begin{aligned} |R(a, a', b, b')| \\ &= |r(\varphi g(a, b), b) - r(\varphi g(a, b'), b') - r(a', b) + r(a', b')| \\ &= |r(\varphi g_{+}, b) + r(\varphi g_{0}, b) - r(\varphi g_{-}, b') - r(\varphi g_{0}, b') - r(a', b) + r(a', b')| \\ &= |R(\varphi g_{0}, a', b, b') + r(\varphi g_{+}, b) - r(\varphi g_{-}, b')|. \end{aligned}$$

The function *R* is not linear so for any arbitrary element $\sum_{x \in V} \alpha_x x \in \mathbb{Q}V$ the equality $R(\sum_{x \in V} \alpha_x x, a', b, b') = \sum_{x \in V} \alpha_x R(x, a', b, b')$ does not necessarily hold but it does hold if $\sum_{x \in V} \alpha_x = 1$. Here $|g_0|_1 = 1 - \alpha$. To get around this problem, we fix a vertex $x_0 \in V[a, b; I]$ and consider $\varphi g_0 + \alpha x_0$;

$$|R(a, a', b, b')| = |R(\varphi g_0, a', b, b') + r(\varphi g_+, b) - r(\varphi g_-, b')| = |R(\varphi g_0 + \alpha x_0, a', b, b') + r(\varphi g_+ - \alpha x_0, b) - r(\varphi g_- - \alpha x_0, b')| \le |R(\varphi g_0 + \alpha x_0, a', b, b')| + |r(\varphi g_+ - \alpha x_0, b)| + |r(\varphi g_- - \alpha x_0, b')|.$$

Now we have an inductive term and two error terms, which can be bounded using corollary 5.6. We know $|\varphi g_{\pm} - \alpha x_0|_1 \leq 2\alpha$ but we need to bound the diameter of the support of $\varphi g_{\pm} - \alpha x_0$. For this we need to bound the diameter of the support of φg_{\pm} as well as their maximal distances to x_0 . Observe that supp $(g_+) \subseteq$ supp(g(a, b)) and $x_0 \in$ supp $(\varphi g(a, b))$, which follows from theorem 4.1(*ii*). Then theorem 4.1(*v*) shows that the diameter of the support of $\varphi g_{\pm} - \alpha x_0$ is bounded by $2C_0 + I - I$.

Similarly supp(φg_{-}) \subseteq supp($\varphi g(a, b')$) whose diameter is bounded by $2C_0 + J - I$. However we do not necessarily have $x_0 \in$ supp($\varphi g(a, b')$) so it remains to bound the maximal distance from x_0 to an element in the support of φg_{-} . Fix a geodesic triangle Δ with corners a, b, b' such that Δ contains the point x_0 , and let x'_0 be the vertex of Δ between a and b' whose distance to a is I. Then $x'_0 \in \text{supp}(\varphi g(a, b'))$ and $d(x'_0, y) \leq C_0 + J - I$ for any element $y \in \text{supp}(\varphi g_-)$ using theorem 4.1(*ii*).

We want to use hyperbolicity to get $d(x_0, x'_0) \le \delta$ but for this we need to know that $(b|b')_a \ge I$. Recall that $d(a, b) \ge d(a', b)$ and then;

$$(b|b')_{a} = \frac{1}{2}(d(a,b) + d(a,b') - d(b,b'))$$

$$\geq \frac{1}{2}(d(a,b) + d(a,b) - 2d(b,b'))$$

$$\geq \frac{1}{2}(D(a,a',b) - 2)$$

$$\geq I.$$
(5.3)

So hyperbolicity gives $d(x_0, x'_0) \le \delta$ and since $C_0 \ge \delta$ we conclude that

$$\operatorname{diam}(\operatorname{supp}(\varphi g_{\pm} - \alpha x_0)) \leq 2C_0 + J - I.$$

Therefore corollary 5.6 gives

$$|r(\varphi g_{\pm} - \alpha x_0, b)| \leq D|\varphi g_{\pm} - \alpha x_0|_1 \operatorname{diam}(\operatorname{supp}(\varphi g_{\pm} - \alpha x_0))$$
$$\leq DL\lambda^{(b|b')_a}(2C_0 + J - I).$$

We want the power of λ in terms of D(a, a', b) and not $(b|b')_a$. However, inequality (5.3) gives $(b|b')_a \ge \frac{1}{2}(D(a, a', b) - 2) \ge \frac{1}{2}(D(a, a', b) - \zeta - 2)$. Thus

$$|r(\varphi g_{\pm} - \alpha x_0, b)| \le DL\lambda^{\frac{1}{2}(D(a,a',b)-\zeta-2)}(2C_0 + J - I).$$

Now consider the inductive term $|R(\varphi g_0 + \alpha x_0, a, b, b')|$. Using barycentric coordinates write $\varphi g_0 + \alpha x_0 = \sum_{x \in V} \alpha_x x$. Then

$$\sum_{x \in V} \alpha_x = |\varphi g_0 + \alpha x_0|_1 = |\varphi g_0|_1 + \alpha = 1.$$

Thus $R(\varphi g_0 + \alpha x_0, a, b, b') = \sum_{x \in V} \alpha_x R(x, a, b, b').$

Before we can apply the inductive hypothesis to R(x, a, b, b') we need to know that $d(x, a) \le \zeta$ and D(x, a', b) < D(a, a', b).

Observe that $\operatorname{supp}(\varphi g_0) \subseteq \operatorname{supp}(\varphi g(a, b))$. Moreover it follows from lemma 4.15 that $x_0 \in \operatorname{supp}(\varphi g_I(a, b))$. Then for all $x \in \operatorname{supp}(\varphi g_0 + \alpha x_0)$ theorem 4.1(*iii*) says $d(x, b) \leq d(a, b) - 1$ and so

$$D(x, a', b) = d(x, b) + d(a', b) \le D(a, a', b) - 1.$$

We still need to show $d(x, a') \leq \zeta$ before we can use the inductive hypothesis. Consider a geodesic triangle Δ with corners a, a', b such that Δ contains x_0 . For $I \leq \aleph \leq J$ let $u_\aleph \in [a, b]$ satisfy $d(a, u_\aleph) = \aleph$. In

particular $u_I = x_0$. To bound d(x, a) we need to consider the possibilities $\aleph \le (a'|b)_a$ and $\aleph > (a'|b)_a$ separately.

Suppose $\aleph \leq (a'|b)_a$. Let $v_{\aleph} \in [a, a']$ be the vertex with $d(v_{\aleph}, a) = \aleph$. By hyperbolicity $d(u_{\aleph}, v_{\aleph}) \leq \delta$. Then using theorem 4.1(*iv*)

$$d(x,a') \le d(x,u_{\aleph}) + d(u_{\aleph},v_{\aleph}) + d(v_{\aleph},a')$$

$$\le C_0 + \delta + d(a,a') - I$$

$$\le d(a,a') - 1.$$

In particular, $d(x, a') \le d(a, a') \le \zeta$.

So suppose $\aleph > (a'|b)_a$ and let w_{\aleph} be the vertex of the [a',b] in the triangle Δ with $d(w_{\aleph},b) = d(u_{\aleph},b)$. Hyperbolicity tells us $d(u_{\aleph},w_{\aleph}) \leq \delta$. Therefore

$$d(x,a') \le d(x,u_{\aleph}) + d(u_{\aleph},w_{\aleph}) + d(w_{\aleph},a')$$

$$\le C_0 + \delta + d(a',b) - d(w_{\aleph},b)$$

$$= C_0 + \delta + d(a',b) - d(u_{\aleph},b)$$

$$= C_0 + \delta + d(a',b) - d(a,b) + d(a,u_{\aleph})$$

$$\le C_0 + \delta - (J-I) + J$$

$$\le J - I - 1$$

$$\le d(a,b) - d(a',b) - 1$$

$$\le d(a,a') - 1$$

and in particular, $d(x, a') \le d(a, a') \le \zeta$. Now the inductive hypothesis holds and

$$|R(x,a',b,b')| \le (A d(x,a') + B) \rho^{D(x,a',b)}$$

$$\leq (A d(a,a') - A + B)\rho^{D(x,a',b)}$$

We need to remove the dependence on *x* so we need an upper bound for $\rho^{D(x,a',b)}$. Since $\rho \in (0,1)$, we need a lower bound on D(x,a',b);

$$D(x, a', b) = d(x, b) + d(a', b)$$

$$\geq d(a, b) - d(a, u_{\aleph}) - d(u_{\aleph}, x) + d(a', b)$$

$$\geq d(a, b) - J - C_0 + d(a', b)$$

$$= D(a, a', b) - (J + C_0).$$

Therefore

$$|R(x, a', b, b')| \le (A d(a, a') - A + B)\rho^{D(a, a', b) - (J - C_0)}$$

and so

$$\begin{aligned} |R(\varphi g_0 + \alpha x_0, a', b, b')| &\leq \sum_{x \in V} \alpha_x |R(x, a', b, b')| \\ &\leq \sum_{x \in V} \alpha_x (A \, d(a, a') - A + B) \rho^{D(a, a', b) - (J - C_0)} \\ &= (A \, d(a, a') - A + B) \rho^{D(a, a', b) - (J - C_0)}. \end{aligned}$$

So to conclude the proof, we put these terms together and obtain

$$|R(a, a', b, b')| \leq |R(\varphi g_0 + \alpha x_0, a', b, b')| + |r(\varphi g_+ - \alpha x_0, b)| + |r(\varphi g_- - \alpha x_0, b')| \leq (A d(a, a') - A + B)\rho^{D(a, a', b) - (J - C_0)} + 2DL\lambda^{\frac{1}{2}(D(a, a', b) - \zeta - 2)}(2C_0 + J - I)$$

which is exactly the inequality that the lemma claims to be true.

Next consider the case where d(a, b) is roughly the same as d(a', b). Set

$$\xi = 2 \max\left\{J+1, \frac{\ln 2 - \ln L}{\ln \lambda}\right\} + 2\zeta + 2.$$
(5.4)

We aim for the following lemma;

Lemma 5.14. Suppose $D(a, a', b) \ge \xi + 1$. If $d := d(a, b) - d(a', b) \le J - I$ then there exists some constant $\eta_2 \in (0, 1)$ such that

$$|R(a, a', b, b')| \le \eta_2 (A\zeta + B) \rho^{D(a, a', b) - 2(C_0 + J)} + 4DL\lambda^{\frac{1}{2}(D(a, a', b) - \zeta - 2)} (2C_0 + J - I).$$

The idea here is to move both *a* and *a'*. As in the proof of lemma 5.13 we consider the overlap between $\varphi g(a, b)$ and $\varphi g(a, b')$. But we also consider the overlap between $\varphi g(a', b)$ and $\varphi g(a', b')$. This gives us four error terms and something to which we can apply the inductive hypothesis.

However, there is an overlap of the overlaps, and on this double overlap the double difference *R* will vanish. This is where the constant η_2 will come from.

Before we can do this we need to know that d(a,b) > J + 1 so that we do use the recursive definition of r. We need this for all four pairings of vertices.

Lemma 5.15. *Suppose* $D(a, a', b) > \xi$ *. Then*

$$d(a,b) > J+1$$
, $d(a,b') > J+1$, $d(a',b) > J+1$, $d(a',b') > J+1$.

Proof. These four inequalities follow from the triangle inequality and the assumption $d(a,b) \ge d(a',b)$.

$$\begin{aligned} d(a,b) &\geq \frac{1}{2} \left(d(a,b) + d(a',b) \right) > \frac{1}{2} \xi &\geq J + 1 + \zeta + 1. \\ d(a,b') &\geq \quad d(a,b) - d(b,b') &> J + 1. \\ d(a',b) &\geq \quad d(a,b) - d(a,a') &> J + 1 + 1. \\ d(a',b') &\geq \quad d(a',b) - d(b,b') &> J + 1. \end{aligned}$$

This tells us that $r(a,b) = 1 + r(\varphi g(a,b),b)$ and so on for the other pairings. Hence

$$R(a, a', b, b') = r(\varphi g(a, b), b) - r(\varphi g(a, b'), b') -r(\varphi g(a', b), b) + r(\varphi g(a', b'), b').$$

Recall that $g(a, b) = \frac{1}{1+J-I} \sum_{\aleph=I}^{J} \operatorname{av}_{B_{R}^{\zeta}} g_{\aleph}(a, b)$. The condition $d \leq J - I$ means that $J - d \geq I$. The idea is to use an argument similar to the proof of lemma 5.13 to split $r(\varphi g(a, b), b) - r(\varphi g(a, b'), b')$ into four terms,

$$r(\varphi g(a,b),b) - r(\varphi g(a,b'),b') = r(\varphi g_0 + \alpha y_0,b) - r(\varphi g_0 + \alpha y_0,b') + r(\varphi g_+ - \alpha y_0,b) - r(\varphi g_- - \alpha y_0,b')$$

for some well-chosen $y_0 \in V$. The third and fourth terms are error terms and can be bounded using corollary 5.6. We can also do this for a' in place of a, although we want to use the same y_0 . This leaves the expression

$$r(\varphi g_0 + \alpha y_0, b) - r(\varphi g_0 + \alpha y_0, b') - r(\varphi g'_0 + \alpha' y_0, b) + r(\varphi g'_0 + \alpha' y_0, b')$$

= $R(\varphi g_0 + \alpha y_0, \varphi g'_0 + \alpha' y_0, b, b')$

and here we want to use an argument similar to the one used in case 1 of the proof of lemma 5.5; we want to use the cancellation between $\varphi g_0 + \alpha y_0$ and $\varphi g'_0 + \alpha' y_0$.

To summarise, we need to pick a suitable y_0 , split up once, bound the terms involving cycles, split up a second time, and apply the inductive hypothesis.

For $I \leq \aleph \leq J$ define new elements $g_{\aleph,+}, g_{\aleph,-}, g_{\aleph,0} \in \mathbb{Q}E$ by

- all coefficients are positive
- $g(a,b) g_{\aleph,+} = g_{\aleph,0} = g(a,b') g_{\aleph,-}$
- $g_{\aleph}(a,b) g_{\aleph}(a,b') = g_{\aleph,+} g_{\aleph,-}$
- $\operatorname{supp}(g_{\aleph,+}) \cap \operatorname{supp}(g_{\aleph,-}) = \emptyset$.

Then set $g_+ = \frac{1}{1+J-I} \sum_{\aleph=I}^{J} \operatorname{av}_{B_R^{\zeta}}(g_{\aleph,+})$. Similarly we can define g_- and g_0 . Observe that $g(a,b) = g_+ + g_0$ and $g(a,b') = g_- + g_0$ but the supports of g_+ and g_- are not necessarily disjoint.

Later on it will be important that we split prior to taking the averages, which is why we have deviated slightly from the method of proof in lemmas 5.9 and 5.13.

For bounding the error terms we need to first bound the size of $|g_{\aleph,\pm}|_1$ and $|g'_{\aleph,\pm}|_1$.

Lemma 5.16. For all \aleph ,

$$\alpha_{\aleph} := |g_{\aleph,+}|_1 = |g_{\aleph,-}|_1 \le \frac{1}{2} L \lambda^{\frac{1}{2}(D(a,a',b)-\zeta-2)}.$$

Proof. Using the argument in the proof of lemma 5.9 we get that

$$|g_{\aleph,+}|_1 = |g_{\aleph,-}|_1 \leq \frac{1}{2}L\lambda^{(b|b')_a}$$

(It was shown for *g* instead of g_{\aleph} but the proof is identical.) So we need to find a lower bound for the Gromov product $(b|b')_a$. We are still assuming $d(a,b) \ge d(a',b)$ so inequality (5.3) in the proof of lemma 5.13 still holds, i.e. $(b|b')_a \ge \frac{1}{2}(D(a,a',b)-2)$. Combining this with the inequality for $|g_{\aleph,\pm}|_1$ finishes the proof of the lemma.

Analogously, we can define $g'_{\aleph,+}, g'_{\aleph,-}, g_{\aleph,0}$ for a' in place of a. There is a version of lemma 5.16 here;

Lemma 5.17. For all ℵ,

$$\alpha'_{\aleph} := |g'_{\aleph,+}|_1 = |g'_{\aleph,-}|_1 \le \frac{1}{2} L \lambda^{\frac{1}{2}(D(a,a',b)-\zeta-2)}.$$

Proof. Using the same argument as the proof of lemma 5.16 gives

$$|g'_{\aleph,+}|_1 = |g'_{\aleph,-}|_1 \le \frac{1}{2}L\lambda^{(b|b')_{a'}}$$

and so we need a lower bound of $(b|b')_{a'}$;

$$(b|b')_{a'} = \frac{1}{2}(d(a',b) + d(a',b') - d(b,b'))$$

$$\geq \frac{1}{2}(d(a,b) - d(a,a') + d(a',b) - 2d(b,b'))$$

$$\geq \frac{1}{2}(D(a,a',b) - \zeta - 2)$$
(5.5)

thus we get the inequality as claimed.

Before we get to the inductive part we quickly deal with the error terms.

Lemma 5.18. Suppose $D(a, a', b) \ge \xi + 1$. Given any vertex $y_0 \in V[a, b; J]$ all four terms

- $|r(\varphi g_+ \alpha y_0, b)|$
- $|r(\varphi g_- \alpha y_0, b')|$
- $|r(\varphi g'_+ \alpha' y_0, b)|$
- $|r(\varphi g'_{-} \alpha' y_0, b')|$

are bounded from above by the expression

$$DL(2C_0 + J - I)\lambda^{\frac{1}{2}(D(a,a',b) - \zeta - 2)}.$$
(5.6)

Proof. We want to use corollary 5.6.

We already know that $\varphi g_+ - \alpha y_0$ is a cycle in QV but we need to bound the diameter of its support. The support of φg_+ is contained in the support of $\varphi g(a, b)$ and it follows from theorem 4.1(*ii*) that $y_0 \in \text{supp}(\varphi g(a, b))$. Then theorem 4.1(*v*) tells us that the diameter of the support of $\varphi g_+ - \alpha y_0$ is bounded by $2C_0 + J - I$.

Therefore, using lemma 5.16 with corollary 5.6 gives

$$|r(\varphi g_{+} - \alpha y_{0}, b)| \leq DL(2C_{0} + J - I)\lambda^{\frac{1}{2}(D(a,a',b)-\zeta-2)}.$$

We want to do a similar thing for the other terms.

The support of φg_{-} is contained in the support of $\varphi g(a, b')$ so using the argument as above all we need to do is bound the distance from an element in the support of $\varphi g(a, b')$ to y_0 . Consider a geodesic triangle with corners a, b, b' that contains the point y_0 . Inequality (5.3) in the proof of lemma 5.13 states $(b|b')_a \ge \frac{1}{2}(D(a, a', b) - 2)$. But $D(a, a', b) \ge \xi$ by assumption and then from the definition of ξ (see equation (5.4)) we obtain $(b|b')_a \ge J$. Hence we can use hyperbolicity to go from y_0 to the geodesic [a, b']. Using this, together with theorem 4.1(v), for every y in the support of $\varphi g(a, b')$;

$$d(y, y_0) \le C_0 + J - I + \delta \le 2C_0 + J - I.$$

Therefore we can bound $|r(\varphi g_{-} - \alpha y_0, b')|_1$.

We want to use a similar argument to bound the remaining two terms. We need to use lemma 5.17 instead of lemma 5.16 but the rest of the argument is very similar. Observe that the support of $\varphi g'_+$, $\varphi g'_-$ is contained in the support of $\varphi g(a', b)$, $\varphi g(a', b')$ respectively. Hence the only thing we

need to do is bound the maximal distance from y_0 to the an element in the support of $\varphi g(a', b)$ or $\varphi g(a', b')$.

For $y \in \operatorname{supp}(\varphi g(a', b))$ we can consider a geodesic triangle Δ whose corners are a, a', b and that contains the point y_0 , and then the assumption $(a'|b)_a$ allows us to use hyperbolicity around the vertex b. This together with theorem 4.1(*iv*) gives $d(y, y_0) \leq C_0 + J - I + \delta$. This leads to the bound for $|r(\varphi g'_+ - \alpha' y_0, b)|$.

For $y' \in \text{supp}(\varphi g(a', b'))$ we need to go further and consider a geodesic triangle Δ' whose corners are a', b, b' such that it shares the side [a', b] with the triangle Δ , as in the picture below.



Inequality (5.5) states $(b|b')_{a'} \ge \frac{1}{2}(D(a, a', b) - \zeta - 2)$, and by assumption this leads to $(b|b')_{a'} \ge J$. Hence we can consider hyperbolicity in Δ' around the vertex a' and conclude that $d(y', y_0) \le C_0 + J - I + 2\delta$, where again we also need to use theorem 4.1(iv). Thus we get the bound on the term $|r(\varphi g'_{-} - \alpha' y_0, b')|$.

Now we can move on to the term $|R(\varphi g_0 + \alpha y_0, \varphi g'_0 + \alpha' y_0, b, b')|$. The idea is to show that there is an overlap between $\varphi g_0 + \alpha y_0$ and $\varphi g'_0 + \alpha' y_0$ on which R = 0 and use an argument similar to that in the proof of lemma 5.9.

Once again we define new elements in $\mathbb{Q}V$; fix an element $y_0 \in V[a, b; J]$ and define $h_+, h_-, h_0 \in \mathbb{Q}V$ by the following properties

- all coefficients are positive
- $h_+ h_- = \varphi g_0 + \alpha y_0 (\varphi g'_0 + \alpha' y_0)$
- $\varphi g_0 + \alpha y_0 h_+ = h_0 = \varphi g'_0 + \alpha' y_0$
- $\operatorname{supp}(h_+) \cap \operatorname{supp}(h_-) = \emptyset$.

So h_0 represents the overlap. These chains depend on the choice of y_0 in V[a, b; J] but from now on we keep y_0 fixed so it is suppressed in the notation.

For any $x \in V$, R(x, x, b, b') = 0, so h_0 does not contribute anything and we need only consider $R(h_+, h_-, b, b')$. Hence we try to find a lower bound for $|h_0|_1$, i.e. a lower bound for how much cancels out.

The idea is to show that any edge $e_0 \in E[a, b; J]$ is *R*-close to every element in the support of $g_{J,0}$ and to every element in the support of $g'_{I-d,0}$.

Then the coefficient of e_0 in $\operatorname{av}_{B_R^{\zeta}} g_{J,0}$ is at least $|g_{J,0}|_1/B_R^{\zeta}$, where B_R^{ζ} is the maximum size of a ball of radius R in the edge metric d^{ζ} , which is finite by lemma 2.47 and the comments following it. This allows us to bound the coefficient of y_0 in $\varphi \operatorname{av}_{B_R^{\zeta}} g_{J,0}$ from below by $|g_{J,0}|_1/2B_R^{\zeta}$. Similarly we will get a lower bound for the coefficient of y_0 in $\varphi \operatorname{av}_{B_R^{\zeta}} g'_{J,0}$ from $\varphi \operatorname{av}_{B_R^{\zeta}} g'_{J-d,0}$ of $|g'_{J-d,0}|_1/2B_R^{\zeta}$. From this we get a lower bound on $|h_0|_1$.

First we look for a lower bound of $|g_{L0}|_1$ and $|g'_{L-d,0}|_1$.

Lemma 5.19. *If* $D(a, a', b) \ge \xi + 1$ *then* $|g_{J,0}|_1 \ge 1 - \lambda$ *and* $|g'_{J-d,0}|_1 \ge 1 - \lambda$. *Proof.* Note that $|g_{J,0}|_1 = 1 - \alpha_J \ge 1 - \frac{1}{2}L\lambda^{\frac{1}{2}(D(a,a',b)-\zeta-2)}$ using lemma 5.16. Moreover, we have

$$\frac{1}{2}(D(a,a',b)-\zeta-2) \geq \frac{\ln 2 - \ln L}{\ln \lambda} + 1$$

by the definition of ξ . Hence $\lambda^{\frac{1}{2}(D(a,a',b)-\zeta-2)} \leq \frac{2}{L}\lambda$ and so $|g_{J,0}|_1 \geq 1 - \lambda$.

Similarly $|g'_{J-d,0}|_1 \ge 1 - \lambda$, except we need to use lemma 5.17 instead of lemma 5.16.

We can use the lower bounds in lemma 5.19 to find lower bounds for the coefficient of y_0 in $\varphi av_{B_R^{\zeta}}(g_{J,0})$ and in $\varphi av_{B_R^{\zeta}}(g'_{J-d,0})$, which will allow us to get the bound in the following lemma.

Lemma 5.20. Suppose $D(a, a', b) \ge \xi + 1$. If $d := d(a, b) - d(a', b) \le J - I$ then there is a constant $\eta_1 \in (0, 1)$ such that for all $I \le \aleph, \aleph' \le J$;

- (i) $\left| \varphi a v_{B_R^{\zeta}}(g_{\aleph,0}) + \alpha_{\aleph} y_0 \right|_1 = 1 = \left| \varphi a v_{B_R^{\zeta}}(g'_{\aleph',0}) + \alpha'_{\aleph'} y_0 \right|_1;$
- (*ii*) $\left| \varphi a v_{B_R^{\zeta}}(g_{J,0}) + \alpha_J y_0 \left(\varphi a v_{B_R^{\zeta}}(g'_{J-d,0}) + \alpha'_{J-d} y_0 \right) \right|_1 \leq 2\eta_1.$

Proof. For the first part;

$$|\varphi \operatorname{av}_{B_R^{\zeta}}(g_{\aleph,0}) + \alpha_{\aleph} y_0|_1 = |g_{\aleph,0}|_1 + \alpha_{\aleph} = 1$$

and similarly for $|\varphi \operatorname{av}_{B_p^{\zeta}}(g'_{\aleph',0}) + \alpha'_{\aleph'}y_0|_1$.

For the second part, we look at the contribution from y_0 to find a bound on how much of the coefficient of y_0 gets cancelled. Let β_J be the coefficient of y_0 in $\varphi av_{B_R^{\zeta}}(g_{J,0})$ and let β'_{J-d} be the coefficient of y_0 in $\varphi av_{B_R^{\zeta}}(g'_{J-d,0})$. Then

$$\begin{aligned} |\varphi a v_{B_{R}^{\zeta}}(g_{J,0}) + \alpha_{J} y_{0} - (\varphi a v_{B_{R}^{\zeta}}(g'_{J-d,0}) + \alpha'_{J-d} y_{0})|_{1} \\ &\leq |\varphi a v_{B_{R}^{\zeta}}(g_{J,0}) - \beta_{J} y_{0}|_{1} + \alpha_{J} \\ &+ |\varphi a v_{B_{R}^{\zeta}}(g'_{J-d,0}) - \beta'_{J-d} y_{0}|_{1} + \alpha'_{J-d} + |\beta_{J} - \beta'_{J-d}| \\ &= |g_{J,0}|_{1} - \beta_{J} + \alpha_{J} + |g'_{J-d,0}|_{1} - \beta'_{J-d} + \alpha'_{J-d} + |\beta_{J} - \beta'_{J-d} \\ &= 2 - 2 \min\{\beta_{J}, \beta'_{J-d}\} \end{aligned}$$

and so we need a lower bound for β_I and β'_{I-d} .

The vertex $y_0 \in V[a, b; J]$ must be the end-point of an edge $e_0 \in E[a, b; J]$ and by lemma 4.15 this edge is contained in the support of $g_J(a, b)$. The support of $g_{J,0}$ is also contained in the support of $g_J(a, b)$ so for any edge ein the support of $g_{J,0}$ it follows from lemma 4.15 that $d^{\zeta}(e_0, e) \leq 3\delta \leq R$.

So e_0 is contained in $B_R^{\zeta}(e)$ for any edge e in the support of $g_{J,0}$ and thus we know that $\beta_J \geq |g_{J,0}|_1/2B_R^{\zeta}$. Then we can apply lemma 5.19 to deduce that $\beta_J \geq (1-\lambda)/2B_R^{\zeta}$.

Analogously, if we can show that e_0 is *R*-close to any edge in the support of $g'_{I-d,0}$ then we would know that $\beta'_{I-d} \ge (1-\lambda)/2B_R^{\zeta}$.

Fix an edge $e'_0 \in E[a', b; J - d]$. The support of $g'_{J-d,0}$ is contained in the support of g(a', b), hence lemma 4.15 tells us that the support of $g'_{J-d,0}$ is contained in $Fl[a', e'_0; 3\delta]$. We picked e'_0 such that $d(b, e'_0) = d(b, e_0)$, and the fact $(a'|b)_a < J$ (see inequality (5.2)) gives $d(b, e_0) \leq (a|a')_b$ so by the edge-hyperbolicity of \mathcal{G} (given in lemma 2.50) $d^{\zeta}(e_0, e'_0) \leq \delta$. Therefore, for any edge e in the support of $g'_{I-d,0}$ we deduce $d^{\zeta}(e, e_0) \leq 4\delta \leq R$.

Thus $\beta'_{I-d} \geq (1-\lambda)/2B_R^{\zeta}$.

So both β_J and β'_{I-d} are bounded from below by $(1-\lambda)/2B_R^{\zeta}$. Therefore

$$\begin{split} |\varphi a \mathbf{v}_{B_{R}^{\zeta}}(g_{J,0}) + \alpha_{J} y_{0} - (\varphi a \mathbf{v}_{B_{R}^{\zeta}}(g_{J-d,0}') + \alpha_{J-d}' y_{0})|_{1} &\leq 2 - 2 \min\{\beta_{J}, \beta_{J-d}'\}\\ &\leq 2 - 2 \left(\frac{1 - \lambda}{2B_{R}^{\zeta}}\right)\\ &= 2\eta_{1} \end{split}$$

where $\eta_1 := 1 - \frac{1-\lambda}{2B_R^{\zeta}}$.

Now we need to use this to give a lower bound on $|h_0|_1$, which equates to an upper bound on $|h_+|_1$ and $|h_-|_1$.

Lemma 5.21. Suppose $D(a, a', b) \ge \xi + 1$. If $d := d(a, b) - d(a', b) \le J - I$ then there exists $\eta_2 \in (0, 1)$ such that;

- (*i*) $|h_+|_1 = |h_-|_1$
- (*ii*) $|h_{\pm}|_1 \leq \eta_2$.

Proof. The first fact can be proven using the augmentation homomorphism, see case 1 of the proof of lemma 5.5 for an example of how the argument works.

So we just need to bound $|h_{\pm}|_1$. Since all the coefficients are positive and the supports are disjoint we know $|h_{+}|_1 + |h_{-}|_1 = |h_{+} - h_{-}|_1$. Then

using the definitions gives

$$\begin{split} |h_{\pm}|_{1} &= \frac{1}{2} |\varphi g_{0} + \alpha y_{0} - (\varphi g_{0}' + \alpha' y_{0})|_{1} \\ &\leq \frac{1}{2(1+J-I)^{2}} \sum_{\aleph,\aleph'=I}^{J} \left| \varphi \mathsf{av}_{B_{R}^{\zeta}}(g_{\aleph,0}) + \alpha_{\aleph} y_{0} - (\varphi \mathsf{av}_{B_{R}^{\zeta}}(g_{\aleph',0}') + \alpha_{\aleph'}' y_{0}) \right|_{1} \end{split}$$

Every summand is bounded by 2, and furthermore lemma 5.20 tells us that the summand with $\aleph = J$ and $\aleph' = J - d$ is bounded by $2\eta_1$ with $\eta_1 \in (0, 1)$. So this gives

$$\begin{split} |h_{\pm}|_{1} &\leq \frac{1}{2(1+J-I)^{2}} \left(2 \left((1+J-I)^{2}-1 \right) + 2\eta_{1} \right) \\ &= 1 - \frac{1-\eta_{1}}{(1+J-I)^{2}} \end{split}$$

and setting $\eta_2 = 1 - \frac{1 - \eta_1}{(1 + J - I)^2} \in (0, 1)$ completes the proof of the lemma. \Box

This means that if we write $h_+ = \sum_{x \in V} \alpha_x^+ x$ and $h_- = \sum_{x' \in V} \alpha_{x'}^- x'$ then

$$R(h_{+},h_{-},b,b') = \frac{1}{|h_{\pm}|_{1}} \sum_{x,x' \in V} \alpha_{x}^{+} \alpha_{x'}^{-} R(x,x',b,b').$$

Now we want to apply our inductive hypothesis to R(x, x', b, b'). For this, we need to know that the conditions for the inductive hypothesis are satisfied, namely we need $d(x, x') \leq \zeta$ and D(x, x', b) < D(a, a', b). Moreover, after we have applied the inductive hypothesis we will want to translate back from being in terms of d(x, x') and D(x, x', b) to being in terms of d(a, a') and D(a, a', b).

Lemma 5.22. Suppose $D(a, a', b) \ge \xi + 1$. If $d := d(a, b) - d(a', b) \le J - I$ then for any $x \in supp(h_+)$ and any $x' \in supp(h_-)$

- (i) $D(a,a',b) 2(C_0 + J) \le D(x,x',b) \le D(a,a',b) 2;$
- (*ii*) $d(x, x') \leq \zeta$.

Proof. (*i*) The support of h_+ is contained in the support of $\varphi g_0 + \alpha y_0$, and the support of g_0 is contained in both the support of g(a, b) and the support of g(a, b'), so it is enough to prove the lemma for $x \in \text{supp}(\varphi g(a, b))$. For such x we have $d(x, b) \leq d(a, b) - 1$ by theorem 4.1(*iii*). Moreover,

$$d(x,b) \ge d(a,b) - d(a,x) \ge d(a,b) - (C_0 + J)$$

using theorem 4.1(iv).



Similarly for any $x' \in \operatorname{supp}(\varphi g'_0) \subseteq \operatorname{supp}(\varphi g(a', b))$ we have

$$d(a',b) - (C_0 - J) \le d(x',b) \le d(a',b) - 1.$$

We still need to find bounds for the possibility that *x* or *x'* is y_0 . By definition $y_0 \in V[a, b; J]$ so $d(y_0, b) = d(a, b) - J$.

(*ii*) It follows from theorem 4.1(*ii*) that $y_0 \in \text{supp}(\varphi g(a, b))$ so we fix an arbitrary $x \in \text{supp}(\varphi g(a, b))$ and consider two possibilities depending on whether $x' = y_0$ or x' is in the support of $\varphi g(a', b)$.

If $x' = y_0$ then $d(x, x') \leq \text{diam}(\text{supp}(\varphi g(a, b))) \leq 2C_0 + J - I \leq \zeta$, using theorem 4.1(*v*).



Recall that $(a'|b)_a \leq J$, which follows from inequality (5.2). So we can use theorem 4.1(iv) together with the assumption $d(a,b) - d(a',b) \leq J - I$ to show that if $x' \in \text{supp}(\varphi_S(a',b))$ then

$$d(x, x') \le 2C_0 + \delta + 2(J - I) \le \zeta.$$

(See the picture above.)

Therefore for any $x \in \text{supp}(\varphi g_0 + \alpha y_0)$ and $x' \in \text{supp}(\varphi g'_0 + \alpha' y_0)$ we have $d(x, x') \leq \zeta$.

With lemma 5.22 we can apply the inductive hypothesis to R(x, x', b, b') for $x \in \text{supp}(h_+)$ and $x' \in \text{supp}(h_-)$.

Corollary 5.23. Suppose $D(a, a', b) \ge \xi + 1$. If $d := d(a, b) - d(a', b) \le J - I$ then

$$|R(h_+, h_-, b, b')| \le \eta_2 (A\zeta + B) \rho^{D(a, a', b) - 2(C_0 + J)}$$

where η_2 is the constant from lemma 5.21.

Proof. Write $h_+ = \sum_{x \in V} \alpha_x^+ x$ and $h_- = \sum_{x' \in V} \alpha_{x'}^- x'$ so that

$$R(h_{+},h_{-},b,b') = \frac{1}{|h_{\pm}|_{1}} \sum_{x,x' \in V} \alpha_{x}^{+} \alpha_{x'}^{-} R(x,x',b,b').$$

Lemma 5.22(*ii*) shows $d(x, x') \le \zeta$ and $D(x, x', b) \le D(a, a', b) - 2$ so we can apply the inductive hypothesis to R(x, x', b, b') to get

$$|R(x, x', b, b')| \le (A d(x, x') + B) \rho^{D(x, x', b)}.$$

Moreover lemma 5.22(*i*) tells us $D(x, x', b) \ge D(a, a', b) - 2(C_0 + J)$. Therefore

$$\begin{aligned} |R(h_{+},h_{-},b,b')| &\leq \frac{1}{|h_{\pm}|_{1}} \sum_{x,x' \in V} \alpha_{x}^{+} \alpha_{x'}^{-} (A\zeta + B) \rho^{D(a,a',b)-2(C_{0}+J)} \\ &= |h_{\pm}|_{1} (A\zeta + B) \rho^{D(a,a',b)-2(C_{0}+J)} \end{aligned}$$

and $|h_{\pm}|_1$ was bounded by η_2 in lemma 5.21.

And so we can finish the inductive step in proving proposition 5.11 for this case, i.e. we can give the proof of lemma 5.14;

Proof of Lemma 5.14. Split up R(a, a', b, b') into the five terms

$$R(a, a', b, b') = R(\varphi g_0 + \alpha y_0, \varphi g'_0 + \alpha' y_0, b, b') + r(\varphi g_+ - \alpha y_0, b) - r(\varphi g_- - \alpha y_0, b') - r(\varphi g'_+ - \alpha' y_0, b) + r(\varphi g'_- - \alpha' y_0, b').$$

The inductive term $R(\varphi g_0 + \alpha y_0, \varphi g'_0 + \alpha' y_0, b, b') = R(h_+, h_-, b, b')$ using corollary 5.23 and lemma 5.18 yields a bound for the four error terms, and combining these bounds gives

$$\begin{aligned} |R(a,a',b,b')| &\leq 4DL\lambda^{\frac{1}{2}(D(a,a',b)-\zeta-2)}(2C_0+J-I) \\ &+\eta_2(A\zeta+B)\rho^{D(a,a',b)-2(C_0+J)}. \end{aligned}$$

To finish the proof of proposition 5.11 it remains to show that we can choose the constants *A*, *B* and ρ appropriately.

Lemma 5.24. For any $\xi > 0$ there exists constants A, B > 0 and $\rho \in (0, 1)$ such that for any a, a' with $1 \le d(a, a') \le \zeta$ and all b, b' with $d(b, b') \le 1$ the following three inequalities hold;

$$2\xi + 2 \le A\lambda^{\frac{1}{2}\xi} \tag{5.7}$$

$$(A d(a, a') - A + B)\rho^{-J - C_0} + K\left(\frac{\lambda^{\frac{1}{2}}}{\rho}\right)^{D(a, a', b)} \le A d(a, a') + B$$
(5.8)

$$\eta_2(A\zeta + B)\rho^{-2(C_0+J)} + 2K\left(\frac{\lambda^{\frac{1}{2}}}{\rho}\right)^{D(a,a',b)} \le A\,d(a,a') + B \tag{5.9}$$

where $K := 2DL(2C_0 + J - I)\lambda^{-\frac{1}{2}(\zeta+2)}$.

Proof. First pick A > K large enough such that $2\xi + 2 \leq A\lambda^{\frac{1}{2}\xi}$. Then pick $\rho_1 \in (\lambda^{\frac{1}{2}}, 1)$ such that $\eta_2 < \rho_1^{2(C_0+J)}$, which is possible since $\eta_2 < 1$ and $\rho_1^{2(C_0+J)} \nearrow 1$ as $\rho_1 \nearrow 1$.

Next choose B > A large enough such that

$$B \ge \frac{\eta_2 A \zeta \rho_1^{-2(C_0+J)} + 2K}{1 - \eta_2 \rho_1^{-2(C_0+J)}}$$

and finally pick $\rho \in (\rho_1, 1)$ such that for any $l = 1, \dots, \zeta$

$$\rho^{-(C_0+J)} \le \frac{Al+B-K}{A(l-1)+B}$$

which is possible since B > A > K and $\rho^{-(C_0+J)} \searrow 1$ as $\rho \nearrow 1$.

It remains to show that the three inequalities (5.7), (5.8), and (5.9) are satisfied. The constant *A* was chosen to satisfy inequality (5.7) but we need to check the other two. For inequality (5.8);

$$(Ad(a,a') - A + B)\rho^{-(C_0+J)} + K\left(\frac{\lambda^{\frac{1}{2}}}{\rho}\right)^{D(a,a',b)} \\ \leq (Ad(a,a') - A + B)\rho^{-(C_0+J)} + K \\ \leq A d(a,a') + B$$

where the first inequality uses $\rho \ge \rho_1 \ge \lambda^{\frac{1}{2}}$ and the second inequality follows from the choice of ρ , since $d(a, a') \le \zeta$. For inequality (5.9);

$$\eta_{2}(A\zeta + B)\rho^{-2(C_{0}+J)} + 2K\left(\frac{\lambda^{\frac{1}{2}}}{\rho}\right)^{D(a,a',b)} \leq \eta_{2}(A\zeta + B)\rho_{1}^{-2(C_{0}+J)} + 2K$$
$$\leq B$$
$$\leq A d(a,a') + B$$

where the first inequality uses $\rho \ge \rho_1 \ge \lambda^{\frac{1}{2}}$, and the second inequality uses the choice of *B*.

Now we can finally prove proposition 5.11;

Proof of Proposition 5.11. Note that if a = a' or b = b' then R(a, a', b, b') = 0 so we may assume they are distinct. In particular $d(a, a') \ge 1$. Set

$$\xi = 2 \max\left\{J+1, \frac{\ln 2 - \ln L}{\ln \lambda}\right\} + 2\zeta + 2\zeta$$

Then use lemma 5.24 to get constants A, B > 0 and $\rho \in (0, 1)$ satisfying the inequalities (5.7), (5.8), and (5.9).

We induct on D(a, a', b).

For $D(a, a', b) \leq \xi$ lemma 5.12 says $|R(a, a', b, b')| \leq 2\xi + 2$ but then by the choice of A we know $|R(a, a', b, b')| \leq A\lambda^{\frac{1}{2}\xi}$. We picked $\rho \geq \lambda^{\frac{1}{2}}$ hence

$$|R(a, a', b, b')| \le A\rho^{\xi} \le A\rho^{D(a, a', b)} \le (A \, d(a, a') + B)\rho^{D(a, a', b)}$$

For the inductive step assume $D(a, a', b) \ge \xi + 1$.

If |d(a, b) - d(a', b)| > J - I then lemma 5.13 combined with inequality (5.8) gives

$$|R(a, a', b, b')| \le (A d(a, a') + B) \rho^{D(a, a', b)}$$

as desired. Otherwise $|d(a, b) - d(a', b)| \le J - I$ and then use lemma 5.14 and inequality (5.9).

Thus we have finished the proof of proposition 5.11 and with that we have also proven proposition 5.10.

5.3 A metric on vertices

With the results we have already established we can now follow the arguments in [MY02] to define a metric. First we must symmetrise the function.

Definition 5.25. Let \mathcal{G} be a uniformly fine, δ -hyperbolic graph. Define a function $s: V \times V \to \mathbb{R}_{\geq 0}$ by

$$s(x,y) := \frac{1}{2} (r(x,y) + r(y,x)).$$

Some properties of *r* immediately carry over to *s* and we summarise these properties in the following lemma.

Lemma 5.26. *Let* G *be a uniformly fine,* δ *-hyperbolic graph.*

- (*i*) The function $s: V \times V \to \mathbb{R}_{\geq 0}$ is invariant under the diagonal action of $\text{Isom}(\mathcal{G})$ on $V \times V$.
- (*ii*) There exists a constant $J + 1 \ge 1$ such that for all $a, b \in V$ we have

$$\frac{1}{J+1}d(a,b) \le s(a,b) \le d(a,b).$$

(iii) There exist constants $N, M \ge 1$ such that for all $a, a', b \in V$

$$|s(a,b) - s(a',b)| \le \frac{1}{2}(1+M)d(a,a') + \frac{1}{2}N.$$

(*iv*) There exists a constant $C_1 > 0$ such that for $a, b \in V$ and $x \in V[a, b]$

$$|s(a,b) - s(a.x) - s(x,b)| \le C_1.$$

(v) There exists constants C > 0 and $\omega \in (0,1)$ such that for $a, a', b, b' \in V$ if $d(a, a') \le 1$ and $d(b, b') \le 1$ then

$$|s(a,b) - s(a',b) - s(a,b') + s(a',b')| \le C\omega^{d(a,b)}.$$

Proof. The function is $Isom(\mathcal{G})$ -invariant since r is $Isom(\mathcal{G})$ -invariant. The second part comparing s to d follows from proposition 5.2. The third part combines lemma 5.5 and proposition 5.8. The fifth part, which is a double difference type result, comes from proposition 5.10. The fourth part, which says geodesics with respect to d are almost geodesic with respect to the wannabe metric s, uses a degenerate form of lemma 5.7, where $b_0 = x$ and $\tilde{a} = a$, to bound

$$|r(a,b) - r(a,x) + r(x,b)| \le C_1.$$

We have to apply lemma 5.7 a second time, swapping the roles of a and b to get

$$|r(b,a) - r(b,x) - r(x,a)| \le C_1$$

and then taking the average gives the inequality in *s*.

For a metric we need the triangle inequality to hold. We don't have it yet but we do have the following lemma.

Lemma 5.27. There exists a constant $C_2 > 0$ such that for all $x, y, z \in V$,

$$s(x,z) \leq s(x,y) + s(y,z) + C_2.$$

Proof. The idea for the proof comes from [MY02, Proposition 11]. Let Δ be a geodesic triangle with corners *x*, *y*, *z*, and let ι_x , ι_y , ι_z be the inner points of Δ , as in definition 2.8. Then

$$\begin{split} s(x,z) &\leq s(x,\iota_y) + s(\iota_y,z) + C_1 \\ &\leq \left(s(x,\iota_z) + \frac{1}{2}(1+M)d(\iota_z,\iota_y) + \frac{1}{2}N \right) \\ &+ \left(\frac{1}{2}(1+M)d(\iota_y,\iota_x) + \frac{1}{2}N + s(\iota_x,z) \right) + C_1 \\ &\leq s(x,\iota_z) + s(\iota_x,z) + (1+M)\delta + N + C_1 \\ &\leq (s(x,y) + C_1) + (s(y,z) + C_1) + (1+M)\delta + N + C_1 \end{split}$$

where the first and fourth inequalities use lemma 5.26(iv), the second inequality uses lemma 5.26(iii), and the third inequality uses δ -hyperbolicity. Then the lemma is finished by setting $C_2 = (1 + M)\delta + N + 3C_1$.

This is not quite the triangle inequality, but we can now define our metric.

Definition 5.28. Let \mathcal{G} be a uniformly fine, δ -hyperbolic graph. Define a function $\hat{d}: V \times V \to \mathbb{R}$ by

$$\hat{d}(x,y) = \begin{cases} s(x,y) + C_2 & \text{if } x \neq y \\ 0 & \text{if } x = y \end{cases}$$

where C_2 is the constant from lemma 5.27.

Theorem 5.29. Let G be a uniformly fine, δ -hyperbolic graph. The function \hat{d} is a metric on V. Moreover it satisfies the following properties;

- (*i*) It is invariant under the action of $\text{Isom}(\mathcal{G})$ on V.
- (*ii*) Let $C_2 \ge 0$ be the constant from lemma 5.27. For all $a, b \in V$

$$\frac{1}{J+1}d(a,b) \leq \hat{d}(a,b) \leq d(a,b) + C_2.$$

(iii) There exists a constant $C_3 > 0$ such that for $a, b \in V$ and $x \in V[a, b]$

$$0 \leq \hat{d}(a,x) + \hat{d}(x,b) - \hat{d}(a,b) \leq C_3.$$

(iv) There exist constants $C > 0, \omega \in (0,1)$ such that for all $a, a', b, b' \in V$ if $d(a, a') \le 1$ and $d(b, b') \le 1$ then

$$\left|\hat{d}(a,b) - \hat{d}(a',b) - \hat{d}(a,b') + \hat{d}(a',b')\right| \le C\omega^{d(a,b)}.$$
(5.10)

Proof. It is symmetric since *s* is defined to be symmetric, and it satisfies the triangle inequality because of lemma 5.27. If $x \neq y \in V$ then by definition $\hat{d}(x, y) \ge C_2 > 0$ so it is positive-definite. Therefore \hat{d} is a metric on *V*.

Moreover it is invariant under graph automorphisms of G because *s* is invariant.

For any $a \neq b \in V$, lemma 5.26(*ii*) says $\frac{1}{J+1}d(a,b) \leq s(a,b) \leq d(a,b)$ and thus we obtain part (*ii*).

Part (*iii*) follows from lemma 5.26(*iv*), if we set $C_3 = C_1 + 2C_2$. (Note that $\hat{d}(a, x) + \hat{d}(x, b) - \hat{d}(a, b)$ is always non-negative by the triangle inequality.)

For part (*iv*) we can use lemma 5.26(v) to get

$$|s(a,b) - s(a',b) - s(a,b') + s(a',b')| \le C\omega^{d(a,b)}$$

If all four vertices are distinct then the four copies of C_2 cancel out and so we get inequality (5.10). If a = a' or b = b' then the left-hand side of inequality (5.10) is zero. Finally if one of $\{a, a'\}$ is equal to one of $\{b, b'\}$ then diam $\{a, a', b, b'\} \le 2$ (with respect to the metric *d*) so the left-hand side of inequality (5.10) is bounded by $8 + 4C_2$ (using part (*ii*)). If we increase *C* (if necessary) such that $8 + 4C_2 \le C\omega^2$ then inequality (5.10) holds here too.

Remark 5.30. In part (*iii*) of theorem 5.29 the notation V[a, b] still refers to geodesics with respect to the original metric d on G. When we talk about actualy geodesics we will only ever consider geodesics in G with respect to the word metric.

We can now define Gromov products and the double difference in terms of this new metric.

Definition 5.31. Let \mathcal{G} be a uniformly fine, δ -hyperbolic graph. Let \hat{d} be the metric on V from theorem 5.29. Define the *Gromov product* (with respect to \hat{d}) of three vertices $a, a', b \in V$ to be

$$\langle a|a'\rangle_b = \frac{1}{2} \big(\hat{d}(a,b) + \hat{d}(a',b) - \hat{d}(a,a')\big).$$

Define the *double difference* (with respect to \hat{d}) of four vertices $a, a', b, b' \in V$ by

$$\langle a, a' | b, b' \rangle = \frac{1}{2} \left(\hat{d}(a, b) - \hat{d}(a', b) - \hat{d}(a, b') + \hat{d}(a', b') \right).$$

Here we are using angular brackets to distinguish from the Gromov product and double difference defined using the word metric on the graph. We could define the double difference in terms of the Gromov product, as in definition 3.15. The generic facts about the double difference in proposition 3.16 still hold. Moreover theorem 5.29 gives us a couple of facts for this new Gromov product and new double difference.

Theorem 5.32. *Let* G *be a uniformly fine,* δ *-hyperbolic graph.*

(*i*) There exists a constant $C_3 > 0$ such that for $a, b \in V$ and $x \in V[a, b]$

$$0\leq \langle a|b\rangle_x\leq \frac{1}{2}C_3.$$

(*ii*) There exist constants $C > 0, \omega \in (0,1)$ such that for all $a, a', b, b' \in V$ if $d(a, a') \le 1$ and $d(b, b') \le 1$ then

$$\left|\langle a,a'|b,b'\rangle\right| \le \frac{1}{2}C\omega^{d(a,b)}.$$
(5.11)

Note that if *x* lies on a geodesic (with respect to the word metric *d* on \mathcal{G}) between *a* and *b* then $(a|b)_x = 0$. Part (*i*) of theorem 5.29 says that $\langle a|b\rangle_x$ is bounded by some global constant.

Before showing how to extend this metric to the boundary of the graph, we need to extend the metric from the vertex set to all of the graph. More generally, in section 6.1 we will show how to extend a metric on the vertex set of a simplicial complex to the whole simplicial complex, and then show in section 6.2 that if we start with the metric \hat{d} constructed here (see theorem 5.29) then we can extend the double difference to the boundary (theorem 6.13).

6 The Metric on the Flow Space

In chapter 5 we defined a new metric \hat{d} on the vertex set of our space X but to proceed with the construction from section 3.4 of a metric on the flow space of X we need to extend the metric \hat{d} to the whole simplicial complex X.

In section 6.1 we will work more generally, with an arbitrary simplicial complex *X* and a metric \hat{d} defined on the vertices of *X*, such that \hat{d} is (A, B)-quasi-isometric to the word metric $d_{\mathcal{G}}$, where \mathcal{G} is the 1-skeleton of *X*. Then in section 6.2 we will prove some properties of this metric on *X* when we start with the \hat{d} defined in section 5.3.

6.1 Extending a metric

In [Min05] Mineyev defined an extension of a metric \hat{d} by taking a bilinear expansion with respect to the barycentric coordinates, see [Min05, Equation (31) on page 436]. If $x = \sum_{u \in V} x_u u$ and $y = \sum_{v \in V} y_v v$ then define

$$\hat{d}(x,y) = \sum_{u,v \in V} x_u y_v \, \hat{d}(u,v).$$
 (6.1)

This extension was also used by Bartels-Lück-Reich in their proof of The Farrell-Jones Conjecture for hyperbolic groups (see [BLR08a, Section 6]). However, this extension does not define a metric, since it does not always satisfy the condition $\hat{d}(x, x) = 0$.

For example, suppose $x \in X$ is not a vertex. Then we can find two vertices $u, v \in V$ such that in barycentric coordinates the values x_u, x_v are

both non-zero. Thus

$$\hat{d}(x,x) \ge x_u x_v \,\hat{d}(u,v) > 0.$$

In this section we show how to define a metric \tilde{d} on X whose restriction to the vertex set coincides with \hat{d} and moreover if $x, y \in X$ are sufficiently far apart then $\tilde{d}(x, y) = \hat{d}(x, y)$, i.e. the distance is given by the bilinear expansion of the barycentric coordinates (see theorem 6.9).

Since the applications of the metric in [Min05] and [BLR08a] only make use of the large-scale geometry of the metric, their arguments work with \tilde{d} instead of \hat{d} , requiring only minor tweaking. For example [Min05, Proposition 38], which was quoted by Bartels-Lück-Reich as [BLR08a, Proposition 6.4], can be tweaked as demonstrated in proposition 7.2, which explains how to modify the proof given by Mineyev to use the extension \tilde{d} instead of \hat{d} . Therefore the results obtained in those papers are valid.

Even though the bilinear extension is not a metric, it nevertheless satisfies the triangle inequality;

Lemma 6.1. Let X be a simplicial complex and let \hat{d} be a metric on the vertex set of X. Extend \hat{d} to a function on X × X by bilinearly extending, as in equation (6.1). Then for all $x, y, z \in X$,

$$\hat{d}(x,z) \le \hat{d}(x,y) + \hat{d}(y,z).$$

Proof. Write *x*, *y*, *z* in barycentric coordinates, and then

$$\begin{split} \hat{d}(x,z) &= \sum_{u,w \in X^{(0)}} x_u z_w \, \hat{d}(u,w) \\ &= \sum_{u,v,w \in X^{(0)}} x_u y_v z_w \, \hat{d}(u,w) \\ &\leq \sum_{u,w \in X^{(0)}} x_u y_v z_w (\hat{d}(u,v) + \hat{d}(v,w)) \\ &= \sum_{u,v,w \in X^{(0)}} x_u y_v z_w \, \hat{d}(u,v) + \sum_{u,v,w \in X^{(0)}} x_u y_v z_w \, \hat{d}(v,w) \\ &= \sum_{u,v \in X^{(0)}} x_u y_v \, \hat{d}(u,v) + \sum_{v,w \in X^{(0)}} y_v z_w \, \hat{d}(v,w) \\ &= \hat{d}(x,y) + \hat{d}(y,z). \end{split}$$

So this bilinear extension is *almost* a metric. It only fails to be a metric locally, with the problem that $\hat{d}(x, x)$ need not be zero. To construct an actual metric we will take the minimum of this bilinear extension and a genuine metric on *X*. We want this to extend the metric on the vertex set so we want the genuine metric on *X* to be greater than \hat{d} on the vertex set.

The metric we will use is a length metric induced by the l^1 -metric on each simplex.

Definition 6.2. Let *X* be a simplicial complex. For any simplex σ of *X* let d_{σ}^1 be the l^1 -metric on σ , i.e. for all points $x, y \in \sigma$,

$$d_{\sigma}^{1}(x,y) := \frac{1}{2} \sum_{u \in \sigma^{(0)}} |x_{u} - y_{u}|.$$
(6.2)

The factor $\frac{1}{2}$ is to ensure that the distance between adjacent vertices is 1.

Then for $x, y \in X$ define a *path in* X *from* x *to* y to be a sequence of points $x = a_0, a_1, \ldots, a_r = y$ in X such that for every $i = 1, \ldots, r$ there is a simplex σ_i of X that contains both a_{i-1} and a_i . The *length* of such a path is the sum $\sum_{i=1}^r d_{\sigma_i}^1(a_{i-1}, a_i)$.

The length of a path does not depend on how we choose the simplices σ_i because if a simplex τ is a subsimplex of σ then $d_{\tau}^1 \equiv d_{\sigma}^1|_{\tau}$.

Then we define a metric on *X* by setting $d_X^1(x, y)$ to be the infimum of lengths of all such paths from *x* to *y*.

We start by giving some properties of this metric.

Proposition 6.3. Let X be a simplicial complex.

- (*i*) For any simplex σ in X, the restriction of d_X^1 to σ coincides with d_{σ}^1 .
- *(ii) The diameter of any simplex is 1, unless the simplex is a vertex.*
- (iii) If $x, y \in X$ have disjoint supports then $d_X^1(x, y) \ge 1$.

Proof. (*i*) Given any two points *x*, *y* contained in one simplex σ , by considering the trivial path $x = a_0, a_1 = y$ we get $d_X^1(x, y) \le d_{\sigma}^1(x, y)$ so we only need to show the inequality in the other direction.

Fix a simplex σ and two points $x, y \in \sigma$. Consider a path a_0, a_1, \ldots, a_r from x to y in X. Since it is a path we can find simplices σ_i for $i = 1, \ldots, r$ such that σ_i contains both a_{i-1} and a_i . Let $Y \subseteq X$ be the union of all the σ_i (which is a subcomplex of X) and denote the vertex set of Y by Y_0 and let $\Delta = \Delta^{Y_0-1}$ be the standard simplex of dimension $Y_0 - 1$, where we identify the vertex set of Δ with Y_0

There is a canonical inclusion $Y \to \Delta$, and so we can think of Y as a subcomplex of Δ . Thus for each *i* we have $d^1_{\sigma_i}(a_{i-1}, a_i) = d^1_{\Delta}(a_{i-1}, a_i)$ and then the triangle inequality tells us that the length of the path a_0, \ldots, a_r is at least $d^1_{\Lambda}(a_0, a_r) = d^1_{\sigma}(x, y)$.

We can do this for any path from *x* to *y* so $d_X^1(x, y) \ge d_{\sigma}^1(x, y)$. Therefore we can conclude that $d_X^1|_{\sigma} \equiv d_{\sigma}^1$.

(*ii*) It follows from equation (6.2) that if two points x, y lie in a common simplex σ then the distance between them is $d_{\sigma}^{1}(x, y) \leq 1$. Taking x, y to be distinct vertices of σ shows that the bound is strict.

(*iii*) If v is a vertex in the support of x but not in the support of y then any path from x to y must eventually take the v-coordinate down from x_v

to zero. Formally, if a_0, \ldots, a_r is a path from x to y let $a_{i,v}$ denote the v-coordinate of a_i . Then the v-coordinate contributes

$$\sum_{i=1}^{r} \frac{1}{2} |a_{i-1,v} - a_{i,v}| \ge \frac{1}{2} \left| \sum_{i=1}^{r} (a_{i-1,v} - a_{i,v}) \right|$$
$$= \frac{1}{2} |a_{0,v} - a_{r,v}| = \frac{1}{2} x_{v}$$

and thus the *v*-coordinate contributes at least $\frac{1}{2}x_v$ to $d_X^1(x, y)$.

Similarly if *w* is in the support of *y* but not the support of *x* then any path must take the *w*-coordinate from zero up to y_w , hence the *w*-coordinate contributes at least $\frac{1}{2}y_w$ to $d_X^1(x, y)$.

Therefore if the support of *x* is disjoint from the support of *y* then we get at least $\frac{1}{2}$ from the coordinates in the support of *x* and at least $\frac{1}{2}$ from the coordinates in the support of *y*. Thus $d_X^1(x, y) \ge 1$.

If take the minimum of the bilinear extension (6.1) and this metric d_X^1 then we could encounter two problems, namely the triangle inequality might not be satisfied and this minimum might not restrict to \hat{d} on the vertex set.

If the metric \hat{d} is quasi-isometric to the word metric on the 1-skeleton of *X* then we can fix both problems by re-scaling the metric d_X^1 .

Definition 6.4. Let *X* be a simplicial complex and let \hat{d} be a metric on the vertices of *X* that is (A, B)-quasi-isometric to the word metric induced by the 1-skeleton of *X*. Set $\hat{d}: X \times X \to \mathbb{R}$ to be the bilinear extension of \hat{d} as in equation (6.1). Define a function $\tilde{d}: X \times X \to \mathbb{R}$ by

$$\tilde{d}(x,y) := \min\left\{\hat{d}(x,y), 3(A+B)d_X^1(x,y)\right\}.$$
(6.3)

To prove that this function satisfies the triangle inequality we will use the following lemma.

Lemma 6.5. Let X be a simplicial complex and let \hat{d} be a metric on the vertices of X that is (A, B)-quasi-isometric to the word metric induced by the 1-skeleton of X. If $\hat{d}: X \times X \to \mathbb{R}$ is the bilinear extension of \hat{d} as in equation (6.1) then for all $x, y, z \in X$,

$$\hat{d}(x,z) \le \hat{d}(x,y) + 2(A+B)d_X^1(y,z).$$
 (6.4)

Proof. First we will consider the special case where y and z lie in a common simplex σ . By definition, we can expand bilinearly and obtain

$$\begin{split} \hat{d}(x,z) &= \sum_{u,w \in X^{(0)}} x_u z_w \, \hat{d}(u,w) \\ &= \sum_{u,w \in X^{(0)}} x_u y_w \, \hat{d}(u,w) + \sum_{u,w \in X^{(0)}} x_u (z_w - y_w) \, \hat{d}(u,w) \\ &= \hat{d}(x,y) + \sum_{u,w \in X^{(0)}} x_u (z_w - y_w) \, \hat{d}(u,w). \end{split}$$

Fix a vertex v_0 of σ . Then $\hat{d}(u, w) \leq \hat{d}(u, v_0) + \hat{d}(v_0, w)$ for any vertices u, w of X. The distance $\hat{d}(u, v_0)$ is independent of the vertex w so we have $\sum_{u,w \in X^{(0)}} x_u(z_w - y_w) \hat{d}(u, v_0) = 0$. Moreover, since v_0 is joined to any vertex of σ by an edge, we know $\hat{d}(v_0, w) \leq A + B$. Therefore

$$\sum_{u,w \in X^{(0)}} x_u(z_w - y_w) \, \hat{d}(u, w) \le \sum_{u,w \in X^{(0)}} x_u |z_w - y_w| (A + B)$$
$$= 2(A + B) d_{\sigma}^1(y, z).$$

But then from proposition 6.3(*i*) we get $d_{\sigma}^{1}(y, z) = d_{X}^{1}(y, z)$.

Putting all this together gives the desired inequality in this special case. For the general case, consider a path $y = a_0, a_1, \ldots, a_r = z$ from y to z (in the sense of definition 6.2). Then

$$\hat{d}(x,z) = \hat{d}(x,y) + \sum_{i=1}^{r} \left(\hat{d}(x,a_i) - \hat{d}(x,a_{i-1}) \right)$$

We know that a_{i-1} and a_i lie in a common simplex so by the special case we have $\hat{d}(x, a_i) - \hat{d}(x, a_{i-1}) \le 2(A + B)d_X^1(a_i, a_{i-1})$. Thus

$$\hat{d}(x,z) \le \hat{d}(x,y) + \sum_{i=1}^{r} 2(A+B)d_X^1(a_i,a_{i-1}) \\ = \hat{d}(x,y) + 2(A+B)l(\underline{a})$$

where l(a) is the length of the path a_0, \ldots, a_r .

This holds for any path from *y* to *z* so we conclude

$$\hat{d}(x,z) \le \hat{d}(x,y) + 2(A+B)d_X^1(y,z).$$

With this lemma, we can prove that the function \tilde{d} is a metric on the simplicial complex.

Proposition 6.6. Let X be a simplicial complex and let \hat{d} be a metric on the vertices of X that is (A, B)-quasi-isometric to the word metric induced by the 1-skeleton of X. Set $\hat{d}: X \times X \to \mathbb{R}$ to be the bilinear extension of \hat{d} as in equation (6.1). Then \tilde{d} defined by equation (6.3) is a metric on X.

Proof. It is symmetric non-negative since it is the minimum of two functions that are both symmetric non-negative. For all $x \in X$,

$$0 \le \tilde{d}(x, x) \le 3(A + B)d_X^1(x, x) = 0$$

so $\tilde{d}(x,x) = 0$. For any two distinct points $x \neq y \in X$ then $d_X^1(x,y) > 0$ since d_X^1 is a metric on X. Moreover we can find vertices $u \neq v$ with $x_u \neq 0$ and $y_v \neq 0$, thus $\hat{d}(x,y) \ge x_u y_v \hat{d}(u,v) > 0$. Therefore $\tilde{d}(x,y) > 0$. The last thing to check for \tilde{d} to be a metric is the triangle inequality. Given three points $x, y, z \in X$ we need to show

$$\tilde{d}(x,z) \le \tilde{d}(x,y) + \tilde{d}(y,z)$$

Both d_X^1 and \hat{d} satisfy the triangle inequality on *X*, where we use lemma 6.1 for \hat{d} , and in the mixed case we can use lemma 6.5.

We want the metric \tilde{d} to restrict to \hat{d} on the vertex set. Hence we need to look at the metric d_X^1 on the vertex set.

Lemma 6.7. Let X be a simplicial complex, and let G be the 1-skeleton of X. For any $u, v \in V$ we have

$$d_{\mathcal{G}}(u,v) = d_X^1(u,v).$$
(6.5)

Proof. Any path in the 1-skeleton is also a path in *X*, so $d_{\mathcal{G}}(u, v) \ge d_X^1(u, v)$. It remains to prove that the length of any path in *X* from *u* to *v* is at least $d_{\mathcal{G}}(u, v)$.

For $k = 0, ..., d := d_{\mathcal{G}}(u, v)$ let $S_k = \{z \in V | d_{\mathcal{G}}(u, z) = k\}$. The idea is that the support of any path from *u* to *v* must meet every S_k and moreover the weight must come from *u* and through each S_k before arriving at *v* (the path could be longer but we are only looking for a lower bound).

Let $u = a_0, a_1, ..., a_r = v$ be a path in *X* from *u* to *v*. So for any *i* there is always a simplex σ_i that contains both a_{i-1} and a_i . Write each $a_i = \sum_{z \in V} a_i^z z$ in barycentric coordinates and for all *k* set

$$w_k(a_i) = \sum_{z \in S_k} a_i^z \in [0, 1]$$

which is the weight of the point a_i in the sphere S_k .

The length of this path <u>a</u> is

$$length(\underline{a}) = \sum_{i=1}^{r} d_{\sigma_{i}}^{1}(a_{i-1}, a_{i})$$

$$= \sum_{i=1}^{r} \sum_{z \in V} \frac{1}{2} |a_{i-1}^{z} - a_{i}^{z}|$$

$$\geq \frac{1}{2} \sum_{i=1}^{r} \sum_{k=0}^{d} \sum_{z \in S_{k}} |a_{i-1}^{z} - a_{i}^{z}|$$

$$\geq \frac{1}{2} \sum_{i=1}^{r} \sum_{k=0}^{d} \left| \sum_{z \in S_{k}} a_{i-1}^{z} - \sum_{z \in S_{k}} a_{i}^{z} \right|$$

$$= \frac{1}{2} \sum_{k=0}^{d} \sum_{i=1}^{r} |w_{k}(a_{i-1}) - w_{k}(a_{i})|.$$

Now we try to find a lower bound for the sums $\sum_{i=1}^{r} |w_k(a_{i-1}) - w_k(a_i)|$.

If k = 0 then

$$\sum_{i=1}^{r} |w_0(a_{i-1}) - w_0(a_i)| \ge \left| \sum_{i=1}^{r} (w_0(a_{i-1}) - w_0(a_i)) \right|$$
$$= |w_0(a_0) - w_0(a_r)|$$
$$= 1$$

since $a_0 = u \in S_0$ and $a_r = v \in S_d$. Similarly for k = d we get

$$\sum_{i=1}^{r} |w_d(a_{i-1}) - w_d(a_i)| \ge |w_d(a_0) - w_d(a_r)| = 1.$$

So we are left with $1 \le k < d$. We claim that for such a k we can find some $i_k \in \{1, ..., d-1\}$ such that $w_k(a_{i_k}) = 1$.

If for now we assume the claim is true, then

$$\sum_{i=1}^{r} |w_k(a_{i-1}) - w_k(a_i)|$$

$$= \sum_{i=1}^{i_k} |w_k(a_{i-1}) - w_k(a_i)| + \sum_{i=i_k+1}^{r} |w_k(a_{i-1}) - w_k(a_i)|$$

$$\ge \left|\sum_{i=1}^{i_k} (w_k(a_{i-1}) - w_k(a_i))\right| + \left|\sum_{i=i_k+1}^{i_k} (w_k(a_{i-1}) - w_k(a_i))\right|$$

$$= |w_k(a_0) - w_k(a_{i_k})| + |w_k(a_{i_k}) - w_k(a_r)|$$

$$= 2$$

and we would get

$$\begin{split} \operatorname{length}(\underline{a}) &\geq \frac{1}{2} \sum_{k=0}^{d} \sum_{i=1}^{r} |w_k(a_{i-1}) - w_k(a_i)| \\ &\geq \frac{1}{2} (1 + 2(d-2) + 1) \\ &= d = d_G(u, v). \end{split}$$

Therefore, to finish the proof of the lemma we need to prove the claim that for all k = 1, ..., d - 1 there is some i_k with $w_k(a_{i_k}) = 1$.

For all *i* there is a simplex containing both a_i and a_{i-1} hence every vertex in the support of a_i is joined to any vertex in the support of a_{i-1} by an edge. In particular, if $w_{k-1}(a_{i-1}) \neq 0$ then $w_{k+1}(a_i) = 0$ because we may move at most one further away from *u* with successive a_i 's. So if we pick i_k minimal with $w_{k+1}(a_{i_k+1}) \neq 0$ then $w_{k-1}(a_{i_k}) = 0$ and $w_{k+1}(a_{i_k}) = 0$. So all the weight of a_{i_k} has to be at the sphere S_k , but $\sum_k w_k(a_{i_k}) = 1$. Therefore $w_k(a_{i_k}) = 1$.

This proves the claim, and with it also finishes the proof of the lemma by the earlier argument. $\hfill \Box$
Now we can prove that \hat{d} restricts to \hat{d} on vertices. In fact we can prove a stronger result.

Lemma 6.8. Let X be a simplicial complex, and let \mathcal{G} denote its 1-skeleton. Let \hat{d} be a metric on the vertices of X that is (A, B)-quasi-isometric to the word metric $d_{\mathcal{G}}$. For all $x, y \in X$ if the support of x is disjoint from the support of y then

$$\tilde{d}(x,y) = \hat{d}(x,y).$$

In particular, \tilde{d} coincides with \hat{d} on the vertex set.

Proof. Start by considering vertices $u, v \in V$. If $u \neq v$ then $d_{\mathcal{G}}(u, v) \geq 1$. Moreover, by assumption we have $\hat{d}(u, v) \leq A d_{\mathcal{G}}(u, v) + B$. Therefore

$$\hat{d}(u,v) \leq A d_{\mathcal{G}}(u,v) + B$$

 $\leq (A+B) d_{\mathcal{G}}(u,v)$

since $B \ge 0$ and then applying lemma 6.7 yields

$$\hat{d}(u,v) \le (A+B) d_X^1(u,v). \tag{(\star)}$$

Note that if u = v then (*) is trivially true, so inequality (*) holds for all vertices $u, v \in V$.

Now consider the general case $x, y \in X$ with disjoint supports. Since the support of x is disjoint from the support of y, we must have $d_X^1(x, y) \ge 1$ by proposition 6.3(*iii*). Moreover, for any $u \in \text{supp}(x)$ proposition 6.3(*ii*) tells us that $d_X^1(u, x) \le 1$, and similarly for $v \in \text{supp}(y)$. Therefore

$$\begin{split} \hat{d}(x,y) &= \sum_{u,v \in V} x_u y_v \, \hat{d}(u,v) \\ &\leq \sum_{u,v \in V} x_u y_v (A+B) d_X^1(u,v) \\ &\leq \sum_{u,v \in V} x_u y_v (A+B) \left(d_X^1(u,x) + d_X^1(x,y) + d_X^1(y,v) \right) \\ &\leq \sum_{u,v \in V} x_u y_v (A+B) 3 d_X^1(x,y) \\ &= 3(A+B) d_X^1(x,y) \end{split}$$

and then $\tilde{d}(x, y) = \hat{d}(x, y)$ since \tilde{d} is defined to be the minimum of these two expressions.

To summarise this section, we have the following theorem;

Theorem 6.9. Let X be a simplicial complex, and let \mathcal{G} denote its 1-skeleton. Let \hat{d} be a metric on the vertex set V of X. If \hat{d} is quasi-isometric to $d_{\mathcal{G}}$ then there exists a metric \tilde{d} on all of X such that for any $x, y \in X$ if the support of x is disjoint from the support of y then $\tilde{d}(x, y) = \hat{d}(x, y)$. In particular, $\tilde{d} \equiv \hat{d}$ on vertices.

Moreover, if \hat{d} is preserved under simplicial automorphisms of X then so is \hat{d} .

We have a metric on all of *X*, and we want to know that it doesn't change the large-scale geometry from \hat{d} .

Proposition 6.10. Let X be a simplicial complex, and let \mathcal{G} denote its 1-skeleton. Let \hat{d} be a metric on the vertex set V of X that is (A, B)-quasi-isometric to $d_{\mathcal{G}}$. The metric \tilde{d} from theorem 6.9 is (1, 2(A + B))-quasi-isometric to \hat{d} .

Proof. Fix $x, y \in X$. If the support of x is disjoint from the support of y then $\tilde{d}(x, y) = \hat{d}(x, y)$, so suppose there exists some $v \in \text{supp}(x) \cap \text{supp}(y)$. Then

$$\hat{d}(x,v) = \sum_{u \in V} x_u \hat{d}(u,v) \le \sum_{u \in V} x_u (A+B) = (A+B)$$

and similarly $\hat{d}(v, y) \leq (A + B)$. Therefore

$$\begin{split} \tilde{d}(x,y) &\leq \tilde{d}(x,v) + \tilde{d}(v,y) \\ &\leq \hat{d}(x,v) + \hat{d}(v,y) \leq 2(A+B) \leq \hat{d}(x,y) + 2(A+B). \end{split}$$

Moreover,

$$\hat{d}(x,y) \leq \hat{d}(x,v) + \hat{d}(v,y) \leq 2(A+B) \leq \tilde{d}(x,y) + 2(A+B).$$

So

$$\hat{d}(x,y) - 2(A+B) \le \tilde{d}(x,y) \le \hat{d}(x,y) + 2(A+B).$$

6.2 Extending our metric

If *X* is a simplicial complex whose 1-skeleton \mathcal{G} is a uniformly fine, Gromov hyperbolic graph then the metric \hat{d} defined in section 5.3 is quasiisometric to the word metric (theorem 5.29(*ii*)) so theorem 6.9 provides an extension \tilde{d} , which is defined on all of *X*.

This extension is preserved under simplicial automorphisms of *X* since \hat{d} is preserved under simplicial automorphisms of *X*.

We constructed the metric \hat{d} to have better convergence properties towards the boundary, which was formulated in terms of the double difference in theorem 5.32(*ii*), and we want the metric \tilde{d} to also satisfy these better convergence properties.

As with the metrics *d* and \hat{d} we can define the *Gromov product with respect* to \tilde{d} of three points $x, y, z \in X$ to be

$$\langle x \wr y \rangle_z = \frac{1}{2} \big(\tilde{d}(x, z) + \tilde{d}(y, z) - \tilde{d}(x, y) \big)$$

and the *double difference with respect to* \tilde{d} of four points $x, x', y, y' \in X$ to be

$$\langle x, x' \wr y, y' \rangle = \frac{1}{2} \left(\tilde{d}(x, y) - \tilde{d}(x, y') - \tilde{d}(x', y) + \tilde{d}(x', y') \right).$$

Remark 6.11. We can relate the double difference with respect to \tilde{d} to the double difference with respect to \hat{d} : Theorem 5.29(*ii*) tells us that \hat{d} is quasi-isometric to d, so proposition 6.10 says that there is some constant $B' \ge 0$ such that \tilde{d} is (1, B')-quasi-isometric to \hat{d} . Hence for any $x, x', y, y' \in X$;

$$\langle x, x'|y, y'\rangle - 2B' \le \langle x, x' \wr y, y'\rangle \le \langle x, x'|y, y'\rangle + 2B'.$$
(6.6)

The inequality from theorem 5.32(ii) holds for the double difference with respect to \tilde{d} , since the metric \tilde{d} agrees with \hat{d} on the vertex set, i.e. we have the following;

Proposition 6.12. Let X be a simplicial complex whose 1-skeleton is a uniformly fine, Gromov hyperbolic graph. Let \hat{d} be the metric from section 5.3 and let \tilde{d} be the extension given by theorem 6.9. There exist constants C > 0 and $\omega \in (0,1)$ such that for all $a, a', b, b' \in V$ if $d(a, a') \leq 1$ and $d(b, b') \leq 1$ then

$$\left|\langle a,a' \wr b,b' \rangle\right| \leq \frac{1}{2} C \omega^{d(a,b)}.$$
(6.7)

If we apply theorem 6.9 to the word metric of the 1-skeleton \mathcal{G} of X then we get a metric $\tilde{d}_{\mathcal{G}}$, and the metric space $(X, \tilde{d}_{\mathcal{G}})$ is Gromov hyperbolic, in the sense of definition 2.11. Therefore we can define the boundary $\partial_s X$ using sequences as in definition 2.29 define a topology on $\overline{X} := X \cup \partial_s X$ using remark 2.34. Now we can finally extend the double difference with respect to \tilde{d} from the vertex set to all of \overline{X} .

Theorem 6.13. Let X be a simplicial complex whose 1-skeleton G is a uniformly fine, Gromov hyperbolic graph. Set S to be the subset of $\overline{X}^4 := \overline{X} \times \overline{X} \times \overline{X} \times \overline{X}$ consisting of points (x, x', y, y') such that

- $x, y \in \partial X \Rightarrow x \neq y;$
- $x, y' \in \partial X \Rightarrow x \neq y';$
- $x', y \in \partial X \Rightarrow x' \neq y;$
- $x', y' \in \partial X \Rightarrow x' \neq y'$.

Then the double difference with respect to \tilde{d} extends continuously to S and satisfies

- (a) $\langle a, a' \wr b, b' \rangle = \langle b, b' \wr a, a' \rangle;$
- (b) $\langle a, a' \wr b, b' \rangle = -\langle a', a \wr b, b' \rangle = -\langle a, a' \wr b', b \rangle;$

- (c) $\langle a, a \wr b, b' \rangle = 0 = \langle a, a' \wr b, b \rangle;$
- (d) $\langle a, a' \wr b, b' \rangle + \langle a', a'' \wr b, b' \rangle = \langle a, a'' \wr b, b' \rangle;$
- (e) $\langle a, b \rangle c, x \rangle + \langle c, a \rangle b, x \rangle + \langle b, c \rangle a, x \rangle = 0;$
- (f) There are constants $\alpha \ge and \beta \ge 0$ such that for all vertices $a, a', b, b' \in V$

$$\frac{1}{\alpha}(a,a'|b,b') - \beta \leq \langle a,a' \wr b,b' \rangle \leq \alpha(a,a'|b,b') + \beta.$$

Proof. This is essentially [Min05, Theorem 35]. There the double difference is extended further to a larger subset of \overline{X}^4 , but the proof works by first extending to the set *S*. To extend it further requires allowing the double difference to take values $\pm \infty$, but here we want the double difference to only take values in \mathbb{R} so we only extend to the smaller set *S*.

A couple of modifications must be made to the proof given by Mineyev in [Min05]. Firstly, his definition of a hyperbolic complex requires the 1-skeleton to be uniformly locally finite, but since we have theorem 5.29 the only property required of the graph is Gromov hyperbolicity.

Secondly, his extension is given by taking the bilinear extension of the metric \hat{d} on the vertex set, but as noted earlier this does not give a metric. Hence we have used a different extension, namely \tilde{d} defined by equation (6.3). However, this extension \tilde{d} coincides with \hat{d} on the vertex set (see lemma 6.8). So we can use the argument in the proof of [Min05, Theorem 35] to extend the double difference to the boundary using sequences of vertices.

When considering sequences of arbitrary points (i.e. not necessarily vertices) we know from lemma 6.8 that $\tilde{d}(x, y) = \hat{d}(x, y)$ whenever the support of x is disjoint from the support of y. So if $(x_n)_{n \in \mathbb{N}}$ is a sequence of points that converges to $x \in \partial X$ and $(y_n)_{n \in \mathbb{N}}$ is a sequence of points that converges to $y \in \overline{X} \setminus \{x\}$ then for sufficiently large n, the support of x_n will be disjoint from the support of y_n and thus we can use the bilinear extension formula (6.1).

For part (f), Mineyev in [Min05, Theorem 35(h)] shows an equivalence between $\langle -, -|-, -\rangle$ and (-, -|-, -) but we can go to $\langle -, - \rangle -, -\rangle$ using remark 6.11.

We know that for points $x, y, z, w \in X$, the double difference relates to the Gromov product via the equation

$$\langle x, y \wr z, w \rangle = \langle y \wr z \rangle_x - \langle y \wr w \rangle_x \tag{6.8}$$

and, in particular $\langle x, y \rangle \langle z, x \rangle = \langle y \rangle \langle z \rangle_x$. Hence we can use theorem 6.13 to extend the Gromov product to the boundary.

Theorem 6.14. Let X be a simplicial complex whose 1-skeleton \mathcal{G} is a uniformly fine, Gromov hyperbolic graph. Set T to be the subset of $\overline{X}^3 := \overline{X} \times \overline{X} \times \overline{X}$ consisting of points (x, y, z) such that

- $x \notin \partial X$;
- $y, z \in \partial X \Rightarrow y \neq z$.

Then the Gromov product $\langle y \rangle z \rangle_x$ with respect to \tilde{d} extends continuously to T and satisfies

$$\langle a, a' \wr b, b' \rangle = \langle a' \wr b \rangle_a - \langle a' \wr b' \rangle_a \tag{6.9}$$

whenever $(a, a', b, b') \in S$ (where S is defined as in theorem 6.13) and $a \notin \partial X$.

Proof. If $(x, y, z) \in T$ then $(x, y, z, x) \in S$ and we can define the Gromov product in terms of the double difference: $\langle y \wr z \rangle_x := \langle x, y \wr z, x \rangle$.

Equation (6.9) follows from theorem 6.13(d), where the properties (a) and (b) are also used implicitly. \Box

This extension of the Gromov product coincides with the supliminf extension given in definition 2.31, since for any $x \in X$ and $y, z \in \overline{X}$ with $y \neq z$, if (y_n) and (z_n) are sequences in X converging to y and z respectively then the continuity of the double difference given by theorem 6.13 gives

$$\liminf_{n \to \infty} \langle y_n \wr z_n \rangle_x = \liminf_{n \to \infty} \langle x, y_n \wr z_n, x \rangle$$
$$= \langle x, y \wr z, x \rangle$$
$$= \langle y \wr z \rangle_x.$$

In particular, this means the facts from proposition 2.33 still hold.

We cannot extend $\langle -, - \rangle -, - \rangle$ to all of \overline{X}^4 because it would need to be $\pm \infty$ in places, as the next lemma demonstrates.

Lemma 6.15. Let X be a simplicial complex whose 1-skeleton G is a uniformly fine, Gromov hyperbolic graph. If $(x_n)_{n \in \mathbb{N}}$ is a sequence in X that converges to a point $x \in \partial X$ then for any $y, z \in \overline{X} \setminus \{x\}$ if $y \neq z$ then

$$\langle x, y \wr x_n, z \rangle \longrightarrow -\infty$$
 as $n \to \infty$.

Proof. Pick a sequence $(z_k)_{k\in\mathbb{N}}$ in X that converges to z. Then by continuity, as $k \to \infty$ we get $\langle z, z_k \wr y, x \rangle \longrightarrow \langle z, z \wr y, x \rangle = 0$. Therefore we can pick k large enough such that $|\langle z, z_k \wr y, x \rangle| \le 1$.

Combining theorems 6.13 and 6.14 gives

$$\begin{aligned} \langle x, y \wr x_n, z \rangle &= \langle z, x_n \wr y, x \rangle \\ &= \langle z, z_k \wr y, x \rangle + \langle z_k, x_n \wr y, x \rangle \\ &\leq 1 + \langle x_n \wr y \rangle_{z_k} - \langle x_n \wr x \rangle_{z_k}. \end{aligned}$$

Since the sequence $(x_n)_{n \in \mathbb{N}}$ converges to a point $x \in \partial_s X$ the Gromov product $\langle x_i \wr x_j \rangle_{z_k}$ tends up to ∞ as $i, j \to \infty$. Hence

$$\langle x_n \wr x \rangle_{z_k} \ge \liminf_{m \to \infty} \langle x_n \wr x_m \rangle_{z_k} \longrightarrow \infty \quad \text{as } n \to \infty.$$

By remark 2.32 we can pick sequences (x'_n) and (y_n) in X with $x'_n \to x$ and $y_n \to \infty$ as $n \to \infty$ such that $\lim_{n\to\infty} \langle x'_n \wr y_n \rangle_{z_k} = \langle x \wr y \rangle_{z_k}$. Then

$$\langle x'_n \wr y_n \rangle_{z_k} \geq \min\{\langle x'_n \wr x_n \rangle_{z_k}, \langle x_n \wr y \rangle_{z_k}, \langle y \wr y_n \rangle_{z_k}\} - 4\delta$$

by proposition 2.33(*ii*). Since the sequences (x_n) and (x'_n) both converge to $x \in \partial_s X$ proposition 2.33(*i*) says $\langle x'_n \wr x_n \rangle_{z_k} \to \infty$ as $n \to \infty$. Also, the Gromov product $\langle y \wr y_n \rangle_{z_k}$ tends to infinity, repeating the argument for $\langle x_n \wr x \rangle_{z_k}$ above. Therefore

$$\lim_{n \to \infty} \langle x_n \wr y \rangle_{z_k} \le \lim_{n \to \infty} \langle x'_n \wr y \rangle_{z_k} + 4\delta$$
$$= \langle x \wr y \rangle_{z_k} + 4\delta$$

which is finite since $x \neq y$.

Bringing this all together,

$$\lim_{n\to\infty} \langle x, y \wr x_n, z \rangle \leq 1 + \langle x \wr y \rangle_{z_k} + 2\delta - \lim_{n\to\infty} \langle x_n \wr x \rangle_{z_k}$$
$$= -\infty.$$

6.3 A metric on the flow space

In section 3.4 we outlined a plan to construct a metric on FS(X) for a metric space X. We needed to extend the double difference to points on the boundary and to do this we constructed a new metric \tilde{d} in the case that X is a simplicial complex whose 1-skeleton is a uniformly fine, δ -hyperbolic graph (see theorem 6.13).

For the rest of this section, we assume that *X* is such a simplicial complex, and that $x_0 \in X$ is a fixed base-point, and we try to apply the construction from section 3.4 to the metric space (X, \tilde{d}) .

Definition 6.16. We start by using the metric to interpret the parameter *t* of a point in the flow space (cf definition 3.14). For any point $(x, y, t) \in FS(X)$ define a map $\tilde{\theta}_{x,y}^{x_0} \colon \overline{\mathbb{R}} \to [-\langle y \wr x_0 \rangle_x, \langle x \wr x_0 \rangle_y]$ by

$$ilde{ heta}_{x,y}^{x_0}(t) = \left\{ egin{array}{ccc} -\langle y \wr x_0
angle_x & ext{if } t \leq -\langle y \wr x_0
angle_x \ t & ext{if } -\langle y \wr x_0
angle_x \leq t \leq \langle x \wr x_0
angle_y \ \langle x \wr x_0
angle_y & ext{if } \langle x \wr x_0
angle_y \leq t \end{array}
ight.$$

With this, we have an idea of how far $(x, y, t)_0$ (the fictitious point at time zero) is from the point of [x, y] that is 'closest' to the base-point. Using the family of maps $(\tilde{\theta}_{x,y}^{x_0}|x, y \in \overline{X})$ we can define a distance from $(x, y, t)_0$ to an arbitrary point $z \in X$.

Definition 6.17. Define a map l_{x_0} : $X \times FS(X) \rightarrow [0, \infty)$ by

$$l_{x_0}(z,(x,y,t)) = \langle x \wr y \rangle_z + \left| \widetilde{ heta}_{x,y}^{x_0}(t) - \langle x,y \wr z,x_0
angle
ight|.$$

This function is well-defined by theorems 6.13 and 6.14, since z and x_0 are both points in X, and not the boundary.

Remark 6.18. If $\underline{x} := [(x, x, 0)] \in FS(X)$ is a stationary point then for any $z \in X$ we have $l_{x_0}(z, \underline{x}) = \tilde{d}(z, x)$.

Thus we can say what the distance is from $(x, y, t)_0$ to z but we are interested in the distance from $(x, y, t)_0$ to $(x', y', t')_0$. We do this by comparing their distances to z.

Definition 6.19. Define a function β_{x_0} : $X \times FS(X) \times FS(X) \to \mathbb{R}$ by

$$\beta_{x_0}(z,c,c') = l_{x_0}(z,c) - l_{x_0}(z,c').$$
(6.10)

The next lemma states a couple of simple properties about the function β_{x_0} . The proofs are straight-forward and hence omitted.

Lemma 6.20. *The function* β_{x_0} *satisfies the following properties;*

- (*i*) β_{x_0} is invariant under the induced diagonal action of Isom(X) on FS(X);
- (*ii*) For any $x, x', z \in X$, $\beta_{x_0}(z, \underline{x}, \underline{x'}) = \tilde{d}(z, x) \tilde{d}(z, x');$
- (iii) β_{x_0} is anti-symmetric with respect to the $FS(X) \times FS(X)$ factors, i.e. for all $z \in X$ and for all $c, c' \in FS(X)$, $\beta_{x_0}(z, c, c') = -\beta_{x_0}(z, c', c)$;
- (*iv*) For all $c_1, c_2, c_3 \in FS(X)$ and all $z \in X$,

$$\beta_{x_0}(z,c_1,c_3) = \beta_{x_0}(z,c_1,c_2) + \beta_{x_0}(z,c_2,c_3).$$

Some more interesting properties;

Lemma 6.21. For any $x, y \in \overline{X}$, if $c_1 = (x, y, t)$ and $c_2 = (x, y, t_2)$ in FS(X) for some $t_1, t_2 \in \mathbb{R}$ then for any $z \in X$,

$$|\beta_{x_0}(z,c_1,c_2)| \le \left| \tilde{\theta}_{x,y}^{x_0}(t_1) - \tilde{\theta}_{x,y}^{x_0}(t_2) \right|.$$

Moreover, if $x \in X$ *then*

$$\beta_{x_0}(x, c_1, c_2) = \tilde{\theta}_{x,y}^{x_0}(t_1) - \tilde{\theta}_{x,y}^{x_0}(t_2).$$

Proof. If $c_1 = (x, y, t_1)$ and $c_2 = (x, y, t_2)$ then the Gromov product $\langle x \wr y \rangle_z$ cancels itself out. The special case with $x \in X$ uses $\langle x, y \wr x, x_0 \rangle = -\langle y \wr x_0 \rangle_x$ (by equation (6.9)) and $\tilde{\theta}_{x,y}^{x_0}(t) + \langle y \wr x_0 \rangle_x$ is always positive (by the definition of $\tilde{\theta}_{x,y}^{x_0}$).

The general case of $x \in \overline{X}$ can be shown by considering the various possibilities in turn. If both $\tilde{\theta}_{x,y}^{x_0}(t_1)$ and $\tilde{\theta}_{x,y}^{x_0}(t_2)$ are greater than $\langle x, y \rangle z, x_0 \rangle$ then $\beta_{x_0}(z, c_1, c_2) = \tilde{\theta}_{x,y}^{x_0}(t_1) - \tilde{\theta}_{x,y}^{x_0}(t_2)$. If they are both less than $\langle x, y \rangle z, x_0 \rangle$ then $\beta_{x_0}(z, c_1, c_2) = \tilde{\theta}_{x,y}^{x_0}(t_2) - \tilde{\theta}_{x,y}^{x_0}(t_1)$. This only leaves the possibility that one is greater and one is lesser. Without loss of generality, we assume that $\tilde{\theta}_{x,y}^{x_0}(t_1) < \langle x, y \rangle z, x_0 \rangle < \tilde{\theta}_{x,y}^{x_0}(t_2)$. Hence

$$\begin{aligned} |\beta_{x_0}(z,c_1,c_2)| &= \left| \left| \langle x,y \rangle z,x_0 \rangle - \tilde{\theta}_{x,y}^{x_0}(t_1) \right| - \left| \langle x,y \rangle z,x_0 \rangle - \tilde{\theta}_{x,y}^{x_0}(t_2) \right| \right| \\ &= \left| 2 \langle x,y \rangle z,x_0 \rangle - \tilde{\theta}_{x,y}^{x_0}(t_1) - \tilde{\theta}_{x,y}^{x_0}(t_2) \right| \end{aligned}$$

and if $2\langle x, y \rangle z, x_0 \rangle \leq \tilde{\theta}_{x,y}^{x_0}(t_1) + \tilde{\theta}_{x,y}^{x_0}(t_2)$ then

$$egin{aligned} &|eta_{x_0}(z,c_1,c_2)| = ilde{ heta}_{x,y}^{x_0}(t_1) + ilde{ heta}_{x,y}^{x_0}(t_2) - 2 \langle x,y \wr z,x_0
angle \ &\leq ilde{ heta}_{x,y}^{x_0}(t_2) - ilde{ heta}_{x,y}^{x_0}(t_1) \end{aligned}$$

since $\langle x, y \wr z, x_0 \rangle \geq \tilde{\theta}_{x,y}^{x_0}(t_1)$ by assumption.

Similarly the assumption $\langle x, y \rangle z, x_0 \rangle \leq \tilde{\theta}_{x,y}^{x_0}(t_2)$ can be used to get the inequality if $2\langle x, y \rangle z, x_0 \rangle \geq \tilde{\theta}_{x,y}^{x_0}(t_1) + \tilde{\theta}_{x,y}^{x_0}(t_2)$.

We have gone through all the possibilities so the inequality in the statement of the lemma must always hold. $\hfill \Box$

Definition 6.22. Define the function L_{x_0} : $FS(X) \times FS(X) \to \mathbb{R}$ by

$$L_{x_0}(c,c') = \sup_{z \in X} \left| \beta_{x_0}(z,c,c') \right|.$$
(6.11)

Informally, the function L_{x_0} should represent the distance between the points of *c*, *c*' at time zero.

Proposition 6.23. The function L_{x_0} is a pseudometric on FS(X). Moreover, it satisfies the following properties;

- (i) For all $x, x' \in X$, $L_{x_0}(\underline{x}, \underline{x'}) = \tilde{d}(x, x')$;
- (*ii*) For any $x, y \in \overline{X}$, if $c_1 = (x, y, t_1)$ and $c_2 = (x, y, t_2)$ in FS(X) then

$$L_{x_0}(c_1,c_2) = \left| \tilde{\theta}_{x,y}^{x_0}(t_1) - \tilde{\theta}_{x,y}^{x_0}(t_2) \right|.$$

Proof. To show that L_{x_0} is a pseudometric, we need to show it is symmetric, positive semi-definite, and satisfies the triangle inequality.

The function β_{x_0} is anti-symmetric, so taking the absolute value makes it symmetric and non-negative. To show that $L_{x_0}(c,c) = 0$ for any $c \in FS(X)$, we use lemma 6.21.

Finally, the triangle inequality follows from lemma 6.20(iv).

Therefore the function L_{x_0} is indeed a pseudometric on FS(X).

Property (*i*) follows from the triangle inequality and lemma 6.20(ii).

It remains to show (*ii*). If $x \notin \partial X$ then it follows from lemma 6.21. So assume $x \in \partial X$. Take a sequence $(x_n)_{n \in \mathbb{N}}$ of points in X that converge to x. Then

$$\begin{aligned} \beta_{x_0}(x_n, c_1, c_2) &= l_{x_0}(x_n, c_1) - l_{x_0}(x_n, c_2) \\ &= \left| \tilde{\theta}_{x, y}^{x_0}(t_1) - \langle x, y \wr x_n, x_0 \rangle \right| - \left| \tilde{\theta}_{x, y}^{x_0}(t_2) - \langle x, y \wr x_n, x_0 \rangle \right| \end{aligned}$$

but $\langle x, y \rangle x_n, x_0 \rangle \to -\infty$ as $n \to \infty$ by lemma 6.15, thus we can find some large *n* such that both $\tilde{\theta}_{x,y}^{x_0}(t_1) - \langle x, y \rangle x_n, x_0 \rangle$ and $\tilde{\theta}_{x,y}^{x_0}(t_2) - \langle x, y \rangle x_n, x_0 \rangle$ are positive. Therefore

$$L_{x_0}(c_1,c_2) \geq \left| \tilde{\theta}_{x,y}^{x_0}(t_1) - \tilde{\theta}_{x,y}^{x_0}(t_2) \right|$$

and lemma 6.21 provides the other inequality, hence we get equality. \Box

To make L_{x_0} into a metric we need to be able to distinguish between two points of FS(X). We do this by not only considering the points at time zero, but along the entire formal generalised geodesic, with an appropriate weighting.

Recall that the flow Φ on FS(X) is given by $\Phi_{\tau}(x, y, t) := (x, y, t + \tau)$.

Definition 6.24. Define the function d_{FS} : $FS(X) \times FS(X) \rightarrow \mathbb{R}$ by

$$d_{FS}(c_1, c_2) = \int_{\mathbb{R}} \frac{L_{x_0}\left(\Phi_{\tau}(c_1), \Phi_{\tau}(c_2)\right)}{2e^{|\tau|}} d\tau.$$
 (6.12)

Proposition 6.25. *The function* d_{FS} *is a metric on* FS(X)*.*

Proof. That d_{FS} is a pseudometric follows immediately from the fact that L_{x_0} is a pseudometric, thus it only remains to prove that if $c_1 \neq c_2 \in FS(X)$ then $d_{FS}(c_1, c_2) \neq 0$. Write $c_1 = (x_1, y_1, t_1)$ and $c_2 = (x_2, y_2, t_2)$. First we show that if $x_1 \neq x_2$ then $d_{FS}(c_1, c_2) > 0$.

If $x_i \notin \partial X$ then $\Phi_{\tau}(c_i) \to \underline{x_i}$ as $\tau \to -\infty$. But $\beta_{x_0}(x_1, \underline{x_1}, \underline{x_2}) = -\tilde{d}(x_1, x_2)$ by lemma 6.20(*ii*). Hence

$$|\beta_{x_0}(x_1, \Phi_{\tau}(c_1), \Phi_{\tau}(c_2))| \longrightarrow \tilde{d}(x_1, x_2) \text{ as } \tau \to -\infty.$$

Therefore if $x_1, x_2 \notin \partial X$ and $x_1 \neq x_2$ then the integral for $d_{FS}(c_1, c_2)$ is non-zero.

Suppose $x_1 \in \partial X$ but $x_2 \notin \partial X$. For $\tau \ll 0$ we know that $l_{x_0}(x_0, \Phi_{\tau}(c_2))$ is bounded by $\langle x_2 \wr y_2 \rangle_{x_0} + \langle x_0 \wr y_2 \rangle_{x_2}$ but

$$l_{x_0}(x_0, \Phi_{\tau}(c_1)) = \langle x_1 \wr y_1 \rangle_{x_0} - (t_1 + \tau)$$

and so $|\beta_{x_0}(x_0, \Phi_{\tau}(c_1), \Phi_{\tau}(c_2))| \to \infty$ as $\tau \to -\infty$. Since L_{x_0} is defined as a supremum over X of β_{x_0} we get that $L_{x_0}(\Phi_{\tau}(c_1), \Phi_{\tau}(c_2))$ is also unbounded, and then the integral defining $d_{FS}(c_1, c_2)$ must be non-zero.

So suppose $x_1, x_2 \in \partial X$ and $x_1 \neq x_2$. First assume $x_1 \neq y_2$. For any $z \in X$ the Gromov product $\langle x_1 \wr x_2 \rangle_z$ is finite (since $x_1 \neq x_2$). Thus

$$\langle x_1 \wr y_1 \rangle_z - \langle x_2 \wr y_2 \rangle_z = (\langle x_1 \wr y_1 \rangle_z - \langle x_1 \wr x_2 \rangle_z) - (\langle x_1 \wr x_2 \rangle_z - \langle x_2 \wr y_2 \rangle_z)$$

= $\langle z, x_2 \wr y_2, x_1 \rangle - \langle z, x_1 \wr x_2, y_1 \rangle.$

As $z \to x_1$ we can make $\langle z, x_2 \wr y_2, x_1 \rangle$ arbitrarily large (by lemma 6.15) and $\langle z, x_1 \wr x_2, y_1 \rangle$ arbitrarily close to 0. In particular, we can pick $z \in X$ such that $\langle x_1 \wr y_1 \rangle_z > \langle x_2 \wr y_2 \rangle_z$. Next set $\tau = t_2 - \langle x_2, y_2 \wr z, x_0 \rangle$, i.e. choose τ to minimise $l_{x_0}(z, \Phi_{\tau}(c_2))$. Then

$$\begin{split} \beta_{x_0}(z, \Phi_{\tau}(c_1), \Phi_{\tau}(c_2)) &= \langle x_1 \wr y_1 \rangle_z + |\theta_{x_1, y_1}^{x_0}(t_1 + \tau) - \langle x_1, y_1 \wr z, x_0 \rangle| \\ &- \langle x_2 \wr y_2 \rangle_z - |\tilde{\theta}_{x_2, y_2}^{x_0}(t_2 + \tau) - \langle x_2, y_2 \wr z, x_0 \rangle| \\ &= \langle x_1 \wr y_1 \rangle_z - \langle x_2 \wr y_2 \rangle_z \\ &+ |\tilde{\theta}_{x_1, y_1}^{x_0}(t_1 + \tau) - \langle x_1, y_1 \wr z, x_0 \rangle| \\ &\geq \langle x_1 \wr y_1 \rangle_z - \langle x_2 \wr y_2 \rangle_z \\ &> 0 \end{split}$$

and it follows that $L_{x_0}(\Phi_{\tau}(c_1), \Phi_{\tau}(c_2)) > 0$ and thus $d_{FS}(c_1, c_2) > 0$.

Similarly, if $x_1, x_2 \in \partial X$ with $x_1 \neq x_2$ and $x_2 \neq y_1$ we get $d_{FS}(c_1, c_2) > 0$. However, if $x_1, x_2 \in \partial X$ with $x_1 \neq x_2$ but $x_1 = y_2$ and $x_2 = y_1$, which intuitively means that c_2 is the formal geodesic c_1 in reverse, then we need a different argument. Fix some τ such that $t_1 + \tau \neq -(t_2 + \tau)$, and pick $z \in X$ such that

$$\langle x_1, y_1 \wr z, x_0 \rangle \geq \max\{t_1 + \tau, -(t_2 + \tau)\}.$$

We can find such a $z \in X$ since if $(z_n)_{n \in \mathbb{N}}$ is a sequence in X with $z_n \to y_1$ as $n \to \infty$ then $\langle x_1, y_1 \wr z_n, x_0 \rangle \to \infty$ as $n \to \infty$ by lemma 6.15. Then

$$\begin{split} \beta_{x_0}(z, \Phi_{\tau}(c_1), \Phi_{\tau}(c_2)) &= \langle x_1 \wr y_1 \rangle_z + |(t_1 + \tau) - \langle x_1, y_1 \wr z, x_0 \rangle| \\ &- (\langle x_2 \wr y_2 \rangle_z + |(t_2 + \tau) - \langle x_2, y_2 \wr z, x_0 \rangle|) \\ &= \langle x_1, y_1 \wr z, x_0 \rangle - (t_1 + \tau) \\ &- (\langle x_1, y_1 \wr z, x_0 \rangle + (t_2 + \tau)) \\ &= -(t_1 + \tau) - (t_2 + \tau) \\ &\neq 0 \end{split}$$

by the choice of τ . Hence $L_{x_0}(\Phi_{\tau}(c_1), \Phi_{\tau}(c_2)) > 0$ and thus $d_{FS}(c_1, c_2) > 0$.

Thus if $x_1 \neq x_2$ then $d_{FS}(c_1, c_2) > 0$. Similarly if $y_1 \neq y_2$. So assume that $x_1 = x_2 =: x$ and $y_1 = y_2 =: y$. If x = y then $c_1 = \underline{x} = c_2$ which contradicts the assumption $c_1 \neq c_2$, so assume $x \neq y$. Then

$$\begin{aligned} |\beta_{x_0}(x_0, \Phi_{\tau}(c_1), \Phi_{\tau}(c_2))| &= \left| \left| \tilde{\theta}_{x,y}^{x_0}(t_1 + \tau) - \langle x, y \wr x_0, x_0 \rangle \right| \\ &- \left| \tilde{\theta}_{x,y}^{x_0}(t_2 + \tau) - \langle x, y \wr x_0, x_0 \rangle \right| \\ &= \left| \tilde{\theta}_{x,y}^{x_0}(t_1 + \tau) - \tilde{\theta}_{x,y}^{x_0}(t_2 + \tau) \right| \end{aligned}$$

1

and we can find some $\tau_0 \in \mathbb{R}$ such that $t_1 + \tau_0 \in (-\langle y \wr x_0 \rangle_x, \langle x \wr x_0 \rangle_y)$ and then $\tilde{\theta}_{x,y}^{x_0}(t_1 + \tau_0) = t_1 + \tau_0$. If $t_1 \neq t_2$ then $\tilde{\theta}_{x,y}^{x_0}(t_2 + \tau_0) \neq t_1 + \tau_0$ and so $L_{x_0}(\Phi_{\tau_0}(c_1), \Phi_{\tau_0}(c_2) > 0$.

Therefore $c_1 \neq c_2$ implies $d_{FS}(c_1, c_2) > 0$, so d_{FS} is a metric on FS(X).

7 Properties of the Flow Space

In section 3.3 we defined the set underlying the flow space, and subsequently a metric on FS(X) was constructed, culminating in section 6.3.

This section is dedicated to properties of the metric space $(FS(X), d_{FS})$. Section 7.1 is about the topology of this metric space, section 7.2 considers what happens we change the base-point of X, and section 7.3 shows how an isometric group action on X induces an isometric action on the flow space.

Then section 7.4 contains some nice results about the convergence of formal geodesics in the flow space.

7.1 The induced topology

Given a simplicial complex *X* whose 1-skeleton is a uniformly fine, Gromov hyperbolic graph we can define the flow space as in section 3.3. This is a topological space, which for now we denote by $(FS(X), \tau)$ and use FS(X)to denote the underlying set. In section 6.3 we defined a metric on the set FS(X). We would like the topology induced by this metric to coincide with the original topology τ as given in definition 3.13.

It follows from the definition of the flow space that the diagonal map $X \rightarrow (FS(X), \tau)$ is an embedding. We can also consider the diagonal map $X \rightarrow (FS(X), d_{FS})$.

Proposition 7.1. Let X be a simplicial complex whose 1-skeleton is a uniformly fine, Gromov hyperbolic graph. The map $(X, \tilde{d}) \rightarrow (FS(X), d_{FS})$ given by sending $x \in X$ to the class $\underline{x} := [(x, x, 0)] \in FS(X)$ is an isometric embedding.

Proof. For any $x, x' \in X$ it follows from lemma 6.20(ii) and the triangle inequality that $L_{x_0}(\underline{x}, \underline{x'}) = \tilde{d}(x, x')$. The geodesics are stationary with respect to the flow so in equation (6.12) we get

$$d_{FS}(\underline{x},\underline{x}') = \int_{\mathbb{R}} \frac{\tilde{d}(x,x')}{2e^{|\tau|}} d\tau = \tilde{d}(x,x').$$

Next we consider what happens away from the stationary geodesics. For this, we would like to quote [Min05, Proposition 48]. However, the proof of that proposition uses [Min05, Proposition 38], whose proof used the bilinear extension of the metric \hat{d} from the vertex set to all of X (see equation (6.1)). Hence we state the latter proposition in our case so that we can fix this small problem, as well as explain why the rest of the argument still holds.

Proposition 7.2. Let X be a simplicial complex whose 1-skeleton is a uniformly fine, Gromov hyperbolic graph. Let S be as in theorem 6.13. There exists a constant $\lambda_0 \in [0,1)$ such that for all $\lambda \in [\lambda_0,1)$ there is a $T \ge 0$ such that for all $a, b, c, u \in \overline{X}$ if

- (*a*, *c*, *u*, *b*) ∈ *S*;
- $(b, c, u, a) \in S;$
- $T \leq \max\{\langle a, c \wr u, b \rangle, \langle b, c \wr u, a \rangle\} =: m$

then $(a, b, u, c) \in S$ *and*

$$|\langle a,b \wr u,c \rangle| \leq \lambda^m.$$

Proof. If a, b, c, u are vertices or boundary points then we can use the proof of [Min05, Proposition 38], which only uses Gromov hyperbolicity of the 1-skeleton as well as facts established in our case in theorem 5.29.

In order to extend to arbitrary points in *X* we need to do a bit more work, since we need to show that under the assumptions given we can use the bilinear extension of \hat{d} , i.e. we want

$$\langle a, b \wr u, c \rangle = \langle a, b | u, c \rangle$$

so that we can write $\langle a, b \rangle u, c \rangle$ in terms of the double difference of vertices in the support of the points *a*, *b*, *u*, *c*.

Let *A*, *B* be the constants appearing in a quasi-isometry between \hat{d} and the word metric on the vertex set of *X*, so that

$$\tilde{d}(x,y) = \min\{\hat{d}(x,y), 3(A+B)d_X^1(x,y)\}.$$

The diameter (with respect to the metric d_X^1) of any simplex is ≤ 1 by proposition 6.3(*ii*). Moreover, if the supports of two points $x, y \in X$ are disjoint then $\tilde{d}(x, y) = \hat{d}(x, y)$, by lemma 6.8. Therefore, if $\tilde{d}(x, y) > 6(A + B)$ then

the supports of *x*, *y* are disjoint and $\tilde{d}(x, y) = \hat{d}(x, y)$. So we are interested in finding a lower bound for the distances used to define $\langle a, b \rangle u, c \rangle$.

Without loss of generality suppose $m = \langle b, c \rangle u, a \rangle$. Theorem 6.14 gives

$$\langle b, c \wr u, a \rangle = \langle c \wr u \rangle_b - \langle c \wr a \rangle_b \\ \leq \min \left\{ \tilde{d}(c, b), \tilde{d}(u, b) \right\}$$

and then the assumption $T \leq m$ gives $T \leq \tilde{d}(c, b), \tilde{d}(u, b)$. Moreover,

$$\langle b, c \wr u, a \rangle = -\langle a, u \wr b, c \rangle = -(\langle u \wr b \rangle_a - \langle u \wr c \rangle_a)$$

$$\leq \min \left\{ \tilde{d}(u, a), \tilde{d}(c, a) \right\}$$

and so $T \leq \tilde{d}(c, a)$, $\tilde{d}(u, a)$ as well.

Thus if we take $T \ge 6(A + B)$ then for all four of these pairings $\tilde{d} \equiv \hat{d}$, and we get $\langle a, b \rangle u, c \rangle = \langle a, b | u, c \rangle$. Since we are allowed to increase *T* as we wish, the rest of the proof works as in [Min05, Proposition 38].

Theorem 7.3. Let X be a simplicial complex whose 1-skeleton is a uniformly fine, Gromov hyperbolic graph. Let $\Delta(\overline{X})$ be the diagonal subset of $\overline{X} \times \overline{X}$. Let $FS(X)^{\mathbb{R}}$ denote the stationary orbits, i.e. the orbits fixed by the flow. The map

$$\left(\overline{X} \times \overline{X} \setminus \Delta(\overline{X})\right) \times \mathbb{R} \to \left(FS(X) \setminus FS(X)^{\mathbb{R}}, \tilde{d}\right); \quad (x, y, t) \mapsto [x, y, t]$$

is a homeomorphism.

Proof. The proof is as for [Min05, Proposition 48]. It works in our case using the results we have already proven, for example our theorem 5.32 covers both [Min05, Theorem 32] and [Min05, Proposition 33].

7.2 Change of base-point

The metric d_{FS} on FS(X) constructed in section 6.3 depends on the choice of base-point $x_0 \in X$.

However, we do not want the topology of FS(X) to depend on this base-point so we must look at what would happen if we chose a different base-point.

Let $d_{FS}^{x_0}$ denote the metric on the flow space using x_0 as base-point.

Proposition 7.4. Let X be a simplicial complex whose 1-skeleton is a fine, Gromov hyperbolic graph. Let $x_0, y_0 \in X$ be two choices of base-point. The map

 $\left(FS(X), d_{FS}^{x_0}\right) \rightarrow \left(FS(X), d_{FS}^{y_0}\right); \quad (x, y, t) \mapsto (x, y, t - \langle x, y \wr x_0, y_0 \rangle)$

is an isometry.

$$\tilde{\theta}_{x,y}^{x_0}(t) - \langle x, y \wr x_0, y_0 \rangle = \tilde{\theta}_{x,y}^{y_0}(t - \langle x, y \wr x_0, y_0 \rangle)$$

and so for any $z \in X$

$$l_{x_0}(z,(x,y,t)) = l_{y_0}(z,(x,y,t-\langle x,y \wr x_0,y_0 \rangle)).$$

From this it follows that

$$d_{FS}^{x_0}((x,y,t),(x',y',t')) = d_{FS}^{y_0}((x,y,t-\langle x,y \wr x_0,y_0 \rangle),(x',y',t'-\langle x',y' \wr x_0,y_0 \rangle))$$

i.e. the map $(x, y, t) \mapsto (x, y, t - \langle x, y \rangle x_0, y_0 \rangle)$ is an isometry.

Therefore, although the metric on the flow space depends on the choice of base-point, the topology of the flow space does not.

7.3 The group action on FS(X)

If *X* is a simplicial complex with a uniformly fine, Gromov hyperbolic 1-skeleton then we have constructed a flow space FS(X). If a group *G* acts simplicially on *X* then the metric \tilde{d} is invariant under *G* and we could consider the action of *G* on FS(X) given by

$$g \cdot (x, y, t) := (gx, gy, t).$$

However, the metric d_{FS} is not necessarily invariant under this group action. The problem arises from the dependence of the metric d_{FS} on the base-point x_0 . Hence we define the action of G on FS(X) via

$$g \cdot (x, y, t) := \left(gx, gy, t + \langle x, y \rangle x_0, g^{-1}x_0 \rangle\right).$$

With this definition, we can show

$$l_{x_0}(gz,g\cdot(x,y,t)) = l_{x_0}(z,(x,y,t))$$

for any $z \in X$. From which it follows that for any $c, c' \in FS(X)$ we have $L_{x_0}(g \cdot c, g \cdot c') = L_{x_0}(c, c')$ and thus d_{FS} is invariant under *G*.

So now the *G*-action on *X* induces a *G*-action on FS(X) that preserves the metric d_{FS} .

7.4 Convergence of formal geodesics

If a space is uniquely geodesic, i.e. between any pair of points there is precisely one geodesic between them, then the flow space is just the space of all generalised geodesics. More generally, the flow space was defined to formally replicate the space of generalised geodesics, even if we start with a Gromov hyperbolic metric space that is not necessarily a geodesic space.

We already know that if $y \in X$ then $(x, y, t) \to \underline{y}$ as $t \to \infty$, where \underline{y} denotes the point of the flow space represented by (y, y, 0). Recall that the flow is defined by $\Phi_{\tau}(x, y, t) = (x, y, t + \tau)$ so more generally we can say that $\Phi_{\tau}(x, y, t) \to y$ as $\tau \to \infty$. In particular, for any $x, x' \in \overline{X}$ and $t, t' \in \mathbb{R}$

$$d_{FS}\left(\Phi_{\tau}(x,y,t),\Phi_{\tau}(x',y,t')\right)\longrightarrow 0 \quad \text{as } \tau \to \infty.$$
(7.1)

If $y \in \partial X$ then there is no such point \underline{y} in the flow space, but we could still ask for (7.1) to hold. However, since a geodesic ray will never reach its endpoint, it is possible that one geodesic is always running behind the other. For example, consider \mathbb{R} as a graph with vertex set \mathbb{Z} and base-point 0. Then for all $\tau \in [0, \infty)$

$$d_{FS}(\Phi_{\tau}(0,\infty,0),\Phi_{\tau}(0,\infty,1)) = 1$$

by proposition 6.23(ii). Hence we have to add the quantifier "there exists a $t_0 \in \mathbb{R}^n$ where this t_0 will bring the parametrisations of the geodesics in line. It turns out, that this is the only change we need.

Theorem 7.5. Let X be a simplicial complex whose 1-skeleton is a uniformly fine, Gromov hyperbolic graph. For any $x, x' \in X, y \in \overline{X}$ and all $t, t' \in \mathbb{R}$ there exists a constant $t_0 \in \mathbb{R}$ such that

$$d_{FS}(\Phi_{\tau}(x,y,t),\Phi_{\tau}(x',y,t'+t_0)) \longrightarrow 0 \quad as \ \tau \to \infty.$$

Proof. This follows from [BLR08a, Theorem 8.9]. That article builds on the work in [Min05] and as such it assumes the 1-skeleton of the simplicial complex is uniformly locally finite, but this assumption is only necessary for quoting the results from [Min05], and these results have been proven for a uniformly fine 1-skeleton in this thesis.

Given the existence of the metric *d* that satisfies nice properties (e.g. theorem 6.13 and proposition 7.2) the proof of the theorem is mainly computational. \Box

Recall that if a group *G* is hyperbolic relative to a finite collection \mathcal{P} of subgroups then, by definition, we can find a fine, Gromov hyperbolic graph Γ on which *G* acts such that there are only finitely many *G*-orbits of

vertices and edges. The construction from section 3.1 applied to this graph yields a simplicial complex $X = X_{\eta}(\Gamma)$ whose 1-skeleton is a uniformly fine, Gromov hyperbolic graph. We can define a boundary of X as in section 2.5, and then we also have the space \overline{X} , on which *G* acts. In this case we can use [BLR08a, Theorem 1.5] to say something stronger.

(Recall that the action of G on the flow space is via equation (7.3).)

Theorem 7.6. Let G be a group that is hyperbolic relative to a finite collection \mathcal{P} of subgroups and let X be the associated simplicial complex from section 3.1. Fix a base-point $x_0 \in X$ and let G act on $G \times \overline{X}$ via the diagonal action.

There is a continuous G-equivariant map $j: G \times \overline{X} \to FS(X)$ that satisfies the following property;

For any $\alpha > 0$ there is a function $f_{\alpha} \colon \mathbb{R} \to [0, \infty)$ with $f_{\alpha}(\tau) \to 0$ as $\tau \to \infty$, and there is a $\beta = \beta(\alpha) > 0$ such that for all $g, h \in G$ and $y \in \overline{X}$, if $d_G(g, h) \le \alpha$ then there exists some $t_0 \in [-\beta, \beta]$ such that for all $\tau \in \mathbb{R}$

$$d_{FS}\left(\Phi_{\tau}j(g,y),\Phi_{\tau+t_0}j(h,y)\right) \le f_{\alpha}(\tau). \tag{7.2}$$

The theorem has a lot of quantifiers so we shall try to give an informal explanation of its statement. The map $j: G \times \overline{X} \to FS(X)$ sends the pair (g, y) to some point along the flow orbit with ends $g \cdot x_0$ and y, i.e. there is some $t_{(g,y)}$ such that $j(g, y) = (g \cdot x_0, y, t_{(g,y)})$. Although we haven't explained how to choose $t_{(g,y)}$ it should vary uniformly continuously with g.

The number t_0 represents how far the formal geodesic $(g \cdot x_0, y, t_{(g,y)})$ is ahead or behind the formal geodesic $(h \cdot x_0, y, t_{(h,y)})$ and so $|t_0|$ should be bounded by some expression involving $\tilde{d}(g \cdot x_0, h \cdot x_0)$ and $|t_{(g,y)} - t_{(h,y)}|$, and this expression would vary uniformly continuously with g, h. Therefore, the bound β on $|t_0|$ should depend only on α .



Then the theorem states that $\Phi_{\tau} j(g, y)$ and $\Phi_{\tau+t_0} j(h, y)$ converge uniformly as $\tau \to \infty$.

Proof of Theorem **7.6**. The proof is as for the proof of [BLR08a, Theorem 1.5], but we need to make one remark here.

To prove the map *j* is continuous they use the fact that \overline{X} is metrisable, so we need to use section 2.6 to show \overline{X} is metrisable for our choice of *X*.

8 A Flow Space for a Relatively Hyperbolic Group

Now we want to summarise what happens in the case of a relatively hyperbolic group.

If a group *G* is hyperbolic relative to a finite set \mathcal{P} of subgroups of *G* then, by definition, we can find a fine, Gromov hyperbolic graph Γ on which *G* acts with certain properties (see definition 2.22 for the full list of properties). To such a graph there is an associated simplicial complex $X = X_{\eta}(\Gamma)$ (which also depends upon a choice of parameter $\eta \geq 1$), and this simplicial complex is contractible and finite-dimensional. Furthermore, the 1-skeleton $\mathcal{G} = \mathcal{G}_{\eta}(\Gamma)$ of *X* is a fine, Gromov hyperbolic graph. This was all done in section 3.1.

The group *G* acts on *X* such that there are only finitely many orbits of simplices, and in particular we can say that G is uniformly fine using lemma 2.19.

Therefore there is a metric space $(FS(X), d_{FS})$ associated to X, which has a flow $\Phi: FS(X) \times \mathbb{R} \to FS(X)$ such that two points c_1, c_2 with the same end-point in X converge under this flow (see theorem 7.5). Moreover, there is an embedding $j: G \times \overline{X} \to FS(X)$ such that for any $y \in \overline{X}$ and $g, h \in$ G the elements j(g, y) and j(h, y) in FS(X) converge uniformly under the flow (see theorem 7.6).

We hope that we will be able to use this flow space to help prove a version of the Farrell-Jones Conjecture for relatively hyperbolic groups, as explained at the very start of this thesis.

Bibliography

- [Ali05] E. Alibegović, *A combination theorem for relatively hyperbolic groups*, Bulletin of the London Mathematical Society **37** (2005), no. 3, 459–466.
- [BH99] M. Bridson and A. Haefliger, *Metric spaces of non-positive curvature*, Springer Verlag, 1999.
- [BLR08a] A. Bartels, W. Lück, and H. Reich, Equivariant covers for hyperbolic groups, Geometry & Topology 12 (2008), no. 3, 1799–1882.
- [BLR08b] _____, On the farrell–jones conjecture and its applications, Journal of Topology 1 (2008), no. 1, 57–86.
- [BLR08c] _____, *The K-theoretic Farrell–Jones conjecture for hyperbolic groups*, Inventiones mathematicae **172** (2008), no. 1, 29–70.
- [Bow97] B. Bowditch, *Relatively hyperbolic groups*, preprint (1997), 63pp.
- [Dah03a] F. Dahmani, *Combination of convergence groups*, Geometry and Topology 7 (2003), no. 2, 933–963.
- [Dah03b] _____, Les groupes relativement hyperboliques et leurs bords, Ph.D. thesis, 2003.
- [Els] T. Elsner, *Systolic groups with isolated flats*, preprint.
- [Far98] B. Farb, *Relatively hyperbolic groups*, Geometric and functional analysis **8** (1998), no. 5, 810–840.
- [GdLH90] E. Ghys and P. de La Harpe, *Sur les groupes hyperboliques d'après mikhael gromov*, Birkhäuser, 1990.
- [GM08] D. Groves and J.F. Manning, *Dehn filling in relatively hyperbolic groups*, Israel journal of mathematics **168** (2008), no. 1, 317–429.

- [Gro87] M. Gromov, *Hyperbolic groups*, Essays in group theory **8** (1987), no. 75-263, 2.
- [Kel55] J. Kelley, *General topology*, Graduate Texts in Mathematics **27** (1955).
- [LR05] W. Lück and H. Reich, *The baum-connes and the farrell-jones conjectures in k-and l-theory*, Handbook of K-theory Volume 2 (2005), 703–842.
- [Min01] I. Mineyev, *Straightening and bounded cohomology of hyperbolic groups*, Geometric and Functional Analysis **11** (2001), no. 4, 807–839.
- [Min05] _____, Flows and joins of metric spaces, Geom. Topol. 9 (2005), 403–482. MR 2140987 (2006b:37059)
- [MS02] D. Meintrup and T. Schick, A model for the universal space for proper actions of a hyperbolic group, New York J. Math 8 (2002), 1–7.
- [MY02] I. Mineyev and G. Yu, *The baum-connes conjecture for hyperbolic groups*, Inventiones Mathematicae **149** (2002), no. 1, 97–122.
- [MY06] I. Mineyev and A. Yaman, *Relative hyperbolicity and bounded cohomology*, Preprint (2006).
- [Osi06] D. Osin, *Relatively hyperbolic groups: intrinsic geometry, algebraic properties, and algorithmic problems, Amer Mathematical Society,* 2006.
- [Szc98] A. Szczepański, *Relatively hyperbolic groups.*, The Michigan Mathematical Journal **45** (1998), no. 3, 611–618.